NORANDA INC Form SC 13D/A March 14, 2005

## UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

## **SCHEDULE 13D**

# Under the Securities Exchange Act of 1934 (Amendment No. 1)\*

#### Noranda Inc.

(Name of Issuer) Common Shares, no par value

(Title of Class of Securities)

655422 10 3

(CUSIP Number)

Andrew J. Beck Torys LLP 237 Park Avenue New York, New York 10017 (212) 880-6000

(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

March 9, 2005

(Date of Event Which Requires Filing of this Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of §§240.13d-1(e), 240.13d-1(f) or 240.13d-1(g), check the following box. o

**Note:** Schedules filed in paper format shall include a signed original and five copies of the schedule, including all exhibits. See §240.13d-7 for other parties to whom copies are to be sent.

(Continued on following pages)

(Page 1 of 17 Pages)

\*The information required on the remainder of this cover page shall not be deemed to be filed for the purpose of Section 18 of the Securities Exchange Act of 1934 ( Act ) or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

## **TABLE OF CONTENTS**

- Item 1. Security and Issuer.
- Item 2. Identity and Background
- Item 3. Source and Amount of Funds or Other Consideration.
- Item 4. Purpose of Transaction.
- Item 5. Interest in Securities of the Issuer.
- <u>Item 6. Contracts, Arrangements, Understandings or Relationships with Respect to Securities of the</u>

Issuer.

- Item 7. Material to be Filed as Exhibits.
- **SIGNATURE**
- **SCHEDULE I BRASCAN CORPORATION**
- SCHEDULE II PARTNERS LIMITED

CUSIP No	o. 65:	5422	2 10 3	Page 2 of 17 pages
			Reporting Person: AN CORPORATION	I.R.S. Identification Nos. of above persons (entities only):
(		0	e Appropriate Box if a Men Joint Filing	nber of a Group (See Instructions):
3.	SEC	Use	Only:	
	Sour OO	ce o	f Funds (See Instructions):	
5. (	Chec	k if	Disclosure of Legal Procee	dings Is Required Pursuant to Items 2(d) or 2(e): o
	Citiz ONT		ip or Place of Organization O	:
Number of		7.	Sole Voting Power: 122,597,952	
Shares Beneficial Owned b Each Reportin	у	8.	Shared Voting Power:	
Person W	ith	9.	Sole Dispositive Power: 122,597,952	
	•	10.	Shared Dispositive Power:	

11.	Aggregate Amount Beneficially Owned by Each Reporting Person: 122,597,952
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions): o
13.	Percent of Class Represented by Amount in Row (11): 41.26%
14.	Type of Reporting Person (See Instructions): CO, HC

CUSIP No	o. 65	5422	2 10 3 Page 3 of 17 pages
			Reporting Person:  I.R.S. Identification Nos. of above persons (entities only):  ERS LIMITED
•		0	the Appropriate Box if a Member of a Group (See Instructions):  Joint Filing
3.	SEC	Use	e Only:
	Sour AF	ce o	f Funds (See Instructions):
5. (	Chec	k if	Disclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e): o
	Citiz ONT		nip or Place of Organization: IO
Number	of	7.	Sole Voting Power:
Shares Beneficia Owned b Each Reportin	lly by	8.	Shared Voting Power: 122,597,952
Person W	ith	9.	Sole Dispositive Power:
		10.	Shared Dispositive Power: 122,597,952

11.	Aggregate Amount Beneficially Owned by Each Reporting Person: 122,597,952
12.	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions): o
13.	Percent of Class Represented by Amount in Row (11): 41.26%
14.	Type of Reporting Person (See Instructions): CO, HC

CUSIP No. 655422 10 3

#### SCHEDULE 13D

Page 4 of 17 pages

## Item 1. Security and Issuer.

This Amendment No. 1 amends the Schedule 13D originally filed with the Securities and Exchange Commission on October 7, 2003 with respect to beneficial ownership of the common shares, no par value (the Common Shares ) of Noranda Inc. (Noranda). This Amendment No. 1 is being filed to report a change to the disclosure previously provided under Item 2. Identity and Background, Item 3. Source and Amount of Funds or Other Consideration, Item 4. Purpose of Transaction, Item 5. Interest in Securities of the Issuer, Item 6. Contracts, Arrangements, Understandings or Relationships with respect to the Securities of the Issuer and Item 7. Material to be Filed as Exhibits.

## Item 2. Identity and Background

- (a) This Schedule 13D is being filed by each of the following persons (the Reporting Persons ):
  - (i) Brascan Corporation (Brascan), a corporation formed under the laws of the Province of Ontario; and
  - (ii) Partners Limited (formerly EdperPartners Limited) ( Partners ), a corporation formed under the laws of the Province of Ontario that owns all of Brascan s Class B Limited Voting Shares and approximately 3% of Brascan s Class A Limited Voting Shares on a fully diluted basis.
- (b)-(c) Schedules I and II hereto set forth a list of all the directors and executive officers (the Scheduled Persons), and their respective principal occupations and addresses, of each of Brascan and Partners.
- (d)-(e) During the last five years, none of the Reporting Persons and, to the Reporting Persons knowledge, none of the Scheduled Persons, has been: (i) convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors); or (ii) a party to a civil proceeding of a judicial or administrative body of competent jurisdiction as a result of which, he, she or it was or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, Federal or State securities laws or finding any violation with respect to such laws.
- (f) Set forth on Schedules I and II hereto are the citizenships of each of the directors and executive officers of each of Brascan and Partners, respectively.

## Item 3. Source and Amount of Funds or Other Consideration.

See item 4.

CUSIP No. 655422 10 3

#### SCHEDULE 13D

Page 5 of 17 pages

## **Item 4. Purpose of Transaction.**

On March 9, 2005, Brascan announced that it will support an issuer bid (the Issuer Bid ) by Noranda, pursuant to which Noranda will offer to purchase approximately 63.4 million Common Shares in exchange for US\$1.25 billion aggregate issue price of junior preferred shares (the Junior Preferred Shares ), consisting of US\$500 million aggregate issue price of Series 1 Shares, US\$500 million aggregate issue price of Series 2 Shares and US\$250 million aggregate issue price of Series 3 Shares.

Brascan also announced its support for a plan (the Merger ) by Noranda to combine Noranda and Falconbridge Limited (Falconbridge ). Following the Issuer Bid and the Merger, Brascan s common share ownership of Noranda will decline from approximately 41% to between 16% and 26%.

Junior Preferred Share Terms

The following is a summary of the material attributes of the Junior Preferred Shares:

Ranking. The Junior Preferred Shares will rank junior to the existing preferred shares of Noranda with respect to priority in the payment of dividends and the distribution of assets of Noranda in the event of any liquidation, dissolution or winding up of Noranda or other distribution of assets of Noranda among its shareholders for the purpose of winding up its affairs.

Dividends: Noranda may pay the dividends on the Junior Preferred Shares either in cash or, except in limited circumstances, in freely-tradable Common Shares (the number of shares determined by dividing the declared dividend amount by 95% of the US dollar equivalent of the weighted average trading price of the Common Shares on the Toronto Stock Exchange for a period of 20 consecutive trading days (the Current Market Price ) ending on the fourth day prior to the date specified for payment of the dividend, less any tax required to be deducted and withheld).

No Extraordinary Dividends on Common Shares. For so long as Junior Preferred Shares having an aggregate issue price of at least US\$300 million remain outstanding, Noranda will not declare, pay or set apart for payment any extraordinary dividend without the prior approval of holders of not less than two-thirds of each series of Junior Preferred Shares present in person or by proxy at a duly called meeting of shareholders.

Redemption: The Series 1 Shares (i) are redeemable, in cash, by Noranda at any time on or before June 30, 2008 at US\$25.25 per share plus all accrued and unpaid dividends and thereafter at US\$25.00 per share plus all accrued and unpaid dividends and (ii) must be redeemed on the date that is five years from the date of issue (the Series 1 Final Redemption Date ) at US\$25.00 per share plus all accrued and unpaid dividends.

CUSIP No. 655422 10 3

SCHEDULE 13D

Page 6 of 17 pages

The Series 2 Shares (i) are redeemable, in cash, by Noranda at any time on or before June 30, 2010 at US\$25.25 per share plus all accrued and unpaid dividends and thereafter until June 29, 2012 at US\$25.00 per share plus all accrued and unpaid dividends and (ii) must be redeemed on June 30, 2012 the Series 2 Final Redemption Date ) at US\$25.00 per share plus all accrued and unpaid dividends.

The Series 3 Shares (i) are redeemable, in cash, by Noranda at any time on or before June 30, 2013 at US\$25.25 per share plus all accrued and unpaid dividends and thereafter until June 29, 2015 at US\$25.00 per share plus all accrued and unpaid dividends and (ii) must be redeemed on June 30, 2015 the Series 3 Final Redemption Date ) at US\$25.00 per share plus all accrued and unpaid dividends.

Conversion. Noranda will be entitled, at its option, in lieu of redemption, to convert all or a part of the applicable series of Junior Preferred Shares on the Series 1 Final Redemption Date, the Series 2 Final Redemption Date or the Series 3 Final Redemption Date (as applicable) into freely tradable Common Shares provided no change of control event has occurred, no extraordinary dividend has been paid and certain other events have not occurred. The number of Common Shares issued on such conversion would be determined by dividing US\$25.00, together with all accrued and unpaid dividends, by the greater of US\$2.00 (as may be adjusted for stock splits and similar occurrences) and 90% of the US dollar equivalent of the Current Market Price determined as at the date specified for conversion.

Retraction. In the event of a change of control event: (i) the holders of the Series 1 Shares will be entitled to require Noranda to redeem all of their Series 1 Shares for US\$25.25 per share plus accrued and unpaid dividends if so redeemed on or before June 30, 2008 and if so redeemed after such date for US\$25.00 per share plus accrued and unpaid dividends; (ii) the holders of the Series 2 Shares will be entitled to require Noranda to redeem all of their Series 2 Shares for US\$25.25 per share plus accrued and unpaid dividends if so redeemed on or before June 30, 2010 and for US\$25.00 per share plus accrued and unpaid dividends if so redeemed after June 30, 2010; and (iii) the holders of the Series 3 Shares will be entitled to require Noranda to redeem all of their Series 3 Shares for US\$25.25 per share plus accrued and unpaid dividends if so redeemed on or before June 30, 2013 and for US\$25.00 per share plus accrued and unpaid dividends if so redeemed after June 30, 2013.

Application of Net Proceeds. From and after June 30, 2010, Noranda will apply the full net proceeds from (i) the issuance of equity securities for cash (other than in certain limited circumstances) or (ii) the sale of any capital assets outside of the ordinary course of business with a sale price exceeding Cdn.\$250 million, to the redemption of the Junior Preferred Shares.

CUSIP No. 655422 10 3

#### SCHEDULE 13D

Page 7 of 17 pages

*Voting:* The Junior Preferred Shares will be entitled, voting collectively, to elect two directors to the board of directors of Noranda and, upon the occurrence of certain events, will be entitled, voting collectively, to elect three additional directors. Except as otherwise provided by law or as specifically provided above, and except for meetings of the holders of Junior Preferred Shares as a class and meetings of all holders of each series of Junior Preferred Shares as a series, the holders of the Junior Preferred Shares will not be entitled as such to receive notice of, or to attend, or to vote at, any meeting of shareholders of Noranda.

## Lock Up Agreement

As a component of the Issuer Bid, Brascan and Noranda have entered into a lock up agreement, whereby Brascan has agreed to deposit to the Issuer Bid, within three business days of the date of the offer, and subject to withdrawal in certain circumstances, all of the Common Shares that it beneficially owns on such deposit date. Brascan may withdraw any and all Common Shares deposited under the Issuer Bid if Brascan receives (i) a bona fide offer from an arms—length third party to acquire, directly or indirectly, any or all of the Common Shares beneficially owned by Brascan or (ii) a proposal for a merger, acquisition, take-over bid, sale of material assets, liquidation, material sale of shares or rights or interests therein or similar transactions; in either case, which the board of directors of Brascan considers would result in a transaction more favourable to Brascan than the Issuer Bid. In addition, Brascan may, on or before the deposit date, dispose of, or enter into an agreement to dispose of, any or all of the Common Shares that it beneficially owns.

A copy of the lock up agreement is attached hereto as Exhibit 5.

## Registration Rights Agreement

Brascan and Noranda have entered into a registration rights agreement that takes effect on the date that the Issuer Bid is completed and provides that, upon Brascan s request and subject to certain conditions, Noranda will register or qualify Common Shares beneficially owned by Brascan under United States securities laws and/or Canadian securities laws. This agreement is designed to facilitate the sale of Brascan s Common Shares held after completion of the Issuer Bid.

A copy of the registration rights agreement is attached hereto as Exhibit 6.

Other than as described or contemplated above, none of the Reporting Persons and, to the Reporting Persons knowledge, the Scheduled Persons, has any current plans or proposals that relate to or would result in:

- (a) the acquisition by any person of additional securities of Noranda, or the disposition of securities of Noranda;
- (b) an extraordinary corporate transaction, such as a merger, reorganization or liquidation, involving Noranda or any of its subsidiaries;

CUSIP No. 655422 10 3 SCHEDULE 13D Page 8 of 17 pages

- (c) a sale or transfer of a material amount of assets of Noranda or any of its subsidiaries;
- (d) any change in the present board of directors or management of Noranda, including any plans or proposals to change the number or term of directors or to fill any existing vacancies on the board;
- (e) any material change in the present capitalization or dividend policy of Noranda;
- (f) any other material change in Noranda s business or corporate structure;
- (g) changes in Noranda s charter, bylaws or instruments corresponding thereto or other actions which may impede the acquisition of control of Noranda by any person;
- (h) causing a class of securities of Noranda to be delisted from a national securities exchange or to cease to be authorized to be quoted in an inter-dealer quotation system of a registered national securities association;
- (i) a class of equity securities of Noranda becoming eligible for termination of registration pursuant to Section 12(g)(4) of the Securities Exchange Act of 1934; or
- (j) any action similar to any of those enumerated above.

#### Item 5. Interest in Securities of the Issuer.

(a)-(b) As of the date hereof, each of the Reporting Persons may be deemed to be the beneficial owner of 122,597,952 Common Shares. Such Common Shares constitute approximately 41.26% of the issued and outstanding Common Shares based on the number of Common Shares outstanding as of March 8, 2005. Brascan may be deemed to have the sole power to vote or direct the vote of the Common Shares beneficially owned by it with respect to those matters described above or to dispose of such Common Shares. Partners may be deemed to have shared power (with Brascan) to vote or direct the vote of the Common Shares beneficially owned by it with respect to those matters described above or to dispose of such Common Shares.

If all holders of Common Shares deposit their common shares under the Issuer Bid, and assuming that (i) the Brascan deposits (and do not withdraw) all the Common Shares it owns as of the date of this statement under the Issuer, and (ii) Noranda is successful in completing the Merger (and by virtue of the Merger acquires all outstanding common shares of Falconbridge) immediately following the completion of these transactions, Brascan will hold approximately 26%, or 96.6 million, of the Common Shares and approximately 41%, or 20.5 million, of the Junior Preferred Shares. If none of the other holders of the Common Shares deposit their shares to the Issuer Bid and Noranda is successful in completing the

CUSIP No. 655422 10 3 SCHEDULE Page 9 of 17 pages

Merger, immediately following the completion of these transactions, Brascan would own approximately 16%, or 59.2 million, of the Common Shares and 100%, or 50 million, of the Junior Preferred Shares.

- (c) None of the Reporting Persons and, to the Reporting Persons knowledge, the Scheduled Persons, has effected any transaction in the Common Shares during the past 60 days, other than as disclosed herein.
- (d) No person is known to any of the Reporting Persons or, to the Reporting Persons knowledge, the Scheduled Persons, to have the right to receive or the power to direct the receipt of dividends from, or proceeds from the sale of, any such Common Shares.
- (e) Not applicable.

Item 6. Contracts, Arrangements, Understandings or Relationships with Respect to Securities of the Issuer. Partners is the major shareholder of Brascan. Agreements entered into in connection with the Issuer Bid are described under item 5.

## Item 7. Material to be Filed as Exhibits.

Corporation and Noranda Inc.

- Exhibit 1 Joint Filing Agreement, dated as of October 6, 2003, between Brascan Corporation and Partners Limited.\*
   Exhibit 2 Subscription Agreement, dated as of July 29, 2003, between Brascan Corporation and Noranda Inc.\*
   Exhibit 3 Letter Agreement, dated July 29, 2003, between Brascan Corporation and Noranda Inc.\*
   Exhibit 4 Letter Agreement, dated August 11, 2003, between Brascan Corporation and Noranda Inc.\*
   Exhibit 5 Lock Up Agreement, dated March 8, 2005, between Brascan Corporation, Brascade Corporation and Noranda Inc.
   Exhibit 6 Registration Rights Agreement, dated March 8, 2005, between Brascan Corporation, Brascade
- Previously filed.

## **Table of Contents**

CUSIP No. 655422 10 3

# SCHEDULE 13D SIGNATURE

Page 10 of 17 pages

After reasonable inquiry and to the best of the undersigned s knowledge and belief, each of the undersigned certifies as to itself that the information set forth in this amendment is true, complete and correct.

Dated: March 11, 2005

## **BRASCAN CORPORATION**

By: /s/ Alan V. Dean Name: Alan V. Dean

Title: Senior Vice-President and Secretary

## **PARTNERS LIMITED**

By: /s/ Brian D. Lawson Name: Brian D. Lawson Title: President and Secretary

CUSIP No. 655422 10 3

#### SCHEDULE 13D

Page 11 of 17 pages

## SCHEDULE I BRASCAN CORPORATION

James J. Blanchard

Citizenship: United States

Business Address: 1200 1 Street N.W., 6th Floor, Washington, D.C. 20036-2412

**Present Principal** 

Occupation or Partner

**Employment:** 

**Employer:** DLA Piper Rudnick

**Employer s Business:** Law firm

**Employer s Address:** Same as Business Address

Jack L. Cockwell

Citizenship: Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 300, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or Group Chairman

**Employment:** 

**Employer:** Brascan Corporation

**Employer s Business:** Real estate, power generation and financial services

**Employer s Address:** Same as Business Address

Sen. J. Trevor Eyton

Citizenship: Canada

**Business Address:** 44 Victoria Street, Suite #400, Toronto, Ontario M5C 1Y2

**Present Principal** 

Occupation or Member of the Senate of Canada

**Employment:** 

**Employer:** The Senate of Canada

**Employer s Business:** Government

**Employer s Address:** Room 561-S, Centre Block, Parliament Buildings, Ottawa, Ontario K1A 0A4

J. Bruce Flatt

Citizenship: Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 300, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or President & Chief Executive Officer

**Employment:** 

**Employer:** Brascan Corporation

**Employer s Business:** Real estate, power generation and financial services

**Employer s Address:** Same as Business Address

Julia E. Foster

Citizenship: Canada

**Business Address:** 151 Bloor Street West, 5th Floor, Toronto, Ontario M5S 1T6

CUSIP No. 655422 10 3 SCHEDULE 13D Page 12 of 17 pages

**Present Principal** 

Occupation or Chair

**Employment:** 

Employer: Ontario Arts Council
Employer s Business: Arts funding organization
Employer s Address: Same as Business Address

James K. Gray

Citizenship: Canada

**Business Address:** 150 Sixth Ave. S.W., PetroCanada Centre, Suite 3370,

Calgary, Alberta T2P 3Y7

**Present Principal** 

Occupation or Corporate Director

**Employment:** 

Employer: N/A
Employer s Business: N/A
Employer s Address: N/A

Robert J. Harding

Citizenship: Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 300, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or Chairman

**Employment:** 

**Employer:** Brascan Corporation

**Employer s Business:** Real estate, power generation and financial services

**Employer s Address:** Same as Business Address

David W. Kerr

Citizenship: Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 300, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or Chairman

**Employment:** 

**Employer:** Noranda Inc.

**Employer s Business:** Mining and metallurgy **Employer s Address:** Same as Business Address

Philip B. Lind

Citizenship: Canada

**Business Address:** 333 Bloor Street E., 10<sup>th</sup> Floor, Toronto, Ontario M4W 1G9

**Present Principal** 

Occupation or Vice-Chairman

**Employment:** 

**Employer:** Rogers Communications Inc.

**Employer s Business:** Diversified communications company

**Employer s Address:** Same as Business Address

Roy MacLaren

Citizenship: Canada

**Business Address:** 425 Russell Hill Road, Toronto, Ontario M5P 2S4

CUSIP No. 655422 10 3 SCHEDULE 13D Page 13 of 17 pages

**Present Principal** 

Occupation or Corporate Director

**Employment:** 

Employer: N/A
Employer s Business: N/A
Employer s Address: N/A

G. Wallace F. McCain

Citizenship: Canada

**Business Address:** 30 St. Clair Ave. W., #1500, Toronto, Ontario M4V 3A2

**Present Principal** 

Occupation or Chairman

**Employment:** 

**Employer:** Maple Leaf Foods Inc.

**Employer s Business:** Processed Food Manufacturer **Employer s Address:** Same as Business Address

Jack M. Mintz

Citizenship: Canada

**Business Address:** 67 Yonge St., #300, Toronto, Ontario M5E 1J8

**Present Principal** 

Occupation or President & Chief Executive Officer

**Employment:** 

Employer: C.D. Howe Institute
Employer s Business: Public policy institute
Employer s Address: Same as Business Address

George E. Myhal

**Citizenship:** Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 300, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or Chief Operating Officer

**Employment:** 

**Employer:** Brascan Corporation

**Employer s Business:** Real estate, power generation and financial services

**Employer s Address:** Same as Business Address

Saul Shulman

Citizenship: Canada

**Business Address:** #2300 200 King Street W., Toronto, Ontario M5H 3W5

**Present Principal** 

Occupation or Chief Executive Officer

**Employment:** 

Employer: MLG Management Inc.
Employer s Business: Business consultants
Employer s Address: Same as Business Address

George S. Taylor

Citizenship: Canada

**Business Address:** R.R. #3, 4675 Line 3, Saint Marys, Ontario N4X 1C6

**Present Principal** 

Occupation or Corporate Director

**Employment:** 

**Employer:** N/A

### **Table of Contents**

CUSIP No. 655422 10 3 SCHEDULE 13D Page 14 of 17 pages

**Employer** s **Business:** N/A **Employer** s **Address:** N/A

Brian D. Lawson

Citizenship: Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 300, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or Chief Financial Officer

**Employment:** 

**Employer:** Brascan Corporation

**Employer s Business:** Real estate, power generation and financial services

**Employer s Address:** Same as Business Address

Jeffrey M. Blidner

Citizenship: Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 300, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or Managing Partner

**Employment:** 

**Employer:** Brascan Corporation

**Employer s Business:** Real estate, power generation and financial services

**Employer s Address:** Same as Business Address

Samuel J.B. Pollock

Citizenship: Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 300, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or Managing Partner

**Employment:** 

**Employer:** Brascan Corporation

**Employer s Business:** Real estate, power generation and financial services

**Employer s Address:** Same as Business Address

### **Table of Contents**

CUSIP No. 655422 10 3 SCHEDULE 13D Page 15 of 17 pages

SCHEDULE II PARTNERS LIMITED

Ian G. Cockwell

Citizenship: Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 4430, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or President & Chief Executive Officer

**Employment:** 

**Employer:** Brookfield Homes Corporation **Employer s Business:** Residential real estate development

**Employer s Address:** Same as Business Address

Jack L. Cockwell see Schedule I

J. Bruce Flatt see Schedule I

Robert J. Harding see Schedule I

David W. Kerr see Schedule I

Brian D. Lawson see Schedule I

George E. Myhal see Schedule I

Timothy E. Price

Citizenship: Canada

**Business Address:** BCE Place, 181 Bay Street, Suite 4420, Toronto, Ontario M5J 2T3

**Present Principal** 

Occupation or Group Chairman, Funds Management

**Employment:** 

**Employer:** Brascan Corporation

**Employer s Business:** Real estate, power generation and financial services

**Employer s Address:** Same as Business Address

## EXHIBIT 5 LOCK-UP AGREEMENT

## **STRICTLY CONFIDENTIAL**

March 8, 2005

Board of Directors Brascan Corporation 181 Bay Street, Suite 300 Toronto, Ontario M5J 2T3

Board of Directors Brascade Corporation 181 Bay Street, Suite 300 Toronto, Ontario M5J 2T3

Attention: Mr. Robert Harding, Chairman of the Board, Brascan Corporation

Mr. Edward C. Kress, Chairman of the Board, Brascade Corporation

Dear Sirs:

The board of directors (the **Board** ) of Noranda Inc. (the **Corporation** ) has approved the terms of an issuer bid (the **Issuer Bid** ), pursuant to which the Corporation may make a bid, on substantially the terms and conditions summarized in Schedule A, to all of its shareholders for a specified number of common shares in return for preferred shares (the **Preferred Shares** ). The Board has also approved the terms of a transaction (the **Insider Bid** ) pursuant to which the Corporation may make a bid for the issued and outstanding common shares of Falconbridge Limited not owned by the Corporation in exchange for common shares of the Corporation, on substantially the terms and conditions summarized in Schedule B.

This letter agreement (the **Agreement**) sets out the terms and conditions upon which Corporation agrees to make an offer (the **Offer**) for a specified number of its issued and outstanding common shares (the **Shares**), pursuant to the Issuer Bid, and sets out the obligations and commitments of the Corporation in connection therewith.

This Agreement also sets out the terms and conditions upon which Brascan Corporation ( **Brascan** ) and Brascade Corporation ( **Brascade** ) (together, the **Sellers** ) each agrees to deposit or cause to be deposited under the Offer all the common shares in the capital of the Corporation that are owned by it or any of its Wholly Owned Subsidiaries (as defined below) on the Date of Deposit (as defined below) and sets out the obligations and commitments of the Sellers in connection therewith.

# ARTICLE 1 THE OFFER

- **1.1** Timing. The Corporation agrees to make the Offer for the number Shares specified in Schedule A (by way of mail or as otherwise permitted under applicable securities laws, including exemptive relief) not later than 21 days following the date hereof.
- **1.2** <u>Conditions Precedent</u>. The Corporation shall not be required to make the Offer (and the Corporation may, without prejudice to any other rights, terminate this Agreement by written notice to the Sellers) if:
  - 1.2.1 prior to the making of the Offer, (i) any act, action, suit or proceeding shall have been taken or threatened before or by any domestic or foreign arbitrator, court or tribunal or governmental agency or other regulatory authority or administrative agency or commission or by an elected or appointed public official or private person (including, without limitation, any individual, corporation, firm, group or other entity) in Canada or elsewhere, whether or not having the force of law, or (ii) any law, regulation, rule or policy shall have been proposed, enacted, promulgated or applied, in the case of both clause (i) and (ii) to cease trade, enjoin, prohibit or impose material limitations or conditions on the purchase by or the sale to the Corporation of the Shares, the issuance of the Preferred Shares or the consummation of the Issuer Bid or the rights of the Corporation to cancel the Shares; or
  - 1.2.2 at the time the Corporation proposes to make the Offer, there exists any prohibition at law against the Corporation making the Offer or taking up and paying for the Shares under the Issuer Bid.

The foregoing conditions are for the mutual benefit of the Corporation and each of the Sellers and may be waived by the Corporation or the Sellers in whole or in part at any time and shall be deemed to have been waived by the Corporation upon the making of the Offer.

# ARTICLE 2 REPRESENTATIONS AND WARRANTIES

- **2.1** Representations and Warranties of the Sellers. The Sellers, on a joint and several basis, hereby represent and warrant to the Corporation that:
  - 2.1.1 <u>Incorporation and Authorization</u>. Each of the Sellers is a corporation duly incorporated and validly existing under the laws of its jurisdiction of incorporation; has all necessary power, authority, capacity and right, and has received all requisite approvals to enter into this Agreement and to deposit the Shares that are owned by it or any of its Wholly Owned Subsidiaries on the Date of Deposit, which Shares will be free and clear of any Encumbrances on the Date of Deposit.

2

#### **Table of Contents**

- 2.1.2 **Enforceability**. This Agreement has been duly executed and delivered by each of the Sellers and constitutes a legal, valid and binding agreement enforceable by the Corporation against each of the Sellers in accordance with its terms subject, however, to limitations with respect to enforcement imposed by law in connection with bankruptcy or similar proceedings, the equitable power of the court to stay proceedings before them and the execution of judgments and to the extent that equitable remedies such as specific performance and injunction are in the discretion of the court from which they are sought.
- 2.1.3 Ownership of Shares. The Sellers own directly or indirectly through Wholly Owned Subsidiaries 122,597,952 Shares (the Sellers Shares). Each of the Sellers has the exclusive right to dispose of its portion of the Sellers Shares as provided in this Agreement; and it is not a party to, bound or affected by or subject to, any charter or by-law provision, statute, regulation, judgment, order, decree or law which would be violated, contravened, breached by, or under which default would occur as a result of the execution and delivery of this Agreement or the consummation of any of the transactions provided for in this Agreement subject only to making necessary filings with or giving notifications to any governmental, administrative or regulatory authority having jurisdiction.
- 2.1.4 **No Agreements**. No person, firm or corporation has any agreement or option, or any right or privilege (whether by law, pre-emptive or contractual) capable of becoming an agreement or option, for the purchase, requisition or transfer from each Seller or the Wholly Owned Subsidiaries of any of each Seller s portion of the Sellers Shares or any interest therein or right thereto, except pursuant to this Agreement.
- 2.1.5 **<u>Yoting</u>**. None of the Sellers or the Wholly Owned Subsidiaries has previously granted or agreed to grant any ongoing proxy in respect of the Sellers Shares or entered into any voting trust, vote pooling or other agreement with respect to the right to vote, call meetings of shareholders or give consents or approvals of any kind as to the Sellers Shares.
- 2.1.6 <u>Consents</u>. No consent, waiver, approval, authorization, exemption, registration, license or declaration of or by, or filing with, or notification to any governmental, administrative or regulatory authority is required to be made or obtained by the Sellers in connection with (i) the execution and delivery by the Sellers and enforcement against the Sellers of this Agreement or (ii) the consummation of any transactions by the Sellers provided for herein, except for, in either case, approval for listing the Preferred Shares on the Toronto Stock Exchange and the filing of press releases and material change reports and early warning reports under applicable securities legislation.
- **2.2** Representations and Warranties of the Corporation. The Corporation hereby represents and warrants to the Sellers that it is a corporation validly existing under the laws of its jurisdiction of incorporation; it has all necessary power, authority, capacity and right, and has received all requisite approvals (other than approval for listing the Preferred Shares on the Toronto Stock Exchange), to enter into this Agreement and to complete the transactions contemplated hereby; and, upon the due execution and delivery of this Agreement by the Sellers, this Agreement shall be a valid and binding agreement

3

### **Table of Contents**

enforceable by the Sellers against the Corporation in accordance with its terms subject, however, to limitations with respect to enforcement imposed by law in connection with bankruptcy or similar proceedings, the equitable power of the courts to stay proceedings before them and the execution of judgments and to the extent that equitable remedies such as specific performance and injunction are in the discretion of the court from which they were sought.

# ARTICLE 3 COVENANTS

- **3.1** Covenants of the Sellers. Each Seller hereby covenants, as to itself, that until this Agreement has been terminated by the Corporation, pursuant to section 5.2, or either of the Sellers, pursuant to section 5.1, it will:
  - 3.1.1 notify the Corporation of (a) any agreement relating to the disposition of all or any of the Sellers Shares, which either Seller may enter into after the date hereof but prior to the completion of the Offer or (b) any offer or proposal of the nature referred to in section 4.2 received by it or announced (or known to the Sellers);
  - 3.1.2 vote in favour of any resolution proposed by the Corporation in order to give effect to or otherwise related to the Issuer Bid (including an amendment to the articles of the Corporation to create the Junior Preferred Shares) whether before or after the making of the Offer; and
  - 3.1.3 if the Preferred Shares received by the Sellers are issued from the class of preferred shares of the Corporation existing as of the date of this Agreement (the **Senior Preferred Shares**), agree to subordinate the Preferred Shares to any and all series of the Senior Preferred Shares that are issued and outstanding as of the date of this Agreement.
- **3.2** Covenants of the Corporation. The Corporation hereby covenants that it will:
  - 3.2.1 use its commercially reasonable efforts to successfully complete the Issuer Bid and the Insider Bid on the terms set out in Schedule A and Schedule B, respectively, which terms may not be varied or waived, except with the prior written consent of the Sellers;
  - 3.2.2 prepare the documents (the **Documents** ) required by the Securities Act (*Ontario*) or other applicable laws in connection with the Issuer Bid in accordance with applicable laws and permit the Sellers and their counsel to review and comment upon drafts of all Documents, and give reasonable consideration to such comments and all information regarding the Issuer Bid provided by the Sellers;
  - 3.2.3 notify in writing the Sellers of the date of the Offer;
  - 3.2.4 deliver forthwith to the Sellers any of the Sellers Shares deposited under the Offer and subsequently withdrawn pursuant to Article 4;

4

### **Table of Contents**

- 3.2.5 at its upcoming annual and special meeting of shareholders, propose a resolution that a class of junior preferred shares (the **Junior Preferred Shares**) of the Corporation be created, which class will be subordinated to the Senior Preferred Shares, and (i) if the Preferred Shares have not yet been issued pursuant to the Issuer Bid, issue the Preferred Shares from the class of Junior Preferred Shares or (ii) if the Preferred Shares have been issued pursuant to the Issuer Bid, convert the issued Preferred Shares into three series of Junior Preferred Shares on identical terms to the Preferred Shares except with respect to ranking on payment of dividends and upon liquidation, dissolution or winding up; and
- 3.2.6 use its best efforts to seek the listing of the Preferred Shares on the Toronto Stock Exchange upon their issuance under the Issuer Bid.

# ARTICLE 4 DEPOSIT AND WITHDRAWAL

- **4.1 Deposit.** Subject to section 4.2, each of the Sellers hereby agrees to deposit or cause to be deposited under the Offer all of the Shares that are owned at the time of deposit by it or any of the direct or indirect Wholly Owned Subsidiaries on or before the Date of Deposit, which Shares will be free and clear of any Encumbrances on the Date of Deposit, together with a duly completed and executed letter of transmittal. For greater certainty, nothing in this Agreement will require either of the Sellers to exercise or cause to be exercised the conversion rights attached to the adjustable rate convertible debentures of the Corporation held by it.
- **4.2** Superior Proposal. The parties hereby acknowledge and agree that the Sellers are not obliged to deposit their Shares pursuant to section 4.1 or, if the Sellers have deposited or caused to be deposited their Shares pursuant to section 4.1, the Sellers may withdraw their Shares from the Offer prior to their Shares being taken up under the terms of the Offer, if (i) the Sellers receive a bona fide offer from an arms—length third party to acquire, directly or indirectly, any or all of the Sellers—Shares or any other securities of the Corporation held by the Sellers (which offer may include the granting of an option to acquire any of such Shares or any joint ownership arrangement with a third party), or (ii) an Acquisition Proposal (as defined below) is announced, which offer or proposal the board of directors of Brascan, in its sole discretion, considers would, if consummated, result in a transaction more favourable to Brascan than the Issuer Bid.

# ARTICLE 5 TERMINATION BY THE SELLER AND THE CORPORATION

- **5.1** <u>Termination by Seller</u>. Either Seller, when not in default in performance of its obligations under this Agreement, may, without prejudice to any other rights, terminate this Agreement by written notice to the Corporation if:
  - 5.1.1 any of the circumstances set out in section 1.2 occur;

-

#### **Table of Contents**

- 5.1.2 Shares deposited under the Offer (including Shares that it deposited) have not, for any reason whatsoever, been taken up and paid for on or before 60 days after the date of the Offer;
- 5.1.3 the Corporation s representations and warranties in section 2.2 are untrue in any material respect; or
- 5.1.4 the Corporation has not complied in any material respect with any of its covenants to the Sellers in section 3.2.
- **5.2** Termination by the Corporation. The Corporation, when not in default in performance of its obligations under this Agreement, may, without prejudice to any other rights, terminate this Agreement by written notice to the Sellers if:
  - 5.2.1 any of the circumstances set out in section 1.2 occur; or
  - 5.2.2 either Seller s representations and warranties in section 2.1 are untrue in any material respect; or
  - 5.2.3 either Seller has not complied in any material respect with any of its covenants to the Corporation in section 3.1.
- **5.3** Effect of Termination. In the case of any termination of this Agreement pursuant to this Article 5, this Agreement shall be of no further force and effect. Such termination shall not relieve any party from liability for any breach of this Agreement prior to such termination.

## ARTICLE 6 GENERAL

#### 6.1 Survival of Representations and

<u>Warranties</u>. The representations and warranties shall terminate upon the exchange by the Corporation of the Shares deposited by the Sellers pursuant to section 4.1 and in accordance with the terms of the Offer.

applicable laws or regulations, or as required by any competent government, judicial or other authority, or in accordance with the requirements of any stock exchange, neither the Corporation, on the one hand, nor the Sellers, on the other hand, shall make any public announcement or statement with respect to this Agreement without the approval of the Sellers or the Corporation, as the case may be, which approval shall not be unreasonably withheld. Moreover,

the parties agree to consult with each other prior to issuing each public announcement or statement with respect to this Agreement.

- **6.3** Assignment. This Agreement shall not be assignable by any party hereto without the consent of the other party.
- **6.4** <u>Time</u>. Time shall be of the essence of this Agreement.

6

#### **Table of Contents**

- **6.5** <u>Currency</u>. All sums of money referred to in this Agreement shall mean Canadian funds, unless otherwise indicated.
- **6.6 Governing Law**. This Agreement shall be governed by and construed in accordance with the laws of the Province of Ontario and the federal laws of Canada applicable therein.
- **Entire Agreement**. This Agreement constitutes the entire agreement and understanding between and among the parties hereto with respect to the subject matter hereof and supercedes any prior agreement, representation or understanding with respect thereto. For greater certainty there are no representations, warranties, undertakings or any other matter of any nature whatsoever as between the parties except as contained herein.
- **6.8 Date.** This Agreement shall be dated as of the date on which it is executed by the Sellers.
- **6.9** Amendments. This Agreement may not be modified, amended, altered or supplemented except upon the execution and delivery of a written agreement executed by all of the parties hereto.
- **6.10 Definitions**. For the purposes of this Agreement, and in addition to other terms defined herein, the terms:
  - 6.10.1 **Acquisition Proposal** means any merger, acquisition, take-over bid, sale of material assets (or any lease, long-term supply agreement or other arrangement having the same economic effect as a sale of material assets), liquidation, material sale of shares or rights or interests therein or thereto, or similar transactions involving the Corporation or any material subsidiary, or a proposal to do so, excluding the Insider Bid.
  - 6.10.2 **Business Day** means any day on which banks in the City of Toronto, Ontario are open for business.
  - 6.10.3 **Date of Deposit** means the third Business Day after the date of the Offer.
  - 6.10.4 **Encumbrance**, for the purpose of sections 2.1.1 and 4.1, means any option, security interest or title retention agreement of any nature or kind, or any grant of, or agreement to grant, any proxy or other right to vote, or any voting trust, vote pooling or other agreement with respect to the right to vote, call meetings of shareholders or give consents or approvals of any kind.
  - 6.10.5 **Junior Preferred Shares** has the meaning attributed to that term in section 3.2.5.
  - 6.10.6 **Senior Preferred Shares** has the meaning attributed to that term in section 3.1.3.
  - 6.10.7 **Shares** shall include any shares into which the Shares may be reclassified, subdivided, consolidated or converted and any rights and benefits arising therefrom including any extraordinary distributions of securities which may be declared in respect

7

### **Table of Contents**

of the Shares and shall also include any Shares issued pursuant to existing rights to purchase Shares under stock option, stock purchase and other plans. For greater certainty, **Shares** shall not include any shares which are issuable upon the exercise by either of the Sellers of its conversion rights attached to the adjustable rate convertible debentures of the Corporation unless and until such conversion rights are exercised.

6.10.8 **Wholly Owned Subsidiaries** means any and all entities, all the participating securities of which are owned by Brascan or Brascade, as applicable, directly or indirectly through other Wholly Owned Subsidiaries.

For the purpose of this Agreement, if the last day of a period of days is not a Business Day, the period shall be extended to the next following day which is a Business Day.

- Agreement is an integral part of the transactions contemplated in the Offer, that the Corporation would not make the Offer and the Sellers would not agree to their covenants to the Corporation herein and to deposit the Sellers Shares in accordance with section 4.1 unless this Agreement was executed, and accordingly acknowledges and agrees that a breach by a party of any covenants or other commitments contained in this Agreement will cause the other party to sustain injury for which it would not have an adequate remedy at law for money damages. Therefore, each of the parties agrees that in the event of any such breach the aggrieved party shall be entitled to the remedy of specific performance of such covenants or commitments and preliminary and permanent injunctive and other equitable relief in addition to any other remedy to which it may be entitled, at law or in equity, and the parties further agree to waive any requirement for the securing or posting of any bond in connection with the obtaining of any injunctive or other equitable relief.
- **6.12** Notices. Any notice, request, consent, agreement or approval which may or is required to be given pursuant to this Agreement shall be in writing and shall be sufficiently given or made if delivered or telecopied in the case of:
  - (a) the Corporation, addressed as follows:

Noranda Inc. 181 Bay Street Suite 200, BCE Place Toronto, Ontario M5J 2T3

**Attention:** General Counsel

Telecopier: (416) 982-7111 Telephone: (416) 982-7423

with a copy to:

McCarthy Tétrault LLP

8

Toronto Dominion Tower Suite 4700 Toronto, Ontario M5K 1E6

**Attention:** Garth M. Girvan

Telecopier: (416) 868-0673 Telephone: (416) 362-1812

(b) the Sellers, addressed as follows:

Brascan Corporation 181 Bay Street Suite 300, Box 762 Toronto, Ontario M5J 2T3

**Attention:** Jeffrey M. Blidner

Telecopier: (416) 363-9491 Telephone: (416) 363-2856

with a copy to:

Torys LLP 79 Wellington St. W. Suite 3000, Box 270 Toronto, Ontario M5K 1N2

**Attention:** Brian Davis

Telecopier: (416) 865-7380 Telephone: (416) 865-7500

or to such other address as the relevant party may from time to time advise by notice in writing given pursuant to this section. The date of receipt of any such notice, request, consent, agreement, or approval shall be deemed to be the date of delivery or telecopy (if during normal business hours, or, if not, the next Business Day).

**6.13** Expenses. Each of the parties shall pay its legal, financial advisory and accounting costs and expenses incurred in connection with the preparation, execution and delivery of this Agreement and all documents and instruments executed or prepared pursuant hereto and any other costs and expenses whatsoever and by whosoever incurred.

## **Table of Contents**

**6.14** Counterparts. This Agreement may be executed in one or more counterparts which together shall be deemed to constitute one valid and binding agreement and delivery of the counterparts may be effected by means of a telecopied transmission.

10

### **Table of Contents**

If the terms and conditions of this letter are acceptable to you, please so indicate by executing and returning the enclosed copy hereof to the undersigned on or before March 8, 2005, failing which this Agreement shall be null and void.

Yours truly,

### NORANDA INC.

By: << Derek G. Pannell>>
Name: Derek G. Pannell

Title: President and Chief Executive

Officer

Agreed and accepted March 8, 2005.

## **BRASCAN CORPORATION**

By: << *Jeffrey M. Blidner*>> Name: Jeffrey M. Blidner Title: Managing Partner

Agreed and accepted March 8, 2005.

## **BRASCADE CORPORATION**

By: << Bryan K. Davis>> Name: Bryan K. Davis

Title: Vice-President and Treasurer

11

# SCHEDULE A Terms of the Issuer Bid

#### 1. General Terms.

The Offer shall be made by issuer bid circular to all common shareholders of the Corporation, prepared in compliance with the *Securities Act* (Ontario) ( **OSA** ) and other applicable securities laws. The Offer shall be made on the terms herein set forth and upon such other terms and conditions as required by law and shall be open for an initial period of at least thirty-five (35) days (calculated in accordance with the OSA) or such longer period as may be required to satisfy all of the conditions set forth in paragraph 3 below.

Upon the terms and subject to the conditions of the Offer, the Corporation will exchange up to 63,377,140 Shares deposited and not withdrawn under the Offer as required by applicable securities laws, subject to proration, in exchange for US\$1.25 billion aggregate principal amount of Preferred Shares. The Preferred Shares will consist of three series of junior preferred shares in tranches of US\$500 million, US\$500 million and US\$250 million. If the Preferred Shares are created out of the existing class of Senior Preferred Shares, the three series will be designated as Series J, Series K and Series L. If the Preferred Shares are created out of the new class of Junior Preferred Shares, the three series will be designated as Series 1, Series 2 and Series 3. The terms of the Preferred Shares are set out in Exhibit A.

Unless the Offer is withdrawn, Shares may be deposited pursuant to the Offer at any time prior to the close of business on the date of the first take-up of or payment for Shares. Any Shares deposited pursuant to the Offer may be withdrawn until exchanged.

- 2. **Price of the Issuer Bid**. The Offer shall be made on the basis of approximately 0.789 Preferred Shares (proportionately from each series of Preferred Shares to be issued, but without the requirement to issue fractional shares; entitlements to fractional shares will be paid in cash) per Share.
- 3. <u>Conditions of the Issuer Bid</u>. The Offer shall be subject to the conditions set out in Exhibit B.

12

# EXHIBIT A

Preferred Share Class Conditions and Series Terms

## Edgar Filing: NORANDA INC - Form SC 13D/A

## **Table of Contents**

The rights, privileges, restrictions and conditions attaching to the Junior Preference Shares are as follows:

- (a) **Series**: The Junior Preference Shares may at any time or from time to time be issued in one or more series. Subject to the following provisions, the board of directors of the Corporation may from time to time before the issue thereof fix the number of shares in, and determine the designation, rights, privileges, restrictions and conditions attaching to the shares of, each series of Junior Preference Shares.
- (b) **Preference**: The Junior Preference Shares are entitled to priority over the common shares and all other shares ranking junior to the Junior Preference Shares with respect to the payment of dividends and the distribution of assets of the Corporation in the event of any liquidation, dissolution or winding up of the Corporation or other distribution of assets of the Corporation among its shareholders for the purpose of winding up its affairs.
- (c) **Idem**: The Junior Preference Shares will rank junior to the Preferred Shares with respect to priority in the payment of dividends and in the distribution of assets of the Corporation in the event of any liquidation, dissolution or winding up of the Corporation or other distribution of assets of the Corporation among its shareholders for the purpose of winding up its affairs.
- (d) **No Priority**: The Junior Preference Shares of each series will rank on a parity with the Junior Preference Shares of every other series with respect to priority in the payment of dividends and in the distribution of assets in the event of any liquidation, dissolution or winding up of the Corporation or other distribution of assets of the Corporation among its shareholders for the purpose of winding up its affairs.
- (e) **Restrictions**: As long as any of the Preferred Shares are outstanding, the Corporation will not at any time, without the approval of the holders of each series of Preferred Shares then outstanding, given in accordance with the terms of the Preferred Shares:
  - (i) declare, pay or set apart for payment any dividends (other than stock dividends payable in shares of the Corporation) on the Junior Preference Shares;
  - (ii) redeem or call for redemption, purchase or otherwise pay off, retire or make any return of capital in respect of any series of Junior Preference Shares,

if such action is prohibited by the terms of the Preferred Shares or the terms of any series of Preferred Shares that are then outstanding.

(f) **Participation upon Liquidation, Dissolution or Winding Up**: In the event of the liquidation, dissolution or winding up of the Corporation or any other distribution of assets of the Corporation among its shareholders for the purpose of

## Edgar Filing: NORANDA INC - Form SC 13D/A

## **Table of Contents**

winding up its affairs, the holders of the Junior Preference Shares will be entitled to payment of an amount equal to the amount paid up on such shares in the case of any liquidation, dissolution, winding up or other distribution which is involuntary, and to payment of an amount equal to the amount paid up thereon plus the premium on redemption applicable at the date thereof, if any, if the same is voluntary, together in all cases with all unpaid dividends accrued thereon (which will for such purpose be treated as accruing up to the date of distribution), the whole before any amount is paid or any assets of the Corporation distributed to the holders of any Common Shares or shares of any other class ranking junior to the Junior Preference Shares but the whole subject to the rights of the holders of any other class of shares of the Corporation entitled to receive the assets of the Corporation upon such distribution in priority to or rateably with the holders of the Junior Preference Shares. Upon payment to the holders of the Junior Preference Shares of the amount so payable to them, they will not be entitled to share in any further distribution of assets of the Corporation.

1A

# Number and Designation of, and Rights, Privileges, Restrictions and Conditions Attaching to, the Junior Preference Shares, Series 1

The first series of Junior Preference Shares of Noranda shall consist of 20,000,000 shares designated as Junior Preference Shares, Series 1 (the <u>Series 1 Shares</u>) with an issue price of \$25.00 per share (the <u>Issue Price</u>) and, in addition to the rights, privileges, restrictions and conditions attaching to the Junior Preference Shares as a class, shall have attached thereto the following rights, privileges, restrictions and conditions (the <u>Series 1 Provisions</u>):

## 1. Interpretation

(1) **Defined Terms.** In these Series 1 Provisions:

**Acquiring Person** means any person who is the Beneficial Owner of a Significant Interest in the outstanding Voting Shares; provided, however, that the term Acquiring Person shall not include:

- (a) Noranda or any Subsidiary of Noranda;
- (b) Brascan:
- (c) any person who becomes the Beneficial Owner of a Significant Interest in the outstanding Voting Shares as a result of one or any combination of:
  - (A) a Voting Share Reduction which, by reducing the number of Voting Shares outstanding, increases the percentage of Voting Shares Beneficially Owned by such person to a Significant Interest in the Voting Shares then outstanding,
  - (B) a Pro Rata Acquisition,
  - (C) an Exempt Acquisition,
  - (D) a Convertible Security Acquisition, or
  - (E) a 123 Share Acquisition;

provided further, however, that if a person shall become the Beneficial Owner of a Significant Interest in the Voting Shares then outstanding by reason of any one or a combination of a Voting Share Reduction, a Pro Rata Acquisition, an Exempt Acquisition, a Convertible Security Acquisition or a 123 Share Acquisition and thereafter becomes the Beneficial Owner of more than an additional 1% of the outstanding Voting Shares (other than pursuant to a Voting Share Reduction, a Pro Rata Acquisition, an Exempt Acquisition, a Convertible Security Acquisition or a 123 Share Acquisition), then as of the date that such person becomes the Beneficial Owner of such additional Voting Shares, such person shall become an Acquiring Person;

(d) any person who, on the Issue Date of the 123 Shares, is the Beneficial Owner of a Significant Interest in the outstanding Voting Shares, provided that if such person thereafter becomes the Beneficial Owner of more than an additional 1% of the outstanding Voting Shares (other than pursuant to a Voting Share Reduction, a Pro Rata Acquisition, an Exempt Acquisition, a Convertible Security Acquisition or a 123 Share Acquisition), then as of the date that such person becomes the Beneficial Owner of such additional Voting Shares, such person shall become an Acquiring Person; and

1B

(e) an underwriter or member of a banking or selling group that acquires Voting Shares from Noranda in connection with a distribution of securities pursuant to a prospectus or by way of private placement.

Adjusted Net Worth means, as at the end of any fiscal quarter, the aggregate value of:

- (a) the outstanding share capital for all shares ranking junior to the Junior Preference Shares;
- (b) without duplication, any surplus, whether contributed or capital;
- (c) retained earnings; and
- (d) consolidated non-controlling interest, all as set forth in Noranda s consolidated balance sheet for such fiscal quarter filed in accordance with applicable securities laws.

**Affiliate**, when used to indicate a relationship with a specified corporation, means a person that directly, or indirectly through one or more controlled intermediaries, controls, or is controlled by, or is under common control with, such specified corporation.

**Associate** , when used to indicate a relationship with a specified person, means (i) a spouse of such specified person, (ii) any person of either sex with whom such specified person is living in a conjugal relationship outside marriage or (iii) any relative of such specified person or of a person mentioned in clauses (i) or (ii) of this definition if that relative has the same residence as the specified person.

**Basic Redemption Price** has the meaning attributed to it in subsection 3(2).

A person shall be deemed the **Beneficial Owner**, and to have **Beneficial Ownership** of and to **Beneficially Own**:

- (i) any securities of which such person or any of such person s Affiliates or Associates is the owner at law or in equity;
- (ii) any securities as to which such person or any of such person s Affiliates or Associates has the right to become the owner at law or in equity, where such right is exercisable immediately or within 60 days of the date of the determination of

Beneficial Ownership and whether or not on condition or the occurrence of any contingency or payment of instalments, upon the exercise of any conversion right, exchange right or purchase right attaching to Convertible Securities, or pursuant to any agreement, arrangement, pledge or understanding, written or oral (other than customary agreements with and between underwriters and/or banking group and/or selling group members with respect to a distribution of securities pursuant to a prospectus or by way of private placement and other than pursuant to pledges of securities in the ordinary course of business); and

1C

- (iii) any securities which are Beneficially Owned within the meaning of clauses (i) or (ii) of this definition by any other person with which, and in respect of which securities, such person is acting jointly or in concert;
  - provided, however, that a person shall not be deemed the Beneficial Owner of, or to have Beneficial Ownership of, or to Beneficially Own, any security:
  - (A) by reason of such security having been deposited or tendered pursuant to a Take-over Bid made by, or Control Transaction involving, such person or any of such person s Affiliates or Associates or any other person referred to in clause (iii) of this definition until the earlier of such deposited or tendered security being accepted unconditionally for payment or exchange or being taken up and paid for;
  - (B) by reason of such person, any of such person s Affiliates or Associates or any other person referred to in clause (iii) of this definition holding such security, if:
    - (I) the ordinary business of the person (in this definition, a <u>Manager</u>) includes the management of investment funds for others and such security is held by the Manager in the ordinary course of such business in the performance of such Manager s duties for the account of any other person (in this definition, a <u>Client</u>) including non-discretionary accounts held on behalf of a Client by a broker or dealer registered under applicable law;
    - (II) the person (in this definition, a <u>Trust Company</u>) is licensed to carry on the business of a trust company under applicable law and, as such, acts as a trustee or administrator or in a similar capacity in relation to the estates of deceased or incompetent persons (each, in this definition, an Estate Account ) or in relation to other accounts (each, in this definition, an Other Account ) and holds such security and is acting in the ordinary course of such duties for the Estate Account or for such Other Accounts;
    - (III) such person is a Crown agent or agency (in this definition, the <u>Crown Agent</u>);

1D

- (IV) the person is established by statute for purposes that include, and the ordinary business or activity of such person (in this definition, a <u>Statutory Body</u>) includes, the management of investment funds for employee benefit plans, pension plans, insurance plans of various public bodies and the Statutory Body holds such security for the purposes of its activities as such; or
- (V) the person (in this definition, an <u>Administrator</u>) is the administrator or trustee of one or more pension funds or plans (each, in this definition, a <u>Plan</u>) registered under the laws of Canada or any province thereof or the similar laws of the jurisdiction by which such Plan is governed or is such a Plan and the Administrator or Plan holds such security for the purposes of its activities as such;

but only if the Manager, the Trust Company, the Crown Agent, the Statutory Body, the Administrator or the Plan, as the case may be, is not then making and has not announced a current intention to make a Take-over Bid or a Control Transaction, other than an Offer to Acquire Voting Shares or other securities pursuant to a distribution by Noranda, or by means of ordinary market transactions (including prearranged trades entered into in the ordinary course of business of such person) executed through the facilities of a stock exchange or an organized over-the-counter market, alone or by acting jointly or in concert with any other person;

## (C) because such person:

- (I) is a Client of the same Manager as another person on whose account the Manager holds such security,
- (II) has an Estate Account or an Other Account with the same Trust Company as another person on whose account the Trust Corporation holds such security, or
- (III) is a Plan with the same Administrator as another Plan on whose account the Administrator holds such security,

## (D) because such person:

- (I) is a Client of a Manager and such security is owned at law or in equity by the Manager,
- (II) has an Estate Account or an Other Account with a Trust Company and such security is owned at law or in equity by the Trust Company, or
- (III) is a Plan and such security is owned at law or in equity by the Administrator of the Plan; or

1E

(E) because such person is the registered holder of securities as a result of carrying on the business of, or acting as nominee for, a securities depositary.

**Board Event** means any one of the following events:

- (a) if four quarterly dividends on the Series 1 Shares are in arrears, whether or not such dividends have been declared and whether or not there are any monies of Noranda properly applicable to the payment of dividends; or
- (b) if the Adjusted Net Worth for any fiscal quarter is less than \$2,500,000,000.

**Board of Directors** means the board of directors of Noranda.

**Brascan** means Brascan Corporation, any Affiliate of Brascan Corporation or any party acting jointly or in concert with Brascan Corporation or any of its Affiliates.

**Business Day** means a day other than a Saturday, a Sunday or any other day which is a statutory or civic holiday in Toronto, Ontario.

**Cash Conversion Price** has the meaning attributed to it in subsection 5(1).

## **Change of Control Event** means the occurrence of:

- (i) the first date of public announcement (which, for purposes of this definition, shall include, without limitation, a report filed pursuant to section 101 of the *Securities Act* (Ontario)) by Noranda or an Acquiring Person of facts indicating that an Acquiring Person has become such;
- (ii) the date of the commencement of, or first public announcement of the intent of any person (other than Brascan, Noranda or any Subsidiary of Noranda) to commence, a Take-over Bid (other than by Brascan, Noranda or any Subsidiary of Noranda); or
- (iii) the date of the commencement of, or first public announcement of the intent of any person (other than Brascan) to commence a Control Transaction (other than by Brascan, Noranda or any Subsidiary of Noranda).

**Change of Control Retraction Date** means the date on which the Series 1 Shares are redeemed pursuant to subsection 4(2)(b)(i), (ii) or (iii), as applicable.

**Common Shares** means the common shares of Noranda, and any other securities into which those shares may be changed or for which those shares may be exchanged (whether or not Noranda is the issuer of such other securities).

**Control Transaction** means any merger, arrangement, amalgamation, reorganization, acquisition, material sale of shares or rights or interests therein or thereto or similar transactions involving Noranda, which will result in any person Beneficially Owning, whether directly as a result of such transaction or together with the Common Shares that such person Beneficially Owned before the completion of such transaction, a Significant Interest in Noranda.

1F

#### **Table of Contents**

**controlled**: a body corporate is controlled by another person if:

- (i) securities entitled to vote in the election of directors carrying more than 50% of the votes for the election of directors are held, directly or indirectly, by or on behalf of the other person; and
- (ii) the votes carried by such securities are entitled, if exercised, to elect a majority of the board of directors of such body corporate,

and controls, controlling and under common control with shall be interpreted accordingly.

**Conversion Notice** has the meaning attributed to it in subsection 5(1).

**Conversion Right** has the meaning attributed to it in subsection 5(1).

Convertible Securities means, at any time, any securities issued by Noranda (including rights, warrants and options but other than the 123 Shares) carrying any purchase, exercise, conversion or exchange right, pursuant to which the holder thereof may acquire Voting Shares or other securities convertible into or exercisable or exchangeable for Voting Shares (in each case, whether such right is exercisable immediately or after a specified period and whether or not on condition or the happening of any contingency).

**Convertible Security Acquisition** means the acquisition of Voting Shares upon the exercise of Convertible Securities acquired by a person pursuant to an Exempt Acquisition or a Pro Rata Acquisition.

Current Market Price means, at any date, the volume weighted average trading price of Common Shares on the Toronto Stock Exchange (or, if not then listed on that exchange, on another exchange or market chosen by the Board of Directors in their sole discretion on which Common Shares are then traded), for a period of 20 consecutive Trading Days ending on the fourth day immediately prior to that date or, if such fourth day is not a Trading Day, on the immediately preceding Trading Day.

**Dividend** and **Dividends** have the respective meanings attributed to them in subsection 2(1)(a).

**Dividend Payment Date** means the last day of the months of March, June, September and December in each year.

**Dividend Period** means the Initial Dividend Period and, thereafter, the period from and including a Dividend Payment Date to but excluding the next succeeding Dividend Payment Date.

**Exempt Acquisition** means an acquisition of Voting Shares or Convertible Securities (i) pursuant to a distribution of Voting Shares or Convertible Securities (and the conversion or exchange of such Convertible Securities) made by Noranda pursuant to a prospectus or private placement provided that the person does not acquire a greater percentage of the securities offered in the distribution than the percentage of Voting Shares Beneficially Owned by that person

# Edgar Filing: NORANDA INC - Form SC 13D/A

immediately prior to the distribution, or (ii) pursuant to an amalgamation, merger or other statutory procedure requiring shareholder approval.

## **Extraordinary Dividend** means:

- (i) any dividend, other than a stock dividend paid wholly in Common Shares, declared or paid on the Common Shares that is, when taken together with the amount or value of all other dividends declared or paid in the 12 month period preceding the date of declaration of the dividend (the measurement period) more than 125% of the aggregate amount or value of the dividends declared or paid on the Common Shares, other than stock dividends paid wholly in Common Shares, during the 12 month period ended on the day prior to the measurement period (or, where there were no dividends paid in the 12 month period ended on the day prior to the measurement period, the period of 12 months ending on the date on which Noranda last paid a dividend, other than a stock dividend paid wholly in Common Shares, on the Common Shares); or
- (ii) any dividend, other than a stock dividend paid wholly in Common Shares, declared or paid on the Common Shares that results in the Noranda Annual Dividend Rate exceeding a 10% compound annual growth rate, measured from and following the Issue Date. For this purpose, the Noranda Annual Dividend Rate is initially the aggregate amount or value of all dividends declared or paid on the Common Shares, other than stock dividends paid wholly in Common Shares, in the 12 month period immediately preceding the Issue Date and thereafter is the aggregate amount or value of all dividends declared or paid in any 12 month period immediately preceding the date of declaration of any other dividend on the Common Shares, together with the dividend then being declared; or
- (iii) any other special dividend on, or distribution with respect to, the Common Shares which is, by its terms or declared intent, declared and paid outside the normal operations or normal dividend procedures of Noranda.

  Final Redemption Date means the day that is five years from the Issue Date.

**freely-tradeable** means, in respect of a Common Share, that (i) the Common Share is transferable and not subject to any escrow requirements or to any cease trade order imposed by a Canadian securities regulatory authority, and (ii) all hold periods imposed by Canadian securities legislation before such Common Share can be traded without a prospectus or in reliance on a prospectus exemption have expired.

**Initial Dividend Period** means the period from and including the Issue Date to but excluding the first Dividend Payment Date after the Issue Date.

**Issue Date** means the date any of the Series 1 Shares are first issued.

**Issue Price** has the meaning attributed to it in the introductory paragraph to these rights, privileges, restrictions attaching to the Series 1 Shares.

**Noranda** means Noranda Inc. and any corporation resulting, surviving or continuing from the merger, arrangement, amalgamation or reorganization of Noranda Inc., or a comparable transaction.

Table of Contents 47

1G

1H

## **Offer to Acquire** shall include:

- (i) an offer to purchase, or a solicitation of an offer to sell, Voting Shares, and
- (ii) an acceptance of an offer to sell Voting Shares, whether or not such offer to sell has been solicited, or any combination thereof, and the person accepting an offer to sell shall be deemed to be making an Offer to Acquire to the person that made the offer to sell.

**Offeror** means a person who has announced an intention to make or who has made a Take-over Bid (excluding an Offer to Acquire Common Shares or other securities made by a Manager, Trust Company, Crown Agent, Statutory Body, Administrator or Plan referred to in clause (B) of the definition of Beneficial Owner pursuant to a distribution by Noranda or by means of ordinary market transactions (including pre-arranged trades entered into in the ordinary course of business of such person) in the circumstances contemplated in said clause (B)), but only so long as the Take-over Bid so announced or made has not been withdrawn or terminated or has not expired.

**Offeror s Securities** means the aggregate of the Voting Shares Beneficially Owned by an Offeror on the date of an Offer to Acquire by an Offeror.

**123 Shares** means, collectively, the Series 1 Shares, the Series 2 Shares and the Series 3 Shares.

**123 Share Acquisition** means, in respect of holders of the 123 Shares, (i) the payment of a Dividend by the issuance of Common Shares pursuant to subsection 2(2)(a) or (ii) the issuance of Common Shares upon exercise by Noranda of its conversion right pursuant to section 5.

**Optional Redemption Price** has the meaning attributed to it in subsection 3(1).

**Pro Rata Acquisition** means an acquisition by a person of Voting Shares or Convertible Securities (i) as a result of a stock dividend, a stock split or other event pursuant to which such person receives or acquires Voting Shares or Convertible Securities on the same pro rata basis as all other holders of Voting Shares of the same class or series of Noranda, (ii) pursuant to a regular dividend reinvestment or other plan of Noranda made available by Noranda to the holders of Voting Shares where such plan permits the holder to direct that the dividends paid in respect of such Voting Shares be applied to the purchase from Noranda of further securities of Noranda, or (iii) pursuant to the receipt and/or exercise of rights (other than the 123 Shares) issued by Noranda to all of the holders of a series or class of Voting Shares on a pro rata basis to subscribe for or purchase Voting Shares or Convertible Securities, provided that the person does not acquire a greater percentage of the securities issuable on exercise of such rights than the percentage of Voting Shares Beneficially Owned by that person immediately prior to the commencement of the offering of rights and that such rights are acquired directly from Noranda and not from any other person.

**Redemption Date** means (i) in the case of a mandatory redemption pursuant to subsection 3(3), the Final Redemption Date, (ii) in the case of a redemption in connection with an exercise of the Retraction Right, the Change of Control Retraction Date, and (iii) in all other cases, the date on which the redemption is to occur.

**Redemption Notice** has the meaning attributed to it in subsection 3(5).

**Redemption Price** means (i) if the redemption occurs on or before June 30, 2008, the Optional Redemption Price, and (ii) if the redemption occurs after June 30, 2008, the Basic Redemption Price.

**Retraction Price** has the meaning attributed to it in subsection 4(2)(b).

**Retraction Right** has the meaning attributed to it in subsection 4(1).

**Series 1 Provisions** has the meaning attributed to it in the introductory paragraph to these rights, privileges, restrictions attaching to the Series 1 Shares.

**Series 1 Shares** has the meaning attributed to it in the introductory paragraph to these Series 1 Provisions.

**Series 123 Directors** means, at any time, those directors of Noranda most recently elected by or appointed on behalf of the holders of the Series 1 Shares, Series 2 Shares and Series 3 Shares, collectively.

**Series 2 Shares** means the second series of Junior Preference Shares of Noranda, designated as Junior Preference Shares, Series 2.

**Series 3 Shares** means the third series of Junior Preference Shares of Noranda, designated as Junior Preference Shares, Series 3.

**Significant Interest**: a person has a Significant Interest in the Voting Shares where that person Beneficially Owns 30% or more of the outstanding Voting Shares.

**Subsidiary**: a body corporate is a Subsidiary of another body corporate if:

- (i) it is controlled by (A) that other, or (B) that other and one or more bodies corporate, each of which is controlled by that other, or (C) two or more bodies corporate, each of which is controlled by that other; or
- (ii) it is a Subsidiary of a body corporate that is that other s Subsidiary.

**Take-over Bid** means an Offer to Acquire Voting Shares, where the Voting Shares subject to the Offer to Acquire and the Offeror's Securities, constitute a Significant Interest in the outstanding Voting Shares at the date of the Offer to Acquire.

**Tax Act** means the *Income Tax Act* (Canada).

**Trading Day** means with respect to any stock exchange or over-the-counter market, a day on which shares may be traded through the facilities of such stock exchange or in such over-the-counter market, and otherwise means a day on which shares may be traded through the facilities of the principal stock exchange on which the Common Shares are listed (or, if the Common Shares are not listed on any stock exchange, then in the over-the-counter market).

Table of Contents 49

1I

1J

**Transfer Agent** means Noranda, or such other person as from time to time may be the registrar and transfer agent for the Series 1 Shares.

**Transferee** and **Transferees** have the meanings attributed to them in subsection 5(2).

- **U.S.** Canadian Exchange Rate means, on any date:
  - (i) if, on such date, the Bank of Canada sets an average noon spot rate of exchange for the conversion of one Canadian dollar into United States dollars, such rate; or
  - (ii) in any other case, the rate for such date for the conversion of one Canadian dollar into United States dollars calculated in such manner as may be determined by the Board of Directors from time to time acting in good faith
- **U.S. Dollar Equivalent** of any amount which is expressed in Canadian dollars means, on any date, the United States dollar equivalent of such amount determined by multiplying such amount by the U.S. Canadian Exchange Rate in effect on such date.

**Voting Share Reduction** means an acquisition or a redemption by Noranda of Voting Shares.

**Voting Shares** means collectively the Common Shares and any other shares in the capital stock or voting interests of Noranda entitled to vote generally in the election of directors.

- (2) **Ranking of Shares**. The expressions in priority to , on a parity with , ranking equally with , ranking prior to a ranking junior to and similar expressions refer to the order of priority only in payment of dividends or in the distribution of assets in the event of any liquidation, dissolution of winding-up of Noranda, whether voluntary or involuntary, or any other distribution of the assets of Noranda for the purpose of winding-up its affairs.
- (3) **Holder.** References to a <u>holder</u> in these Series 1 Provisions mean the registered owner of such shares.
- (4) **Extended Meanings.** References in these Series 1 Provisions to words importing the singular number only include the plural and *vice versa*, words importing any gender include all genders and words importing persons include individuals, corporations, limited and unlimited liability companies, general and limited partnership, associations, trusts, unincorporated organizations, joint ventures and governmental authorities. The term including means including without limiting the generality of the foregoing .
- (5) **References to Statutes.** Except as otherwise noted herein, reference to any statute in these Series 1 Provisions is to that statute as in force from time to time, including any regulations, rules, policy statements or guidelines made under that statute, and includes any statute which may be enacted in substitution of that statute.
- (6) **References to Sections**. Reference in these Series 1 Provisions to any section or subdivision is to the designated section or subdivision of these Series 1 Provisions.

1K

(7) Acting Jointly or in Concert. For purposes of these Series 1 Provisions, a person is acting jointly or in concert with every other person who is a party to any agreement, commitment or understanding (whether formal or informal and whether or not in writing) with the first mentioned person to acquire or offer to acquire or vote Voting Shares (other than (i) in the case of agreements to acquire Voting Shares, customary agreements with and between underwriters and/or banking group and/or selling group members with respect to a distribution of securities pursuant to a prospectus or by way of private placement and other than pursuant to pledges of securities in the ordinary course of business and (ii) in the case of agreements to vote Voting Shares, agreements to vote for a limited purpose and without the intention of acquiring or exercising control).

## (8) Other Payment Matters.

- (a) If any date on which any Dividend on the Series 1 Shares is payable or on or by which any other action is required to be taken by Noranda under these Series 1 Provisions is not a Business Day, then the Dividend will be payable, or such other action will be required to be taken, on or by the next day that is a Business Day.
- (b) In the event of the non-receipt of a cheque by a holder of Series 1 Shares entitled to the cheque, or the loss or destruction of the cheque, Noranda, on being furnished with reasonable evidence of non-receipt, loss or destruction, and an indemnity reasonably satisfactory to Noranda, will issue to the holder a replacement cheque for the amount of the original cheque.
- (c) Notwithstanding any other provision of these Series 1 Provisions, Noranda may deduct or withhold from any payment, distribution, issuance or delivery to be made pursuant to these share terms any amounts required or permitted by law to be deducted or withheld from any such payment, distribution, issuance or delivery and will remit such amounts to the relevant tax authority as required. Where the cash component of any payment, distribution, issuance or delivery to be made pursuant to these share terms is less than the amount that Noranda is so required or permitted to deduct or withhold, Noranda will be permitted to deduct and withhold from any non-cash payment, distribution, issuance or delivery to be made pursuant to these share terms any amounts required or permitted by law to be deducted or withheld from any such payment, distribution, issuance or delivery and to dispose of such property in order to remit any amount required to be remitted to any relevant tax authority. Notwithstanding the above, the amount of any payment, distribution, issuance or delivery made pursuant to these Series 1 Provisions will be considered to be the amount of the payment, distribution, issuance or delivery plus any amount deducted or withheld pursuant to this subsection 1(8)(c).

## (9) Currency.

Except as otherwise provided in these Series 1 Provisions, all references to currency herein are to lawful money of the United States.

## (10) Designation Under the Tax Act.

1L

The amount of is specified in respect of each Series 1 Share for the purposes of subsection 191(4) of the Tax Act.

## 2. Dividends

## (1) Payment of Dividends.

- (a) The holders of Series 1 Shares will be entitled to receive and Noranda will pay thereon, if, as and when declared by the Board of Directors out of the assets of Noranda properly applicable to the payment of dividends, fixed preferential cumulative cash dividends (each a <u>Dividend</u> and collectively the <u>Dividends</u>) in an amount per Series 1 Share per annum equal to \$1.50, payable quarterly, in equal instalments of \$0.375 per Series 1 Share, with respect to each Dividend Period other than the Initial Dividend Period, on the Dividend Payment Date immediately following the end of each such Dividend Period.
- (b) The dividend payable in respect of the Initial Dividend Period, if declared by the Board of Directors, will be payable on June 30, 2005 in the amount per Series 1 Share obtained when:
  - (i) \$1.50 is multiplied by the number of days beginning on the Issue Date up to but excluding June 30, 2005;
  - (ii) the result in paragraph (i) divided by 365.
- (c) In any case where Dividends are payable for a period that ends on a date other than a Dividend Payment Date on which a Dividend was paid, Dividends will be paid in an amount per Series 1 Share obtained when:
  - (i) \$1.50 is multiplied by the number of days beginning on the date of the immediately preceding Dividend Payment Date up to but excluding the date fixed for payment; and
  - (ii) the result in paragraph (i) is divided by 365.
- (d) If on any Dividend Payment Date the dividend payable on such date is not paid in full on all the Series 1 Shares then issued and outstanding, such dividend, or the unpaid part thereof, will be paid at a subsequent date or dates determined by the Board of Directors on which Noranda has sufficient monies properly applicable to the payment of such dividends. The holders of Series 1 Shares will not be entitled to any dividends other than or in excess of the preferential cumulative dividends hereinbefore provided.

## (2) Method of Payment.

(a) Dividends on the Series 1 Shares will (other than in the case of redemption, in which case payment of dividends will be made on surrender of the certificate representing the Series 1 Shares to be redeemed) be paid by cheque, in lawful money of Canada,

1**M** 

payable at par at any branch in Canada of a chartered bank or trust company, in the amount, rounded to the nearest whole cent (\$0.01) of the applicable dividend or, at the election of Noranda and subject to applicable law and any requirement to obtain regulatory relief, by issuing freely-tradeable, fully paid and non-assessable Common Shares. The mailing by ordinary unregistered first class prepaid mail of such a cheque or share certificate, as applicable, to a holder of Series 1 Shares to the address of the holder as it appears on the register of holders, or, in the case of joint holders, to the address of that one whose name appears first in the register of holders as one of such joint holders, on or before the applicable Dividend Payment Date will be deemed to be payment and will satisfy and discharge all liabilities for dividends payable on that Dividend Payment Date to the extent of the amount represented by the cheque (plus any tax required to be deducted and withheld from the payment) or, calculated in accordance with subsection 2(2)(b), such share certificate unless, in case of payment by cheque, such cheque is not paid on due presentation. Dividends represented by a cheque which has not been duly presented for payment within two years after it was issued or that otherwise remain unclaimed for a period of two years from the date on which they were declared to be payable and set apart for payment will be forfeited to Noranda. Noranda may pay dividends in any other manner as it may agree with any particular holder.

- (b) If Noranda elects to pay a Dividend by issuing Common Shares in accordance with subsection 2(2)(a), the number of Common Shares to be issued as payment of a Dividend will be equal to the quotient obtained by dividing (i) the Dividend payable by (ii) 95% of the U.S. Dollar Equivalent of the Current Market Price as of the Dividend Payment Date, less the amount of any tax required to be deducted and withheld by Noranda. In any case where the aggregate number of Common Shares to be issued to a holder on payment of a Dividend pursuant to this section includes a fraction of a Common Share, Noranda, in lieu of delivering a fractional share, will pay by cheque an amount equal to the balance of the Dividend payable not otherwise satisfied by the delivery of Common Shares. Noranda cannot pay a Dividend by issuing Common Shares if one or more of the following events has occurred:
  - (i) the issuance or delivery of such Common Shares is prohibited pursuant to any agreement or arrangement entered into by Noranda to assure Noranda s solvency or continued operation;
  - (ii) the issuance or delivery of such Common Shares is prohibited by law or by any regulatory or other authority having jurisdiction over Noranda and acting in conformity with law;
  - (iii) for any reason beyond its control, Noranda is unable to issue Common Shares or is unable to deliver such Common Shares; or
  - (iv) such Common Shares are not listed on each exchange on which the Common Shares are then listed for trading.

## 3. Redemption

- (1) **Redemption On or Before June 30, 2008.** On or before June 30, 2008, but subject to section 7 and to applicable law, Noranda, may upon giving notice as hereinafter provided, redeem at any time the whole or from time to time any part of the then outstanding Series 1 Shares from, subject to subsection 3(4), any one or more of the holders thereof as the Board of Directors may in its sole discretion determine by payment of an amount in cash for each share to be redeemed of \$25.25, together with an amount equal to accrued and unpaid dividends up to but excluding the date fixed for redemption and calculated in accordance with subsection 2(1)(c) (the <u>Optional Redemption Price</u>).
- Redemption From July 1, 2008 to the Final Redemption Date. From and including July 1, 2008 to the Final Redemption Date, but subject to section 7 and to applicable law, Noranda, may upon giving notice as hereinafter provided, redeem at any time the whole or from time to time any part of the then outstanding Series 1 Shares from, subject to subsection 3(4), the holders thereof as the Board of Directors may in its sole discretion determine by payment of an amount in cash for each share redeemed of \$25.00, together with an amount equal to all accrued and unpaid dividends up to but excluding the date fixed for redemption and calculated in accordance with subsection 2(1)(c) (the <u>Basic Redemption Price</u>).
- (3) **Mandatory Redemption on the Final Redemption Date.** Subject to section 7 and to applicable law and to any notice of conversion given by Noranda pursuant to section 5, and notwithstanding any other redemptions of Series 1 Shares made by Noranda pursuant to any other provision hereof, Noranda shall, on the Final Redemption Date, redeem the then outstanding Series 1 Shares by payment of an amount in cash for each share redeemed equal to the Basic Redemption Price. If, as a result of applicable law or any provision herein, Noranda fails to redeem all of the Series 1 Shares on the Final Redemption Date, Noranda shall redeem as soon thereafter as possible the remaining Series 1 Shares. For greater certainty, until Noranda is obligated, under the provisions attached to the Series 2 Shares and the Series 3 Shares, to redeem such shares, no redemption of such shares may be made after the Final Redemption Date until all Series 1 Shares have been redeemed.
- (4) **Partial Redemption.** On or before June 30, 2009, if any of the then issued and outstanding Series 1 Shares are to be redeemed, Series 1 Shares will be redeemed as nearly as practicable on a pro rata basis with the Series 2 Shares and Series 3 Shares then outstanding.
- Notice of Redemption. In the case of any Redemption pursuant to subsections (1), (2) or (3) of this section 3, Noranda will give to each holder of Series 1 Shares to be redeemed a notice in writing of the intention of Noranda to redeem such shares (the <u>Redemption Notice</u>) not less than 30 days and not more than 60 days prior to the Redemption Date, but in any event, Noranda will redeem the then outstanding Series 1 Shares on the Redemption Date. The Redemption Notice must set out the number of Series 1 Shares held by the person to whom it is addressed which are to be redeemed, the Redemption Price and the Redemption Date. A Redemption Notice will be validly and effectively given on the date on which it is sent and such notice will be given and sent by ordinary unregistered first class prepaid mail or by delivery addressed to each holder of Series 1 Shares to be redeemed at the last address of such holder as it appears on the register of holders, or, in the case of joint holders, to the address of that one whose name appears first in the register of holders as one of such joint holders, provided that the accidental failure or omission to give any such notices as

Table of Contents 54

1N

aforesaid to one or more of such holders shall not affect the validity of the redemption as to the other holders of the Series 1 Shares to be redeemed.

- Method of Payment. On the Redemption Date, Noranda will pay or cause to be paid to or to the order of the holders of the Series 1 Shares to be redeemed, the Redemption Price thereof on presentation and surrender at the principal office of the Transfer Agent, or at any other place or places within Canada designated in the Redemption Notice, of the certificate or certificates for the Series 1 Shares so called for redemption, together with such other documents as may be reasonably required to effect the redemption of the Series 1 Shares. Payment will be made by cheque payable at par at any branch in Canada of a chartered bank or trust company. If only a part of the Series 1 Shares represented by any certificate are redeemed, a new certificate for the balance will be issued at the expense of Noranda. Subject to subsection 3(5), from and after the date specified in any Redemption Notice, the Series 1 Shares called for redemption will be deemed to be redeemed and the holders of those Series 1 Shares will cease to be entitled to dividends and will not be entitled to exercise any of the rights of shareholders in respect of those Series 1 Shares unless payment of the Redemption Price is not duly made by Noranda on presentation and surrender of the certificate or certificates representing the Series 1 Shares.
- Openosit of Redemption Price. At any time after the Redemption Notice is given, Noranda will have the right to deposit the redemption price of any or all Series 1 Shares called for redemption with any chartered bank or trust company in Canada named in the Redemption Notice, including the Transfer Agent (other than Noranda), to the credit of a special account for the respective holders of those Series 1 Shares to be paid on surrender to Noranda or that chartered bank or trust company of the certificate or certificates representing those Series 1 Shares. Any such deposit will constitute payment and satisfaction of the redemption price of the Series 1 Shares for which the deposit is made and the rights of the holders of those shares will be limited to receiving the proportion of the redemption price so deposited applicable to those shares, without interest, on presentation and surrender of the certificate or certificates representing the Series 1 Shares being redeemed. Noranda will be entitled to any interest on such deposit.
- (8) **Application of Net Proceeds to Redeem Series 1 Shares.** From and including June 30, 2010, Noranda will, subject to section 7 and applicable law, apply the net proceeds it realizes from:
  - (i) any issuance for cash of Common Shares, Preferred Shares, Junior Preference Shares or any securities that, under Canadian generally accepted accounting principles, would be treated as equity on the balance sheet of Noranda (which, for greater certainty, will not include the issue of Common Shares pursuant to any share compensation arrangement, including any stock option plan, share purchase plan or dividend reinvestment plan of Noranda, or pursuant to the terms of the Preferred Shares or the 123 Shares), or
  - (ii) the sale by Noranda of any capital assets outside the ordinary course of business exceeding Cdn. \$250 million, in any one transaction or series of related transactions (and for greater certainty, the entire amount of the net proceeds shall be applied net of any taxes paid or payable);

Table of Contents 55

10

1P

to redeem the 123 Shares; provided that from and including June 30, 2010 until June 29, 2012, so long as any Series 1 Shares are outstanding, the net proceeds will be used first to redeem any then outstanding Series 1 Shares; further provided that from and including June 30, 2012 until June 29, 2015, so long as any Series 1 Shares or Series 2 Shares are then outstanding, the net proceeds will be applied as nearly as practicable on a pro rata basis to any then outstanding Series 1 Shares and Series 2 Shares and from and including June 30, 2015, the net proceeds will be applied as nearly as practicable on a pro rata basis to any then outstanding Series 1 Shares, Series 2 Shares and Series 3 Shares.

## 4. Retraction Right

(1) **Right to Require Retraction.** Subject to section 7, applicable law, and compliance with subsections 4(2), (3) and (4), upon the occurrence of a Change of Control Event, a holder shall have the right at any time after the occurrence of a Change of Control Event (the <u>Retraction Right</u>) to require Noranda to redeem all of the holder s then outstanding Series 1 Shares.

## (2) Retraction Procedure.

- (a) In order to exercise its Retraction Right, a holder must give Noranda notice of such exercise, and on or before the Change of Control Retraction Date, tender to Noranda, at its registered office or at any place at which the Series 1 Shares may be transferred, the certificate or certificates representing the Series 1 Shares which the holder wishes Noranda to redeem. Such tender shall be irrevocable unless payment of the Retraction Price shall not be duly made by Noranda to the holder on or before the Change of Control Retraction Date. In the event that payment of the Retraction Price is not made by Noranda on or before the Change of Control Retraction Date, Noranda shall forthwith thereafter return the holder s deposited share certificate or certificates to the holder.
- (b) Subject to section 7 and applicable law, Noranda shall redeem Series 1 Shares duly tendered pursuant an exercise of the Retraction Right as follows:
  - (i) upon the occurrence of an event contemplated under paragraph (i) of the definition of a Change of Control Event, Noranda shall redeem (x) on the 30 day following the occurrence of such event, all Series 1 Shares that have been tendered to Noranda on or before the third Business Day immediately prior to such 30th day and (y) thereafter, from time to time, within three Business Days following the exercise of the Retraction Right, all Series 1 Shares that have been tendered to Noranda;
  - (ii) upon the occurrence of an event contemplated under paragraph (ii) of the definition of a Change of Control Event , Noranda shall redeem, (x) on the date upon which Common Shares are first taken up under the Take-over Bid, all Series 1 Shares that have been tendered to Noranda on or before the third Business Day immediately prior to the date on which Common Shares are first taken up and (y) thereafter, from time to time, on the third Business Day

10

following the exercise of the Retraction Right, all Series 1 Shares that have been tendered to Noranda; and

(iii) upon the occurrence of an event contemplated under paragraph (iii) of the definition of a Change of Control Event , Noranda shall redeem, (x) on the date that the Control Transaction is completed, all Series 1 Shares that have been tendered to Noranda on or before the third Business Day immediately prior to the date that the Control Transaction is completed and (y) thereafter, from time to time, within three Business Days following the exercise of the Retraction Right, all Series 1 Shares that have been tendered to Noranda:

in each case, by payment of an amount in cash for each Series 1 Share redeemed equal to (i) if the redemption occurs on or before June 30, 2008, the Optional Redemption Price, and (ii) if the redemption occurs after June 30, 2008, the Basic Redemption Price (the <u>Retraction Price</u>). For purposes of determining the Retraction Price, the date fixed for redemption shall be the date upon which Common Shares are first taken up under a Take-over Bid or the date that a Control Transaction is contemplated, as applicable, and, thereafter, the third Business Day immediately following the date on which Series 1 Shares are tendered by a holder to Noranda pursuant to subsection 4(2)(a).

- (c) The provisions of section 3 shall apply to any redemption pursuant to this section 4 except to the extent that such provisions are inconsistent with the provisions of this section 4.
- (3) **Exception.** Notwithstanding anything contained in this section 4, Noranda shall not redeem any Series 1 Shares tendered to Noranda pursuant to the exercise of the Retraction Right upon the occurrence of an event contemplated under paragraph (ii) or (iii) of the definition of Change of Control Event if the applicable Take-over Bid or Control Transaction made or announced is withdrawn or terminated or expires without any person becoming an Acquiring Person and, upon any such withdrawal, termination or expiry, any notice delivered pursuant to subsection 4(2)(a) shall be null and void and of no further force or effect.
- (4) **Notice.** Noranda shall give written notice of the occurrence of any Change of Control Event to all holders within five Business Days of the date on which Noranda first becomes aware of such occurrence.

## 5. Conversion into Common Shares at the Option of Noranda On the Final Redemption Date.

(1) **Conversion Right.** Subject to section 7, applicable law and any requirement to obtain regulatory relief, Noranda, at its option (the <u>Conversion Right</u>), may, upon giving written notice (the <u>Conversion Notice</u>) at least 30 days but no more than 60 days prior to the Final Redemption Date, convert all or any part of the then outstanding Series 1 Shares into that number of freely-tradeable, fully paid and non-assessable Common Shares per Series 1 Share determined by dividing the aggregate of \$25.00, plus an amount equal to accrued and unpaid dividends up to but excluding the Final Redemption Date and calculated in accordance with subsection 2(1)(c) (the whole

1R

constituting the <u>Cash Conversion Price</u>), by the greater of \$2.00 (as adjusted from time to time by the Board of Directors acting in good faith to give effect to any split, division, consolidation, combination or similar change in the capital of Noranda that affects the number of Common Shares then outstanding) and 90% of the U.S. Dollar Equivalent of the Current Market Price as of the Final Redemption Date.

- (2) **Conversion Notice.** The Conversion Notice must set out the number of Series 1 Shares held by the person to whom it is addressed which are to be converted. The Conversion Notice also must advise the holder that the Common Shares will be registered in the name of the holder of Series 1 Shares to be converted unless the Transfer Agent receives from the holder, at least 10 Business Days before the Final Redemption Date, at any principal office of the Transfer Agent, written notice in form and substance satisfactory to the Transfer Agent directing Noranda to arrange for the registration of such Common Shares in some other name or names (the <u>Transferee</u> or the <u>Transferees</u> ) and stating the name or names (with addresses), accompanied by payment to the Transfer Agent of any transfer tax which may be payable by reason of the transfer and a written declaration, if required by any applicable law or by Noranda, as to the residence of the Transferee(s) and such other matters as may be required by law or requested by Noranda in order to determine the entitlement of the Transferee(s) to such Common Shares, in which case such Common Shares will be registered in the name or names so directed in the written notice. Such conversion will be deemed to have been made at 5:00 p.m. (Toronto time) on the Final Redemption Date, so that the rights of the holder of such Series 1 Shares as the holder thereof will cease at such time and the person or persons entitled to receive Common Shares upon such conversion will be treated for all purposes as having become the holder or holders of record of such Common Shares at such time.
- (3) **Pro Rata Conversion.** If less than all of the then outstanding Series 1 Shares are at any time to be converted at the option of Noranda, then: (i) the particular Series 1 Shares to be so converted will be selected on a pro rata basis; and (ii) any Series 1 Shares not converted pursuant to this section 5 will be redeemed pursuant to subsection 3(3).
- (4) **Prohibition on Conversion.** Noranda cannot exercise the Conversion Right hereunder if: (i) subject to Section 5(7), a Change of Control Event has occurred; (ii) an Extraordinary Dividend has been paid; or (iii) any one or more of the following events has occurred:
  - (a) the issuance or delivery of such Common Shares is prohibited pursuant to any agreement or arrangement entered into by Noranda to assure Noranda s solvency or continued operation;
  - (b) the issuance or delivery of such Common Shares is prohibited by law or by any regulatory or other authority having jurisdiction over Noranda and acting in conformity with law;
  - (c) for any reason beyond its control, Noranda is unable to issue Common Shares or is unable to deliver such Common Shares; or
  - (d) such Common Shares are not listed on each exchange on which the Common Shares are then listed for trading.

of Series 1

1**S** 

(5) **Fractional Shares.** Where the aggregate number of Common Shares to be delivered to a holder of Series 1 Shares pursuant to Noranda s exercise of the Conversion Right includes a fraction of a Common Share, Noranda, in lieu of delivering such fractional share, will pay by cheque an amount equal to the balance of the Cash Conversion Price not otherwise satisfied by the delivery of the Common Shares.

## (6) Delivery of Shares and Share Certificates.

- (a) On any conversion of Series 1 Shares pursuant to the exercise of the Conversion Right, the share certificates for Common Shares resulting therefrom will be delivered in the name of the holder of the Series 1 Shares converted or, if applicable, in the name of the Transferee; provided that such holder of Series 1 Shares will (i) pay any applicable security transfer taxes including, without limitation, any documentary, stamp, transfer or other taxes that may be payable in respect of any transfer involved in the issuance or delivery of such shares to a person other than such holder of Series 1 Shares or (ii) will have evidenced to the satisfaction of Noranda that such taxes, if any, have been paid.
- (b) Subject to these Series 1 Provisions, upon the exercise of the Conversion Right, any holder of Series 1 Shares to be converted (or, if applicable, any Transferee) will be deemed to have become a holder of Common Shares of record for all purposes on the Final Redemption Date, notwithstanding any delay in the delivery of certificates representing the Common Shares for which such Series 1 Shares have been converted.
- (c) Upon exercise of the Conversion Right, Noranda is not required to issue Common Shares to any person whose address is in, or whom Noranda or its transfer agent has reason to believe is a resident of, any jurisdiction outside of Canada, to the extent that such issue would require compliance by Noranda with the securities or other laws of such jurisdiction.
- (7) **Exception.** Notwithstanding section 5(4)(i), Noranda shall be entitled to exercise its Conversion Right hereunder in accordance with its terms if the applicable Change of Control Event prohibiting such conversion is the occurrence of (ii) or (iii) under the definition of Change of Control Event and the applicable Take-over Bid or Control Transaction made or announced is withdrawn or terminated or expires without any person becoming an Acquiring Person, upon any such withdrawal, termination or expiry.

#### 6. Purchase for Cancellation

Subject to section 7 and to applicable law, Noranda may, at any time, if obtainable, purchase for cancellation, by private agreement or otherwise, at the lowest price or prices at which, in the opinion of the Board of Directors, such shares are obtainable, all or from time to time any part of the then outstanding Series 1 Shares.

1T

## 7. Restrictions on Dividends and Retirement of Shares

- (1) **General Restrictions.** As long as any Series 1 Shares are outstanding, Noranda will not at any time, without the approval of the holders of the Series 1 Shares given as provided in section 11:
  - (a) declare, pay or set apart for payment any dividends (other than stock dividends payable in shares of Noranda ranking junior to the Series 1 Shares) on shares of Noranda ranking junior to the Series 1 Shares;
  - (b) except out of the net cash proceeds of a substantially concurrent issue of shares of Noranda ranking junior to the Series 1 Shares, redeem or call for redemption, purchase or otherwise pay off, retire or make any return of capital in respect of any shares of Noranda ranking as to capital junior to the Series 1 Shares;
  - (c) redeem or call for redemption, purchase or otherwise retire for value or make any return of capital in respect of less than all of the Series 1 Shares then outstanding;
  - (d) except pursuant to any purchase obligation, sinking fund, retraction privilege or mandatory redemption provisions attaching thereto, redeem or call for redemption, purchase or otherwise pay off, retire or make any return of capital in respect of any Preferred Shares, ranking on a parity with the Series 1 Shares; or
- (e) issue any additional Series 1 Shares or any shares ranking senior to or on a parity with the Series 1 Shares, unless, in each such case, all accrued dividends up to and including those payable on the Dividend Payment Date for the last completed period for which dividends are payable on the Series 1 Shares in respect of which the rights of holders have not been extinguished, and all dividends then accrued on all other shares ranking senior to or on a parity with the Series 1 Shares with respect to the payment of dividends up to the immediately preceding respective date or dates for payment in respect of which the rights of holders of those shares have not been extinguished, have been declared, paid or set apart for payment.
- (2) **Extraordinary Dividends.** As long as 123 Shares having an aggregate issue price of at least \$300 million remain outstanding, Noranda will not at any time, without the approval of the holders of the Series 1 Shares given as provided in section 10 declare, pay or set apart for payment any Extraordinary Dividends.

## 8. Voting Rights and Election of Directors

(1) **Election of Directors.** The holders of Series 1 Shares together with the holders of the Series 2 Shares and the holders of the Series 3 Shares will have at all times the right, voting collectively, to elect two directors to the Board of Directors at each meeting of shareholders of Noranda at which directors are to be elected. Forthwith upon the occurrence of a Board Event, the holders of the Series 1 Shares together with the holders of the Series 2 Shares and the holders of the Series 3 Shares will, but only for so long as a Board Event is continuing, be entitled, voting collectively, to elect three additional directors (for a total of five). Upon the occurrence of a Board Event, the Board of Directors will, at its election and subject to applicable law, either (i) appoint such additional number

1U

of directors to which the holders of the Series 1 Shares (together with the holders of the Series 2 Shares and holders of the Series 3 Shares) are entitled to elect as a result of such occurrence and as may be proposed by the Series 123 Directors, or (ii) forthwith call a special meeting of the applicable shareholders of Noranda for the purpose of electing such additional number of directors and, if necessary, increasing the maximum number of directors that may be elected. The holders of Series 1 Shares are entitled to receive notice of and to attend all annual and special meetings of the Shareholders of Noranda called for the purpose of electing directors as provided in this subsection 8(1) and, for such purpose only, to one vote in respect of each Series 1 Share held at all such meetings.

- (2) **Vacancies.** If a Series 123 Director ceases to be a director for any reason other than as a result of a Board Event ceasing to continue, the Board of Directors will fill the vacancy thereby created by appointing as soon as reasonably possible that individual who is nominated by the remaining Series 123 Directors, provided that if there are no remaining 123 Directors then in office, the Board of Directors will select the individuals to be appointed.
- (3) **Meeting of Shareholders.** Except as otherwise provided by law or as specifically provided herein and except for meetings of the holders of Junior Preference Shares as a class and meetings of all holders of Series 1 Shares as a series, the holders of the Series 1 Shares will not be entitled as such to receive notice of, or to attend, or to vote at, any meeting of shareholders of Noranda.
- (4) Meeting of Holders of Series 1 Shares as a Series or Junior Preference Shares as a Class. At any meeting of holders of Series 1 Shares as a series or together with one or more other series or Junior Preference Shares as a class, each such holder will be entitled to one vote in respect of each Series 1 Shares held.

## 9. Issue of Additional Common Shares

For greater certainty, nothing in these Series 1 Provisions will affect or restrict the right of Noranda to increase the number of the Common Shares or to issue additional Common Shares from time to time.

## 10. Approval of Holders of Series 1 Shares

In addition to any other approvals required by law, the approval of all amendments to the rights, privileges, restrictions and conditions attaching to the Series 1 Shares as a series and any other approval to be given by the holders of Series 1 Shares as a series will be deemed to have been sufficiently given if given by a resolution carried by the affirmative vote of not less than 66²/3% of the votes cast at a meeting of the holders of Series 1 Shares duly called and held on not less than 21 days notice at which the holders of at least a majority of the outstanding Series 1 Shares are present or are represented by proxy, or if no quorum is present at such meeting, at an adjourned meeting at which the holders of the Series 1 Shares then present would form the necessary quorum. If at the meeting the holders of a majority of the outstanding Series 1 Shares are not present or represented by proxy within 30 minutes after the time appointed for the meeting, the meeting will be adjourned to such date, not less than 15 days afterwards, and to such time and place as the chairman may designate, and not less than 10 days written notice will be given of the adjourned meeting. Subject to the foregoing, the formalities to be observed in respect of the giving of notice of any meeting or

1V

adjourned meeting and the conduct of any such meeting will be those prescribed from time to time in the by-laws of Noranda with respect to meetings of shareholders.

## 11. Capital Reorganizations and Amalgamations of Noranda

In the event of a capital reorganization, merger or amalgamation of Noranda or comparable transaction affecting the Series 1 Shares, holders of the Series 1 Shares will be entitled to receive pursuant to the exercise of the Conversion Right, and each such holder will accept, the number of shares or other securities or consideration of Noranda or a corporation resulting, surviving or continuing from the capital reorganization, merger, amalgamation or comparable transaction that such holder would have received had such holder been a holder of Common Shares immediately prior to the record date of the capital reorganization, merger, amalgamation or comparable transaction.

#### 12. Notices

- (1) **Notice to Noranda.** Subject to applicable law, any notice, request or other communication to be given to Noranda by a holder of Series 1 Shares must be in writing and will be valid and effective if given by mail (postage prepaid) or by electronic communication or by delivery to the registered office of Noranda and addressed to the attention of the Secretary. Any such notice, request or other communication, if given by mail, electronic communication or delivery, will be deemed to have been given and received only on actual receipt by Noranda.
- (2) **Presentation and Surrender of Certificates.** Any presentation and surrender by a holder of Series 1 Shares to Noranda or the Transfer Agent of certificates representing Series 1 Shares in connection with the redemption of Series 1 Shares must be made by registered mail (postage prepaid) or by delivery to the registered office of Noranda or to such office of the Transfer Agent as may be specified by Noranda, in each case addressed to the attention of the Secretary of Noranda. Any such presentation and surrender of certificates will be deemed to have been made and to be effective only on actual receipt by Noranda or the Transfer Agent, as the case may be. Any such presentation and surrender of certificates made by registered mail will be at the sole risk of the holder mailing the same.
- (3) Notice to Holders of Series 1 Shares. Subject to applicable law, any notice, including but not limited to the Redemption Notice and the Conversion Notice, request or other communication to be given to a holder of Series 1 Shares by or on behalf of Noranda must be in writing and will be valid and effective if given by ordinary unregistered first class mail (postage prepaid) or by electronic communication or by delivery to the address of the holder recorded in the securities register of Noranda or, in the case of joint holders, to the address of that one whose name appears first in the register of holders as one of such joint holders. Any such notice, request or other communication, if given by mail, will be deemed to have been given and received on the fifth Business Day following the date of mailing and, if given by electronic communication or by delivery, will be deemed to have been given and received on the date of electronic communication or delivery. Accidental failure or omission to give any notice, request or other communication to one or more holders of Series 1 Shares, or any defect in such notice, will not invalidate or otherwise alter or affect any action or proceeding to be taken by Noranda pursuant to that notice, request or other communication.

1W

## 13. Tax Election

Noranda will elect, in the manner and within the time provided under Section 191.2 in Part VI.1 of the Tax Act or any successor or replacement provision of similar effect, and take all other necessary action under the Tax Act, to pay or cause payment of the tax under Part VI.1 of the Tax Act at a rate such that the corporate holders of the Series 1 Shares will not be required to pay tax under Part IV.1 of the Tax Act on dividends received on such shares.

## 14. Return of Unclaimed Funds to Noranda

Subject to applicable laws, Noranda will have the right, with respect to any funds deposited by Noranda to any chartered bank or trust company in respect of amounts due to holders of Series 1 Shares, on or after the second anniversary date of the deposit of such funds to any chartered bank or trust company, to require that such chartered bank or trust company return to Noranda any funds which remain unclaimed by holders of the Series 1 Shares. If at any time following the return of the unclaimed funds to Noranda, either Noranda or the chartered bank or trust company receives a request from a holder of Series 1 Shares for the holder s unclaimed entitlement to any amount due to the holder in respect of the holder s Series 1 Shares, Noranda will promptly arrange for the payment of such amount to the holder.

#### 15. Wire or Electronic Transfer of Funds

Notwithstanding any other right, privilege, restriction or condition attaching to the Series 1 Shares, Noranda may, at its option, make any payment due to a holder of Series 1 Shares hereunder by way of a wire or electronic transfer of funds to each registered holder of Series 1 Shares. In the event that a payment is made by way of a wire or electronic transfer of funds, Noranda will be responsible for any applicable charges or fees relating to the making of such transfer. As soon as practicable following the determination by Noranda that a payment is to be made by way of a wire or electronic transfer of funds, Noranda will notify each registered holder of Series 1 Shares at the address of such holder as it appears on the register of holders. Such notice by Noranda will request that each registered holder of Series 1 Shares provide the particulars of an account of such holder with a chartered bank or trust company in Canada to which the wire or electronic transfer of funds will be directed. In the event that Noranda does not receive account particulars from a registered holder of Series 1 Shares prior to the date such payment is to be made, Noranda will deposit the funds otherwise payable to such holder in a special account or accounts in trust for each respective holder. The making of a payment by way of a wire or electronic transfer of funds or, in the case where a registered holder of the Series 1 Shares has not provided Noranda with account particulars for a wire or electronic transfer of funds, the deposit by Noranda of the funds otherwise payable to such holder in a special account or accounts in trust for such holder, will be deemed to constitute payment by Noranda on the date thereof and will satisfy and discharge all liabilities of Noranda for such payment to the extent of the amount represented by such transfer.

## 16. Additional Series

As long as any of the 123 Shares are outstanding, Noranda may not issue any additional Preferred Shares, other than Cumulative Redeemable Preferred Shares, Series F issuable upon the conversion of Cumulative Redeemable Preferred Shares, Series G and Cumulative Redeemable

1X

Preferred Shares, Series G issuable upon the conversion of Cumulative Redeemable Preferred Shares, Series F, without the approval of the holders of the 123 Shares, expressed by resolution of the holders of all such shares voting together. For this purpose, the approval to be given will be deemed to have been sufficiently given if given by a resolution carried by the affirmative vote of not less than 66²/3% of the votes cast at a meeting of the holders of 123 Shares duly called and held on not less than 21 days notice at which the holders of at least a majority of the outstanding 123 Shares are present or are represented by proxy, or if no quorum is present at such meeting, at an adjourned meeting at which the holders of the 123 Shares then present would form the necessary quorum. If at the meeting the holders of a majority of the outstanding 123 Shares are not present or represented by proxy within 30 minutes after the time appointed for the meeting, the meeting will be adjourned to such date, not less than 15 days afterwards, and to such time and place as the chairman may designate, and not less than 10 days written notice will be given of the adjourned meeting. Subject to the foregoing, the formalities to be observed in respect of the giving of notice of any meeting or adjourned meeting and the conduct of any such meeting will be those prescribed from time to time in the by-laws of Noranda with respect to meetings of shareholders.

1A

# Number and Designation of, and Rights, Privileges, Restrictions and Conditions Attaching to, the Junior Preference Shares, Series 2

The second series of Junior Preference Shares of Noranda shall consist of 20,000,000 shares designated as Junior Preference Shares, Series 2 (the <u>Series 2 Shares</u>) with an issue price of \$25.00 per share (the <u>Issue Price</u>) and, in addition to the rights, privileges, restrictions and conditions attaching to the Junior Preference Shares as a class, shall have attached thereto the following rights, privileges, restrictions and conditions (the <u>Series 2 Provisions</u>):

## 1. Interpretation

(1) **Defined Terms.** In these Series 2 Provisions:

**Acquiring Person** means any person who is the Beneficial Owner of a Significant Interest in the outstanding Voting Shares; provided, however, that the term Acquiring Person shall not include:

- (a) Noranda or any Subsidiary of Noranda;
- (b) Brascan:
- (c) any person who becomes the Beneficial Owner of a Significant Interest in the outstanding Voting Shares as a result of one or any combination of:
  - (A) a Voting Share Reduction which, by reducing the number of Voting Shares outstanding, increases the percentage of Voting Shares Beneficially Owned by such person to a Significant Interest in the Voting Shares then outstanding,
  - (B) a Pro Rata Acquisition,
  - (C) an Exempt Acquisition,
  - (D) a Convertible Security Acquisition; or
  - (E) a 123 Share Acquisition,

provided further, however, that if a person shall become the Beneficial Owner of a Significant Interest in the Voting Shares then outstanding by reason of any one or a combination of a Voting Share Reduction, a Pro Rata Acquisition, an Exempt Acquisition, a Convertible Security Acquisition or a 123 Share Acquisition and thereafter becomes the Beneficial Owner of more than an additional 1% of the outstanding Voting Shares (other than pursuant to a Voting Share Reduction, a Pro Rata Acquisition, an Exempt Acquisition, a Convertible Security Acquisition), or a 123 Share Acquisition), then as of the date that such person becomes the Beneficial Owner of such additional Voting Shares, such person shall become an Acquiring Person;

- (d) any person who, on the Issue Date of the 123 Shares, is the Beneficial Owner of a Significant Interest in the outstanding Voting Shares, provided that if such person thereafter becomes the Beneficial Owner of more than an additional 1% of the outstanding Voting Shares (other than pursuant to a Voting Share Reduction, a Pro Rata Acquisition, an Exempt Acquisition, a Convertible Security Acquisition or a 123 Share Acquisition), then as of the date that such person becomes the Beneficial Owner of such additional Voting Shares, such person shall become an Acquiring Person; and
- (e) an underwriter or member of a banking or selling group that acquires Voting Shares from Noranda in connection with a distribution of securities pursuant to a prospectus or by way of private placement.

**Adjusted Net Worth** means, as at the end of any fiscal quarter, the aggregate value of:

- (a) the outstanding share capital for all shares ranking junior to the Junior Preference Shares;
- (b) without duplication, any surplus, whether contributed or capital;
- (c) retained earnings; and
- (d) consolidated non-controlling interest, all as set forth in Noranda s consolidated balance sheet for such fiscal quarter filed in accordance with applicable securities laws.

**Affiliate**, when used to indicate a relationship with a specified corporation, means a person that directly, or indirectly through one or more controlled intermediaries, controls, or is controlled by, or is under common control with, such specified corporation.

**Associate** , when used to indicate a relationship with a specified person, means (i) a spouse of such specified person, (ii) any person of either sex with whom such specified person is living in a conjugal relationship outside marriage or (iii) any relative of such specified person or of a person mentioned in clauses (i) or (ii) of this definition if that relative has the same residence as the specified person.

**Basic Redemption Price** has the meaning attributed to it in subsection 3(2).

A person shall be deemed the **Beneficial Owner**, and to have **Beneficial Ownership** of and to **Beneficially Own**:

- (i) any securities of which such person or any of such person s Affiliates or Associates is the owner at law or in equity;
- (ii) any securities as to which such person or any of such person s Affiliates or Associates has the right to become the owner at law or in equity, where such right is exercisable immediately or within 60 days of the date of the determination of

Table of Contents 66

1B

1C

Beneficial Ownership and whether or not on condition or the occurrence of any contingency or payment of instalments, upon the exercise of any conversion right, exchange right or purchase right attaching to Convertible Securities, or pursuant to any agreement, arrangement, pledge or understanding, written or oral (other than customary agreements with and between underwriters and/or banking group and/or selling group members with respect to a distribution of securities pursuant to a prospectus or by way of private placement and other than pursuant to pledges of securities in the ordinary course of business); and

- (iii) any securities which are Beneficially Owned within the meaning of clauses (i) or (ii) of this definition by any other person with which, and in respect of which securities, such person is acting jointly or in concert;
  - provided, however, that a person shall not be deemed the Beneficial Owner of, or to have Beneficial Ownership of, or to Beneficially Own , any security:
  - (A) by reason of such security having been deposited or tendered pursuant to a Take-over Bid made by, or Control Transaction involving, such person or any of such person s Affiliates or Associates or any other person referred to in clause (iii) of this definition until the earlier of such deposited or tendered security being accepted unconditionally for payment or exchange or being taken up and paid for;
  - (B) by reason of such person, any of such person s Affiliates or Associates or any other person referred to in clause (iii) of this definition holding such security, if:
    - (I) the ordinary business of the person (in this definition, a <u>Manager</u>) includes the management of investment funds for others and such security is held by the Manager in the ordinary course of such business in the performance of such Manager s duties for the account of any other person (in this definition, a <u>Client</u>) including non-discretionary accounts held on behalf of a Client by a broker or dealer registered under applicable law;
    - (II) the person (in this definition, a <u>Trust Company</u>) is licensed to carry on the business of a trust company under applicable law and, as such, acts as a trustee or administrator or in a similar capacity in relation to the estates of deceased or incompetent persons (each, in this definition, an Estate Account ) or in relation to other accounts (each, in this definition, an Other Account ) and holds such security and is acting in the ordinary course of such duties for the Estate Account or for such Other Accounts;
    - (III) such person is a Crown agent or agency (in this definition, the <u>Crown Agent</u>);

1D

- (IV) the person is established by statute for purposes that include, and the ordinary business or activity of such person (in this definition, a <u>Statutory Body</u>) includes, the management of investment funds for employee benefit plans, pension plans, insurance plans of various public bodies and the Statutory Body holds such security for the purposes of its activities as such; or
- (V) the person (in this definition, an <u>Administrator</u>) is the administrator or trustee of one or more pension funds or plans (each, in this definition, a <u>Plan</u>) registered under the laws of Canada or any province thereof or the similar laws of the jurisdiction by which such Plan is governed or is such a Plan and the Administrator or Plan holds such security for the purposes of its activities as such;

but only if the Manager, the Trust Company, the Crown Agent, the Statutory Body, the Administrator or the Plan, as the case may be, is not then making and has not announced a current intention to make a Take-over Bid or a Control Transaction, other than an Offer to Acquire Voting Shares or other securities pursuant to a distribution by Noranda, or by means of ordinary market transactions (including prearranged trades entered into in the ordinary course of business of such person) executed through the facilities of a stock exchange or an organized over-the-counter market, alone or by acting jointly or in concert with any other person;

## (C) because such person:

- (I) is a Client of the same Manager as another person on whose account the Manager holds such security,
- (II) has an Estate Account or an Other Account with the same Trust Company as another person on whose account the Trust Corporation holds such security, or
- (III) is a Plan with the same Administrator as another Plan on whose account the Administrator holds such security,

## (D) because such person:

- (I) is a Client of a Manager and such security is owned at law or in equity by the Manager,
- (II) has an Estate Account or an Other Account with a Trust Company and such security is owned at law or in equity by the Trust Company, or
- (III) is a Plan and such security is owned at law or in equity by the Administrator of the Plan; or

1E

(E) because such person is the registered holder of securities as a result of carrying on the business of, or acting as nominee for, a securities depositary.

**Board Event** means any one of the following events:

- (a) if four quarterly dividends on the Series 2 Shares are in arrears, whether or not such dividends have been declared and whether or not there are any monies of Noranda properly applicable to the payment of dividends; or
- (b) if the Adjusted Net Worth for any fiscal quarter is less than \$2,500,000,000.

**Board of Directors** means the board of directors of Noranda.

**Brascan** means Brascan Corporation, any Affiliate of Brascan Corporation or any party acting jointly or in concert with Brascan Corporation or any of its Affiliates.

**Business Day** means a day other than a Saturday, a Sunday or any other day which is a statutory or civic holiday in Toronto, Ontario.

**Cash Conversion Price** has the meaning attributed to it in subsection 5(1).

## **Change of Control Event** means the occurrence of:

- (i) the first date of public announcement (which, for purposes of this definition, shall include, without limitation, a report filed pursuant to section 101 of the *Securities Act* (Ontario)) by Noranda or an Acquiring Person of facts indicating that an Acquiring Person has become such;
- (ii) the date of the commencement of, or first public announcement of the intent of any person (other than Brascan, Noranda or any Subsidiary of Noranda) to commence a Take-over Bid (other than by Brascan, Noranda or any Subsidiary of Noranda); or
- (iii) the date of the commencement of, or first public announcement of the intent of any person (other than Brascan) to commence a Control Transaction (other than by Brascan, Noranda or any Subsidiary of Noranda).

**Change of Control Retraction Date** means the date on which the Series 2 Shares are redeemed pursuant to subsection 4(2)(b)(i), (ii) or (iii), as applicable.

**Common Shares** means the common shares of Noranda, and any other securities into which those shares may be changed or for which those shares may be exchanged (whether or not Noranda is the issuer of such other securities).

**Control Transaction** means any merger, arrangement, amalgamation, reorganization, acquisition, material sale of shares or rights or interests therein or thereto or similar transactions involving Noranda, which will result in any person Beneficially Owning, whether directly as a result of such transaction or together with the Common Shares that such person Beneficially Owned before the completion of such transaction, a Significant Interest in Noranda.

1F

**controlled**: a body corporate is controlled by another person if:

- (i) securities entitled to vote in the election of directors carrying more than 50% of the votes for the election of directors are held, directly or indirectly, by or on behalf of the other person; and
- (ii) the votes carried by such securities are entitled, if exercised, to elect a majority of the board of directors of such body corporate,

and controls, controlling and under common control with shall be interpreted accordingly.

**Conversion Notice** has the meaning attributed to it in subsection 5(1).

**Conversion Right** has the meaning attributed to it in subsection 5(1).

Convertible Securities means, at any time, any securities issued by Noranda (including rights, warrants and options but other than the 123 Shares) carrying any purchase, exercise, conversion or exchange right, pursuant to which the holder thereof may acquire Voting Shares or other securities convertible into or exercisable or exchangeable for Voting Shares (in each case, whether such right is exercisable immediately or after a specified period and whether or not on condition or the happening of any contingency).

**Convertible Security Acquisition** means the acquisition of Voting Shares upon the exercise of Convertible Securities acquired by a person pursuant to an Exempt Acquisition or a Pro Rata Acquisition.

Current Market Price means, at any date, the volume weighted average trading price of Common Shares on the Toronto Stock Exchange (or, if not then listed on that exchange, on another exchange or market chosen by the Board of Directors in their sole discretion on which Common Shares are then traded), for a period of 20 consecutive Trading Days ending on the fourth day immediately prior to that date or, if such fourth day is not a Trading Day, on the immediately preceding Trading Day.

**Dividend** and **Dividends** have the respective meanings attributed to them in subsection 2(1)(a).

**Dividend Payment Date** means the last day of the months of March, June, September and December in each year.

**Dividend Period** means the Initial Dividend Period and, thereafter, the period from and including a Dividend Payment Date to but excluding the next succeeding Dividend Payment Date.

**Exempt Acquisition** means an acquisition of Voting Shares or Convertible Securities (i) pursuant to a distribution of Voting Shares or Convertible Securities (and the conversion or exchange of such Convertible Securities) made by Noranda pursuant to a prospectus or private placement provided that the person does not acquire a greater percentage of the securities offered in the distribution than the percentage of Voting Shares Beneficially Owned by that person

# Edgar Filing: NORANDA INC - Form SC 13D/A

immediately prior to the distribution, or (ii) pursuant to an amalgamation, merger or other statutory procedure requiring shareholder approval.

## **Extraordinary Dividend** means:

- (i) any dividend, other than a stock dividend paid wholly in Common Shares, declared or paid on the Common Shares that is, when taken together with the amount or value of all other dividends declared or paid in the 12 month period preceding the date of declaration of the dividend (the measurement period) more than 125% of the aggregate amount or value of the dividends declared or paid on the Common Shares, other than stock dividends paid wholly in Common Shares, during the 12 month period ended on the day prior to the measurement period (or, where there were no dividends paid in the 12 month period ended on the day prior to the measurement period, the period of 12 months ending on the date on which Noranda last paid a dividend, other than a stock dividend paid wholly in Common Shares, on the Common Shares); or
- (ii) any dividend, other than a stock dividend paid wholly in Common Shares, declared or paid on the Common Shares that results in the Noranda Annual Dividend Rate exceeding a 10% compound annual growth rate, measured from and following the Issue Date. For this purpose, the Noranda Annual Dividend Rate is initially the aggregate amount or value of all dividends declared or paid on the Common Shares, other than stock dividends paid wholly in Common Shares, in the 12 month period immediately preceding the Issue Date and thereafter is the aggregate amount or value of all dividends declared or paid in any 12 month period immediately preceding the date of declaration of any other dividend on the Common Shares, together with the dividend then being declared; or
- (iii) any other special dividend on, or distribution with respect to, the Common Shares which is, by its terms or declared intent, declared and paid outside the normal operations or normal dividend procedures of Noranda. **Final Redemption Date** means June 30, 2012.

**freely-tradeable** means, in respect of a Common Share, that (i) the Common Share is transferable and not subject to any escrow requirements or to any cease trade order imposed by a Canadian securities regulatory authority, and (ii) all hold periods imposed by Canadian securities legislation before such Common Share can be traded without a prospectus or in reliance on a prospectus exemption have expired.

**Initial Dividend Period** means the period from and including the Issue Date to but excluding the first Dividend Payment Date after the Issue Date.

**Issue Date** means the date any of the Series 2 Shares are first issued.

**Issue Price** has the meaning attributed to it in the introductory paragraph to these rights, privileges, restrictions attaching to the Series 2 Shares.

**Noranda** means Noranda Inc. and any corporation resulting, surviving or continuing from the merger, arrangement, amalgamation or reorganization of Noranda Inc., or a comparable transaction.

Table of Contents 72

1G

1H

# **Offer to Acquire** shall include:

- (i) an offer to purchase, or a solicitation of an offer to sell, Voting Shares, and
- (ii) an acceptance of an offer to sell Voting Shares, whether or not such offer to sell has been solicited, or any combination thereof, and the person accepting an offer to sell shall be deemed to be making an Offer to Acquire to the person that made the offer to sell.

**Offeror** means a person who has announced an intention to make or who has made a Take-over Bid (excluding an Offer to Acquire Common Shares or other securities made by a Manager, Trust Company, Crown Agent, Statutory Body, Administrator or Plan referred to in clause (B) of the definition of Beneficial Owner pursuant to a distribution by Noranda or by means of ordinary market transactions (including pre-arranged trades entered into in the ordinary course of business of such person) in the circumstances contemplated in said clause (B)), but only so long as the Take-over Bid so announced or made has not been withdrawn or terminated or has not expired.

**Offeror s Securities** means the aggregate of the Voting Shares Beneficially Owned by an Offeror on the date of an Offer to Acquire by an Offeror.

**123 Shares** means, collectively, the Series 1 Shares, the Series 2 Shares and the Series 3 Shares.

**123 Share Acquisition** means, in respect of holders of the 123 Shares, (i) the payment of a Dividend by the issuance of Common Shares pursuant to subsection 2(2)(a) or (ii) the issuance of Common Shares upon exercise by Noranda of its conversion right pursuant to section 5.

**Optional Redemption Price** has the meaning attributed to it in subsection 3(1).

**Pro Rata Acquisition** means an acquisition by a person of Voting Shares or Convertible Securities (i) as a result of a stock dividend, a stock split or other event pursuant to which such person receives or acquires Voting Shares or Convertible Securities on the same pro rata basis as all other holders of Voting Shares of the same class or series of Noranda, (ii) pursuant to a regular dividend reinvestment or other plan of Noranda made available by Noranda to the holders of Voting Shares where such plan permits the holder to direct that the dividends paid in respect of such Voting Shares be applied to the purchase from Noranda of further securities of Noranda, or (iii) pursuant to the receipt and/or exercise of rights (other than the 123 Shares) issued by Noranda to all of the holders of a series or class of Voting Shares on a pro rata basis to subscribe for or purchase Voting Shares or Convertible Securities, provided that the person does not acquire a greater percentage of the securities issuable on exercise of such rights than the percentage of Voting Shares Beneficially Owned by that person immediately prior to the commencement of the offering of rights and that such rights are acquired directly from Noranda and not from any other person.

**Redemption Date** means (i) in the case of a mandatory redemption pursuant to subsection 3(3), the Final Redemption Date, (ii) in the case of a redemption in connection with an exercise of the Retraction Right, the Change of Control Retraction Date, and (iii) in all other cases, the date on which the redemption is to occur.

1I

**Redemption Notice** has the meaning attributed to it in subsection 3(5).

**Redemption Price** means (i) if the redemption occurs on or before June 30, 2008, the Optional Redemption Price, and (ii) if the redemption occurs after June 30, 2008, the Basic Redemption Price.

**Re-Set Dividend Period** means the period from and including June 30, 2010 to but excluding June 30, 2012.

**Re-Set Dividend Rate** means the greater of (i) 6.25%, and (ii) the Seven Year U.S. Treasury Bond Yield plus 2.05%.

**Retraction Price** has the meaning attributed to it in subsection 4(2)(b).

**Retraction Right** has the meaning attributed to it in subsection 4(1).

**Series 1 Shares** means the first series of Junior Preference Shares of Noranda, designated as Junior Preference Shares, Series 1.

**Series 123 Directors** means, at any time, those directors of Noranda most recently elected by or appointed on behalf of the holders of the Series 1 Shares, Series 2 Shares and Series 3 Shares, collectively.

**Series 2 Provisions** has the meaning attributed to it in the introductory paragraph to these rights, privileges, restrictions and conditions attaching to the Series 2 Shares.

**Series 2 Shares** has the meaning attributed to it in the introductory paragraph to these Series 2 Provisions.

**Series 3 Shares** means the third series of Junior Preference Shares of Noranda, designated as Junior Preference Shares, Series 3.

**Seven Year U.S. Treasury Bond Yield** means on any date shall mean the average of the yields determined by two registered Canadian investment dealers, selected by the Board of Directors, as being the yield to maturity on such date compounded semi-annually and calculated in accordance with generally accepted financial practice, which a non-callable United States Treasury Bond would carry if issued in United States dollars in the United States at 100% of its principal amount on such date with a term to maturity of seven years.

**Significant Interest**: a person has a Significant Interest in the Voting Shares where that person Beneficially Owns 30% or more of the outstanding Voting Shares.

**Subsidiary**: a body corporate is a Subsidiary of another body corporate if:

- (i) it is controlled by (A) that other, or (B) that other and one or more bodies corporate, each of which is controlled by that other, or (C) two or more bodies corporate, each of which is controlled by that other; or
- (ii) it is a Subsidiary of a body corporate that is that other s Subsidiary.

# Edgar Filing: NORANDA INC - Form SC 13D/A

#### **Table of Contents**

1J

**Take-over Bid** means an Offer to Acquire Voting Shares, where the Voting Shares subject to the Offer to Acquire and the Offeror's Securities, constitute a Significant Interest in the outstanding Voting Shares at the date of the Offer to Acquire.

**Tax Act** means the *Income Tax Act* (Canada).

**Trading Day** means with respect to any stock exchange or over-the-counter market, a day on which shares may be traded through the facilities of such stock exchange or in such over-the-counter market, and otherwise means a day on which shares may be traded through the facilities of the principal stock exchange on which the Common Shares are listed (or, if the Common Shares are not listed on any stock exchange, then in the over-the-counter market).

**Transfer Agent** means Noranda, or such other person as from time to time may be the registrar and transfer agent for the Series 2 Shares.

**Transferee** and **Transferees** have the meanings attributed to them in subsection 5(2).

- **U.S.** Canadian Exchange Rate means, on any date:
  - (i) if, on such date, the Bank of Canada sets an average noon spot rate of exchange for the conversion of one Canadian dollar into United States dollars, such rate; or
  - (ii) in any other case, the rate for such date for the conversion of one Canadian dollar into United States dollars calculated in such manner as may be determined by the Board of Directors from time to time acting in good faith.
- **U.S. Dollar Equivalent** of any amount which is expressed in Canadian dollars means, on any date, the United States dollar equivalent of such amount determined by multiplying such amount by the U.S. Canadian Exchange Rate in effect on such date.

**Voting Share Reduction** means an acquisition or a redemption by Noranda of Voting Shares.

**Voting Shares** means collectively the Common Shares and any other shares in the capital stock or voting interests of Noranda entitled to vote generally in the election of directors.

- (2) **Ranking of Shares.** The expressions in priority to , on a parity with , ranking equally with , ranking prior to ranking junior to and similar expressions refer to the order of priority only in payment of dividends or in the distribution of assets in the event of any liquidation, dissolution of winding-up of Noranda, whether voluntary or involuntary, or any other distribution of the assets of Noranda for the purpose of winding-up its affairs.
- (3) **Holder.** References to a <u>holder</u> in these Series 2 Provisions mean the registered owner of such shares.
- (4) **Extended Meanings.** References in these Series 2 Provisions to words importing the singular number only include the plural and *vice versa*, words importing any gender include all genders and words importing persons include individuals, corporations, limited and unlimited liability companies, general and limited partnership, associations, trusts, unincorporated

1K

organizations, joint ventures and governmental authorities. The term including means including without limiting the generality of the foregoing .

- (5) **References to Statutes.** Except as otherwise noted herein, reference to any statute in these Series 2 Provisions is to that statute as in force from time to time, including any regulations, rules, policy statements or guidelines made under that statute, and includes any statute which may be enacted in substitution of that statute.
- (6) **References to Sections**. Reference in these Series 2 Provisions to any section or subdivision is to the designated section or subdivision of these Series 2 Provisions.
- Acting Jointly or in Concert. For purposes of these Series 2 Provisions, a person is acting jointly or in concert with every other person who is a party to any agreement, commitment or understanding (whether formal or informal and whether or not in writing) with the first mentioned person to acquire or offer to acquire or vote Voting Shares (other than (i) in the case of agreements to acquire Voting Shares, customary agreements with and between underwriters and/or banking group and/or selling group members with respect to a distribution of securities pursuant to a prospectus or by way of private placement and other than pursuant to pledges of securities in the ordinary course of business and (ii) in the case of agreements to vote Voting Shares, agreements to vote for a limited purpose and without the intention of acquiring or exercising control).

## (8) Other Payment Matters.

- (a) If any date on which any Dividend on the Series 2 Shares is payable or on or by which any other action is required to be taken by Noranda under these Series 2 Provisions is not a Business Day, then the Dividend will be payable, or such other action will be required to be taken, on or by the next day that is a Business Day.
- (b) In the event of the non-receipt of a cheque by a holder of Series 2 Shares entitled to the cheque, or the loss or destruction of the cheque, Noranda, on being furnished with reasonable evidence of non-receipt, loss or destruction, and an indemnity reasonably satisfactory to Noranda, will issue to the holder a replacement cheque for the amount of the original cheque.
- (c) Notwithstanding any other provision of these Series 2 Provisions, Noranda may deduct or withhold from any payment, distribution, issuance or delivery to be made pursuant to these share terms any amounts required or permitted by law to be deducted or withheld from any such payment, distribution, issuance or delivery and will remit such amounts to the relevant tax authority as required. Where the cash component of any payment, distribution, issuance or delivery to be made pursuant to these share terms is less than the amount that Noranda is so required or permitted to deduct or withhold, Noranda will be permitted to deduct and withhold from any non-cash payment, distribution, issuance or delivery to be made pursuant to these share terms any amounts required or permitted by law to be deducted or withheld from any such payment, distribution, issuance or delivery and to dispose of such property in order to remit any amount required to be remitted to any relevant tax authority. Notwithstanding the above, the amount of any payment, distribution, issuance or

1L

delivery made pursuant to these Series 2 Provisions will be considered to be the amount of the payment, distribution, issuance or delivery plus any amount deducted or withheld pursuant to this subsection 1(8)(c).

## (9) Currency.

Except as otherwise provided in these Series 2 Provisions, all references to currency herein are to lawful money of the United States.

# (10) Designation Under the Tax Act.

The amount of is specified in respect of each Series 2 Share for the purposes of subsection 191(4) of the Tax Act.

## 2. Dividends

# (1) Payment of Dividends.

- (a) Until June 30, 2010, the holders of Series 2 Shares will be entitled to receive and Noranda will pay thereon, if, as and when declared by the Board of Directors out of the assets of Noranda properly applicable to the payment of dividends, fixed preferential cumulative cash dividends (each a <u>Dividend</u> and collectively, the <u>Dividends</u>) in an amount per Series 2 Share per annum equal to \$1.5625, payable quarterly, in equal instalments of \$0.3906 per Series 2 Share, with respect to each Dividend Period other than the Initial Dividend Period, on the Dividend Payment Date immediately following the end of each such Dividend Period.
- (b) The dividend payable in respect of the Initial Dividend Period, if declared by the Board of Directors, will be payable on June 30, 2005 in the amount per Series 2 Share obtained when:
  - (i) \$1.5625 is multiplied by the number of days beginning on the Issue Date up to but excluding June 30, 2005; and
  - (ii) the result in paragraph (i) is divided by 365.
- (c) During the Re-Set Dividend Period, the Series 2 Shares will be entitled to receive and Noranda will pay thereon, if, as and when declared by the Board of Directors out of the assets of Noranda properly applicable to the payment of dividends, fixed preferential cumulative cash dividends in an amount per Series 2 Share per annum equal to the Re-Set Dividend Rate, payable quarterly, in equal instalments per Series 2 Share, with respect to each Dividend Period, on the Dividend Payment Date immediately following the end of each such Dividend Period.
- (d) In any case where Dividends are payable for a period that ends on a date other than a Dividend Payment Date on which a Dividend was paid, Dividends will be paid in an amount per Series 2 Share obtained when:

1M

- (i) \$1.5625 is multiplied by the number of days beginning on the date of the immediately preceding Dividend Payment Date up to but excluding the date fixed for payment; and
- (ii) the result in paragraph (i) is divided by 365.
- (e) If on any Dividend Payment Date the dividend payable on such date is not paid in full on all the Series 2 Shares then issued and outstanding, such dividend, or the unpaid part thereof, will be paid at a subsequent date or dates determined by the Board of Directors on which Noranda has sufficient monies properly applicable to the payment of such dividends. The holders of Series 2 Shares will not be entitled to any dividends other than or in excess of the preferential cumulative dividends hereinbefore provided.

# (2) Method of Payment.

- (a) Dividends on the Series 2 Shares will (other than in the case of redemption, in which case payment of dividends will be made on surrender of the certificate representing the Series 2 Shares to be redeemed) be paid by cheque, in lawful money of Canada, payable at par at any branch in Canada of a chartered bank or trust company, in the amount, rounded to the nearest whole cent (\$0.01) of the applicable dividend or, at the election of Noranda and subject to applicable law and any requirement to obtain regulatory relief, by issuing freely-tradeable, fully paid and non-assessable Common Shares. The mailing by ordinary unregistered first class prepaid mail of such a cheque or share certificate, as applicable, to a holder of Series 2 Shares to the address of the holder as it appears on the register of holders, or, in the case of joint holders, to the address of that one whose name appears first in the register of holders as one of such joint holders on or before the applicable Dividend Payment Date will be deemed to be payment and will satisfy and discharge all liabilities for dividends payable on that Dividend Payment Date to the extent of the amount represented by the cheque (plus any tax required to be deducted and withheld from the payment) or, calculated in accordance with subsection 2(2)(b), such share certificate unless, in case of payment by cheque, such cheque is not paid on due presentation. Dividends represented by a cheque which has not been duly presented for payment within two years after it was issued or that otherwise remain unclaimed for a period of two years from the date on which they were declared to be payable and set apart for payment will be forfeited to Noranda. Noranda may pay dividends in any other manner as it may agree with any particular holder.
- (b) If Noranda elects to pay a Dividend by issuing Common Shares in accordance with subsection 2(2)(a), the number of Common Shares to be issued as payment of a Dividend will be equal to the quotient obtained by dividing (i) the Dividend payable by (ii) 95% of the U.S. Dollar Equivalent of the Current Market Price as of the Dividend Payment Date, less the amount of any tax required to be deducted and withheld by Noranda. In any case where the aggregate number of Common Shares to be issued to a holder on payment of a Dividend pursuant to this section includes a fraction of a Common Share, Noranda, in lieu of delivering a fractional share, will pay by cheque an amount equal to the balance of the Dividend payable not otherwise

satisfied by the delivery of Common Shares. Noranda cannot pay a Dividend by issuing Common Shares if one or more of the following events has occurred:

- (i) the issuance or delivery of such Common Shares is prohibited pursuant to any agreement or arrangement entered into by Noranda to assure Noranda s solvency or continued operation;
- (ii) the issuance or delivery of such Common Shares is prohibited by law or by any regulatory or other authority having jurisdiction over Noranda and acting in conformity with law;
- (iii) for any reason beyond its control, Noranda is unable to issue Common Shares or is unable to deliver such Common Shares; or
- (iv) such Common Shares are not listed on each exchange on which the Common Shares are then listed for trading.

# 3. Redemption

- (1) **Redemption On or Before June 30, 2010.** On or before June 30, 2010, but subject to section 7 and to applicable law, Noranda, may upon giving notice as hereinafter provided, redeem at any time the whole or from time to time any part of the then outstanding Series 2 Shares from, subject to subsection 3(4), any one or more of the holders thereof as the Board of Directors may in its sole discretion determine by payment of an amount in cash for each share to be redeemed of \$25.25, together with an amount equal to accrued and unpaid dividends up to but excluding the date fixed for redemption and calculated in accordance with subsection 2(1)(d) (the <u>Optional Redemption Price</u>).
- Redemption From July 1, 2010 to June 29, 2012. From and including July 1, 2010 to and including June 29, 2012, but subject to section 7 and to applicable law, Noranda, may upon giving notice as hereinafter provided, redeem at any time the whole or from time to time any part of the then outstanding Series 2 Shares from, subject to subsection 3(4), the holders thereof as the Board of Directors may in its sole discretion determine by payment of an amount in cash for each share redeemed of \$25.00, together with an amount equal to accrued and unpaid dividends up to but excluding the date fixed for redemption and calculated in accordance with subsection 2(1)(d) (the <u>Basic Redemption Price</u>).
- (3) Mandatory Redemption on the Final Redemption Date. Subject to section 7 and to applicable law and to any notice of conversion given by Noranda pursuant to section 5, and notwithstanding any other redemptions of Series 2 Shares made by Noranda pursuant to any other provision hereof, Noranda shall, on the Final Redemption Date, redeem the then outstanding Series 2 Shares by payment of an amount in cash for each share redeemed equal to the Basic Redemption Price. If, as a result of applicable law or any provision herein, Noranda fails to redeem all of the Series 2 Shares on the Final Redemption Date, Noranda shall redeem as soon thereafter as possible the remaining Series 2 Shares, as nearly as practicable on a pro rata basis with any then outstanding Series 1 Shares.

Table of Contents 79

1N

10

- (4) **Partial Redemption.** On or before June 30, 2011 but subject to the immediately following sentences, if any of the then issued and outstanding Series 2 Shares are to be redeemed, Series 2 Shares will be redeemed as nearly as practicable on a pro rata basis with the Series 1 Shares and Series 3 Shares then outstanding. From June 30, 2010 to June 29, 2012, no Series 2 Shares may be redeemed hereunder until all the Series 1 Shares have been redeemed. Following June 30, 2012, any redemption of the Series 2 Shares hereunder shall be made as nearly as practicable on a pro rata basis with any Series 1 Shares then outstanding.
- Notice of Redemption. In the case of any Redemption pursuant to subsections (1), (2) or (3) of this section 3, Noranda will give to each holder of Series 2 Shares to be redeemed a notice in writing of the intention of Noranda to redeem such shares (the Redemption Notice ) not less than 30 days and not more than 60 days prior to the Redemption Date, but in any event, Noranda will redeem the then outstanding Series 2 Shares on the Redemption Date. The Redemption Notice must set out the number of Series 2 Shares held by the person to whom it is addressed which are to be redeemed, the Redemption Price and the Redemption Date. A Redemption Notice will be validly and effectively given on the date on which it is sent and such notice will be given and sent by ordinary unregistered first class prepaid mail or by delivery addressed to each holder of Series 2 Shares to be redeemed at the last address of such holder as it appears on the register of holders, or, in the case of joint holders, to the address of that one whose name appears first in the register of holders as one of such joint holders, provided that the accidental failure or omission to give any such notices as aforesaid to one or more of such holders shall not affect the validity of the redemption as to the other holders of the Series 2 Shares to be redeemed.
- Method of Payment. On the Redemption Date, Noranda will pay or cause to be paid to or to the order of the holders of the Series 2 Shares to be redeemed, the Redemption Price thereof on presentation and surrender at the principal office of the Transfer Agent, or at any other place or places within Canada designated in the Redemption Notice, of the certificate or certificates for the Series 2 Shares so called for redemption, together with such other documents as may be reasonably required to effect the redemption of the Series 2 Shares. Payment will be made by cheque payable at par at any branch in Canada of a chartered bank or trust company. If only a part of the Series 2 Shares represented by any certificate are redeemed, a new certificate for the balance will be issued at the expense of Noranda. Subject to subsection 3(5), from and after the date specified in any Redemption Notice, the Series 2 Shares called for redemption will be deemed to be redeemed and the holders of those Series 2 Shares will cease to be entitled to dividends and will not be entitled to exercise any of the rights of shareholders in respect of those Series 2 Shares unless payment of the Redemption Price is not duly made by Noranda on presentation and surrender of the certificate or certificates representing the Series 2 Shares.
- (7) **Deposit of Redemption Price.** At any time after the Redemption Notice is given, Noranda will have the right to deposit the redemption price of any or all Series 2 Shares called for redemption with any chartered bank or trust company in Canada named in the Redemption Notice, including the Transfer Agent (other than Noranda), to the credit of a special account for the respective holders of those Series 2 Shares to be paid on surrender to Noranda or that chartered bank or trust company of the certificate or certificates representing those Series 2 Shares. Any such deposit will constitute payment and satisfaction of the redemption price of the Series 2 Shares for which the deposit is made and the rights of the holders of those shares will be limited to receiving the proportion of the redemption price so deposited applicable to those shares, without interest, on presentation and

1P

surrender of the certificate or certificates representing the Series 2 Shares being redeemed. Noranda will be entitled to any interest on such deposit.

- (8) **Application of Net Proceeds to Redeem Series 2 Shares.** From and including June 30, 2010, Noranda will, subject to section 7 and applicable law, apply the net proceeds it realizes from:
  - (i) any issuance for cash of Common Shares, Preferred Shares, Junior Preference Shares or any securities that, under Canadian generally accepted accounting principles, would be treated as equity on the balance sheet of Noranda (which, for greater certainty, will not include the issue of Common Shares pursuant to any share compensation arrangement, including any stock option plan, share purchase plan or dividend reinvestment plan of Noranda, or pursuant to the terms of the Preferred Shares or the 123 Shares), or
  - (ii) the sale by Noranda of any capital assets outside the ordinary course of business exceeding Cdn. \$250 million, in any one transaction or series of related transactions (and for greater certainty, the entire amount of the net proceeds shall be applied net of any taxes paid or payable);

to redeem the 123 Shares; provided that from and including June 30, 2010 until June 29, 2012, so long as any Series 1 Shares are outstanding, the net proceeds will be used first to redeem any then outstanding Series 1 Shares; further provided that from and including June 30, 2012 until June 29, 2015, so long as any Series 1 Shares or Series 2 Shares are then outstanding, the net proceeds will be applied as nearly as practicable on a pro rata basis to any then outstanding Series 1 Shares and Series 2 Shares and from and including June 30, 2015, the net proceeds will be applied as nearly as practicable on a pro rata basis to any then outstanding Series 1 Shares, Series 2 Shares and Series 3 Shares.

# 4. Retraction Right

(1) **Right to Require Retraction.** Subject to section 7, applicable law, and compliance with subsections 4(2), (3) and (4), upon the occurrence of a Change of Control Event, a holder shall have the right at any time after the occurrence of a Change of Control Event (the <u>Retraction Right</u>) to require Noranda to redeem all of the holder s then outstanding Series 2 Shares.

## (2) Retraction Procedure.

(a) In order to exercise its Retraction Right, a holder must give Noranda notice of such exercise, and on or before the Change of Control Retraction Date, tender to Noranda, at its registered office or at any place at which the Series 2 Shares may be transferred, the certificate or certificates representing the Series 2 Shares which the holder wishes Noranda to redeem. Such tender shall be irrevocable unless payment of the Retraction Price shall not be duly made by Noranda to the holder on or before the Change of Control Retraction Date. In the event that payment of the Retraction Price is not made by Noranda on or before the Change of Control Retraction Date, Noranda shall forthwith thereafter return the holder s deposited share certificate or certificates to the holder.

10

- (b) Subject to section 7 and applicable law, Noranda shall redeem Series 2 Shares duly tendered pursuant an exercise of the Retraction Right as follows:
  - (i) upon the occurrence of an event contemplated under paragraph (i) of the definition of a Change of Control Event , Noranda shall redeem (x) on the 30 day following the occurrence of such event, all Series 2 Shares that have been tendered to Noranda on or before the third Business Day immediately prior to such 30th day and (y) thereafter, from time to time, within three Business Days following the exercise of the Retraction Right, all Series 2 Shares that have been tendered to Noranda;
  - (ii) upon the occurrence of an event contemplated under paragraph (ii) of the definition of a Change of Control Event , Noranda shall redeem, (x) on the date upon which Common Shares are first taken up under the Take-over Bid, all Series 2 Shares that have been tendered to Noranda on or before the third Business Day immediately prior to the date on which Common Shares are first taken up; and (y) thereafter, from time to time, on the third Business Day following the exercise of the Retraction Right, all Series 2 Shares that have been tendered to Noranda;
  - (iii) upon the occurrence of an event contemplated under paragraph (iii) of the definition of a Change of Control Event , Noranda shall redeem, (x) on the date that the Control Transaction is completed, all Series 2 Shares that have been tendered to Noranda on or before the third Business Day immediately prior to the date that the Control Transaction is completed and (y) thereafter, from time to time, on the third Business Day following the exercise of the Retraction Right, all Series 2 Shares that have been tendered to Noranda:

in each case, by payment of an amount in cash for each Series 2 Share redeemed equal to (i) if the redemption occurs on or before June 30, 2010, the Optional Redemption Price, and (ii) if the redemption occurs after June 30, 2010, the Basic Redemption Price (the <u>Retraction Price</u>). For purposes of determining the Retraction Price, the date fixed for redemption shall be the date upon which Common Shares are first taken up under a Take-over Bid or the date that a Control Transaction is contemplated, as applicable, and, thereafter, the third Business Day immediately following the date on which Series 2 Shares are tendered by a holder to Noranda pursuant to subsection 4(2)(a).

- (c) The provisions of section 3 shall apply to any redemption pursuant to this section 4 except to the extent that such provisions are inconsistent with the provisions of this section 4.
- (3) **Exception.** Notwithstanding anything contained in this section 4, Noranda shall not redeem any Series 2 Shares tendered to Noranda pursuant to the exercise of the Retraction Right upon the occurrence of an event contemplated under paragraph (ii) or (iii) of the definition of Change of Control Event if the applicable Take-over Bid or Control Transaction made or announced is withdrawn or terminated or expires without any person becoming an Acquiring Person and, upon

1R

any such withdrawal, termination or expiry, any notice delivered pursuant to subsection 4(2)(a) shall be null and void and of no further force or effect.

(4) **Notice.** Noranda shall give written notice of the occurrence of any Change of Control Event to all holders within five Business Days of the date on which Noranda first becomes aware of such occurrence.

# 5. Conversion into Common Shares at the Option of Noranda On the Final Redemption Date.

- (1) Conversion Right. Subject to section 7, applicable law and any requirement to obtain regulatory relief, Noranda, at its option (the <u>Conversion Right</u>), may, upon giving written notice (the <u>Conversion Notice</u>) at least 30 days but no more than 60 days prior to the Final Redemption Date, convert all or any part of the then outstanding Series 2 Shares into that number of freely-tradeable, fully paid and non-assessable Common Shares per Series 2 Share determined by dividing the aggregate of \$25.00, plus an amount equal to accrued and unpaid dividends up to but excluding the Final Redemption Date and calculated in accordance with subsection 2(1)(d) (the whole constituting the <u>Cash Conversion Price</u>), by the greater of \$2.00 (as adjusted from time to time by the Board of Directors acting in good faith to give effect to any split, division, consolidation, combination or similar change in the capital of Noranda that affects the number of Common Shares then outstanding) and 90% of the U.S. Dollar Equivalent of the Current Market Price as of the Final Redemption Date.
- (2)**Conversion Notice.** The Conversion Notice must set out the number of Series 2 Shares held by the person to whom it is addressed which are to be converted. The Conversion Notice also must advise the holder that the Common Shares will be registered in the name of the holder of Series 2 Shares to be converted unless the Transfer Agent receives from the holder, at least 10 Business Days before the Final Redemption Date, at any principal office of the Transfer Agent, written notice in form and substance satisfactory to the Transfer Agent directing Noranda to arrange for the registration of such Common Shares in some other name or names (the <u>Transferee</u> or the <u>Transferee</u> ) and stating the name or names (with addresses), accompanied by payment to the Transfer Agent of any transfer tax which may be payable by reason of the transfer and a written declaration, if required by any applicable law or by Noranda, as to the residence of the Transferee(s) and such other matters as may be required by law or requested by Noranda in order to determine the entitlement of the Transferee(s) to such Common Shares, in which case such Common Shares will be registered in the name or names so directed in the written notice. Such conversion will be deemed to have been made at 5:00 p.m. (Toronto time) on the Final Redemption Date, so that the rights of the holder of such Series 2 Shares as the holder thereof will cease at such time and the person or persons entitled to receive Common Shares upon such conversion will be treated for all purposes as having become the holder or holders of record of such Common Shares at such time.
- (3) **Pro Rata Conversion.** If less than all of the then outstanding Series 2 Shares are at any time to be converted at the option of Noranda, then: (i) the particular Series 2 Shares to be so converted will be selected on a pro rata basis; and (ii) any Series 2 Shares not converted pursuant to this section 5 will be redeemed pursuant to subsection 3(3).

- (4) **Prohibition on Conversion.** Noranda cannot exercise the Conversion Right hereunder if: (i) subject to Section 5(7), a Change of Control Event has occurred; (ii) an Extraordinary Dividend has been paid; or (iii) any one or more of the following events has occurred:
  - (a) the issuance or delivery of such Common Shares is prohibited pursuant to any agreement or arrangement entered into by Noranda to assure Noranda s solvency or continued operation;
  - (b) the issuance or delivery of such Common Shares is prohibited by law or by any regulatory or other authority having jurisdiction over Noranda and acting in conformity with law;
  - (c) for any reason beyond its control, Noranda is unable to issue Common Shares or is unable to deliver such Common Shares; or
  - (d) such Common Shares are not listed on each exchange on which the Common Shares are then listed for trading.
- (5) **Fractional Shares.** Where the aggregate number of Common Shares to be delivered to a holder of Series 2 Shares pursuant to Noranda s exercise of the Conversion Right includes a fraction of a Common Share, Noranda, in lieu of delivering such fractional share, will pay by cheque an amount equal to the balance of the Cash Conversion Price not otherwise satisfied by the delivery of the Common Shares.

# (6) Delivery of Shares and Share Certificates.

- (a) On any conversion of Series 2 Shares pursuant to the exercise of the Conversion Right, the share certificates for Common Shares resulting therefrom will be delivered in the name of the holder of the Series 2 Shares converted or, if applicable, in the name of the Transferee; provided that such holder of Series 2 Shares will (i) pay any applicable security transfer taxes including, without limitation, any documentary, stamp, transfer or other taxes that may be payable in respect of any transfer involved in the issuance or delivery of such shares to a person other than such holder of Series 2 Shares or (ii) will have evidenced to the satisfaction of Noranda that such taxes, if any, have been paid.
- (b) Subject to these Series 2 Provisions upon the exercise of the Conversion Right any holder of Series 2 Shares to be converted (or, if applicable, any Transferee) will be deemed to have become a holder of Common Shares of record for all purposes on the Final Redemption Date, notwithstanding any delay in the delivery of certificates representing the Common Shares for which such Series 2 Shares have been converted.
- (c) Upon exercise of the Conversion Right, Noranda is not required to issue Common Shares to any person whose address is in, or whom Noranda or its transfer agent has reason to believe is a resident of, any jurisdiction outside of Canada, to the extent that such issue would require compliance by Noranda with the securities or other laws of such jurisdiction.

Table of Contents 84

1**S** 

1T

(7) **Exception.** Notwithstanding section 5(4)(i), Noranda shall be entitled to exercise its Conversion Right hereunder in accordance with its terms if the applicable Change of Control Event prohibiting such conversion is the occurrence of (ii) or (iii) under the definition of Change of Control Event and the applicable Take-over Bid or Control Transaction made or announced is withdrawn or terminated or expires without any person becoming an Acquiring Person, upon any such withdrawal, termination or expiry.

## 6. Purchase for Cancellation

Subject to section 7 and to applicable law, Noranda may, at any time, if obtainable, purchase for cancellation, by private agreement or otherwise, at the lowest price or prices at which, in the opinion of the Board of Directors, such shares are obtainable, all or from time to time any part of the then outstanding Series 2 Shares.

# 7. Restrictions on Dividends and Retirement of Shares

- (1) <u>General Restrictions</u>. As long as any Series 2 Shares are outstanding, Noranda will not at any time, without the approval of the holders of the Series 2 Shares given as provided in section 11:
  - (a) declare, pay or set apart for payment any dividends (other than stock dividends payable in shares of Noranda ranking junior to the Series 2 Shares) on shares of Noranda ranking junior to the Series 2 Shares;
  - (b) except out of the net cash proceeds of a substantially concurrent issue of shares of Noranda ranking junior to the Series 2 Shares, redeem or call for redemption, purchase or otherwise pay off, retire or make any return of capital in respect of any shares of Noranda ranking as to capital junior to the Series 2 Shares;
  - (c) redeem or call for redemption, purchase or otherwise retire for value or make any return of capital in respect of less than all of the Series 2 Shares then outstanding;
  - (d) except pursuant to any purchase obligation, sinking fund, retraction privilege or mandatory redemption provisions attaching thereto, redeem or call for redemption, purchase or otherwise pay off, retire or make any return of capital in respect of any Preferred Shares, ranking on a parity with the Series 2 Shares; or
- (e) issue any additional Series 2 Shares or any shares ranking senior to or on a parity with the Series 2 Shares, unless, in each such case, all accrued dividends up to and including those payable on the Dividend Payment Date for the last completed period for which dividends are payable on the Series 2 Shares in respect of which the rights of holders have not been extinguished, and all dividends then accrued on all other shares ranking senior to or on a parity with the Series 2 Shares with respect to the payment of dividends up to the immediately preceding respective date or dates for payment in respect of which the rights of holders of those shares have not been extinguished, have been declared, paid or set apart for payment.

1U

(2) **Extraordinary Dividends.** As long as 123 Shares having an aggregate issue price of at least \$300 million remain outstanding, Noranda will not at any time, without the approval of the holders of the Series 2 Shares given as provided in section 10 declare, pay or set apart for payment any Extraordinary Dividends.

## 8. Voting Rights and Election of Directors

- (1) **Election of Directors.** The holders of Series 2 Shares together with the holders of Series 1 Shares and the holders of the Series 3 Shares will have at all times the right, voting collectively, to elect two directors to the Board of Directors at each meeting of shareholders of Noranda at which directors are to be elected. Forthwith upon the occurrence of a Board Event, the holders of the Series 2 Shares together with the holders of the Series 1 Shares and the holders of the Series 3 Shares will, but only for so long as a Board Event is continuing, be entitled, voting collectively, to elect three additional directors (for a total of five). Upon the occurrence of a Board Event, the Board of Directors will, at its election and subject to applicable law, either (i) appoint such additional number of directors to which the holders of the Series 2 Shares (together with the holders of the Series 1 Shares and holders of the Series 3 Shares) are entitled to elect as a result of such occurrence and as may be proposed by the Series 123 Directors, or (ii) forthwith call a special meeting of the applicable shareholders of Noranda for the purpose of electing such additional number of directors and, if necessary, increasing the maximum number of directors that may be elected. The holders of Series 2 Shares are entitled to receive notice of and to attend all annual and special meetings of the Shareholders of Noranda called for the purpose of electing directors as provided in this subsection 8(1) and, for such purpose only, to one vote in respect of each Series 2 Share held at all such meetings.
- (2) **Vacancies.** If a Series 123 Director ceases to be a director for any reason other than as a result of a Board Event ceasing to continue, the Board of Directors will fill the vacancy thereby created by appointing as soon as reasonably possible that individual who is nominated by the remaining Series 123 Directors, provided that if there are no remaining 123 Directors then in office, the Board of Directors will select the individuals to be appointed.
- (3) **Meeting of Shareholders.** Except as otherwise provided by law or as specifically provided herein and except for meetings of the holders of Junior Preference Shares as a class and meetings of all holders of Series 2 Shares as a series, the holders of the Series 2 Shares will not be entitled as such to receive notice of, or to attend, or to vote at, any meeting of shareholders of Noranda.
- (4) Meeting of Holders of Series 2 Shares as a Series or Junior Preference Shares as a Class. At any meeting of holders of Series 2 Shares as a series or together with one or more other series or Junior Preference Shares as a class, each such holder will be entitled to one vote in respect of each Series 2 Shares held.

# 9. Issue of Additional Common Shares

For greater certainty, nothing in these Series 2 Provisions will affect or restrict the right of Noranda to increase the number of the Common Shares or to issue additional Common Shares from time to time.

1V

# 10. Approval of Holders of Series 2 Shares

In addition to any other approvals required by law, the approval of all amendments to the rights, privileges, restrictions and conditions attaching to the Series 2 Shares as a series and any other approval to be given by the holders of Series 2 Shares as a series will be deemed to have been sufficiently given if given by a resolution carried by the affirmative vote of not less than 66²/3% of the votes cast at a meeting of the holders of Series 2 Shares duly called and held on not less than 21 days notice at which the holders of at least a majority of the outstanding Series 2 Shares are present or are represented by proxy, or if no quorum is present at such meeting, at an adjourned meeting at which the holders of the Series 2 Shares then present would form the necessary quorum. If at the meeting the holders of a majority of the outstanding Series 2 Shares are not present or represented by proxy within 30 minutes after the time appointed for the meeting, the meeting will be adjourned to such date, not less than 15 days afterwards, and to such time and place as the chairman may designate, and not less than 10 days written notice will be given of the adjourned meeting. Subject to the foregoing, the formalities to be observed in respect of the giving of notice of any meeting or adjourned meeting and the conduct of any such meeting will be those prescribed from time to time in the by-laws of Noranda with respect to meetings of shareholders.

# 11. Capital Reorganizations and Amalgamations of Noranda

In the event of a capital reorganization, merger or amalgamation of Noranda or comparable transaction affecting the Series 2 Shares, holders of the Series 2 Shares will be entitled to receive pursuant to the exercise of the Conversion Right, and each such holder will accept, the number of shares or other securities or consideration of Noranda or a corporation resulting, surviving or continuing from the capital reorganization, merger, amalgamation or comparable transaction that such holder would have received had such holder been a holder of Common Shares immediately prior to the record date of the capital reorganization, merger, amalgamation or comparable transaction.

## 12. Notices

- (1) **Notice to Noranda.** Subject to applicable law, any notice, request or other communication to be given to Noranda by a holder of Series 2 Shares must be in writing and will be valid and effective if given by mail (postage prepaid) or by electronic communication or by delivery to the registered office of Noranda and addressed to the attention of the Secretary. Any such notice, request or other communication, if given by mail, electronic communication or delivery, will be deemed to have been given and received only on actual receipt by Noranda.
- (2) **Presentation and Surrender of Certificates.** Any presentation and surrender by a holder of Series 2 Shares to Noranda or the Transfer Agent of certificates representing Series 2 Shares in connection with the redemption of Series 2 Shares must be made by registered mail (postage prepaid) or by delivery to the registered office of Noranda or to such office of the Transfer Agent as may be specified by Noranda, in each case addressed to the attention of the Secretary of Noranda. Any such presentation and surrender of certificates will be deemed to have been made and to be effective only on actual receipt by Noranda or the Transfer Agent, as the case may be. Any such presentation and surrender of certificates made by registered mail will be at the sole risk of the holder mailing the same.

1W

Notice to Holders of Series 2 Shares. Subject to applicable law, any notice, including but not limited to the Redemption Notice and the Conversion Notice, request or other communication to be given to a holder of Series 2 Shares by or on behalf of Noranda must be in writing and will be valid and effective if given by ordinary unregistered first class mail (postage prepaid) or by electronic communication or by delivery to the address of the holder recorded in the securities register of Noranda or, in the case of joint holders, to the address of that one whose name appears first in the register of holders as one of such joint holders. Any such notice, request or other communication, if given by mail, will be deemed to have been given and received on the fifth Business Day following the date of mailing and, if given by electronic communication or by delivery, will be deemed to have been given and received on the date of electronic communication or delivery. Accidental failure or omission to give any notice, request or other communication to one or more holders of Series 2 Shares, or any defect in such notice, will not invalidate or otherwise alter or affect any action or proceeding to be taken by Noranda pursuant to that notice, request or other communication.

## 13. Tax Election

Noranda will elect, in the manner and within the time provided under Section 191.2 in Part VI.1 of the Tax Act or any successor or replacement provision of similar effect, and take all other necessary action under the Tax Act, to pay or cause payment of the tax under Part VI.1 of the Tax Act at a rate such that the corporate holders of the Series 2 Shares will not be required to pay tax under Part IV.1 of the Tax Act on dividends received on such shares.

## 14. Return of Unclaimed Funds to Noranda

Subject to applicable laws, Noranda will have the right, with respect to any funds deposited by Noranda to any chartered bank or trust company in respect of amounts due to holders of Series 2 Shares, on or after the second anniversary date of the deposit of such funds to any chartered bank or trust company, to require that such chartered bank or trust company return to Noranda any funds which remain unclaimed by holders of the Series 2 Shares. If at any time following the return of the unclaimed funds to Noranda, either Noranda or the chartered bank or trust company receives a request from a holder of Series 2 Shares for the holder s unclaimed entitlement to any amount due to the holder in respect of the holder s Series 2 Shares, Noranda will promptly arrange for the payment of such amount to the holder.

## 15. Wire or Electronic Transfer of Funds

Notwithstanding any other right, privilege, restriction or condition attaching to the Series 2 Shares, Noranda may, at its option, make any payment due to a holder of Series 2 Shares hereunder by way of a wire or electronic transfer of funds to each registered holder of Series 2 Shares. In the event that a payment is made by way of a wire or electronic transfer of funds, Noranda will be responsible for any applicable charges or fees relating to the making of such transfer. As soon as practicable following the determination by Noranda that a payment is to be made by way of a wire or electronic transfer of funds, Noranda will notify each registered holder of Series 2 Shares at the address of such holder as it appears on the register of holders. Such notice by Noranda will request that each registered holder of Series 2 Shares provide the particulars of an account of such holder with a chartered bank or trust company in Canada to which the wire or electronic transfer of funds

1X

will be directed. In the event that Noranda does not receive account particulars from a registered holder of Series 2 Shares prior to the date such payment is to be made, Noranda will deposit the funds otherwise payable to such holder in a special account or accounts in trust for each respective holder. The making of a payment by way of a wire or electronic transfer of funds or, in the case where a registered holder of the Series 2 Shares has not provided Noranda with account particulars for a wire or electronic transfer of funds, the deposit by Noranda of the funds otherwise payable to such holder in a special account or accounts in trust for such holder, will be deemed to constitute payment by Noranda on the date thereof and will satisfy and discharge all liabilities of Noranda for such payment to the extent of the amount represented by such transfer.

## 16. Additional Series

As long as any of the 123 Shares are outstanding, Noranda may not issue any additional Preferred Shares, other than Cumulative Redeemable Preferred Shares, Series F issuable upon the conversion of Cumulative Redeemable Preferred Shares, Series G and Cumulative Redeemable Preferred Shares, Series G issuable upon the conversion of Cumulative Redeemable Preferred Shares, Series F, without the approval of the holders of the 123 Shares, expressed by resolution of the holders of all such shares voting together. For this purpose, the approval to be given will be deemed to have been sufficiently given if given by a resolution carried by the affirmative vote of not less than 66²/3% of the votes cast at a meeting of the holders of 123 Shares duly called and held on not less than 21 days notice at which the holders of at least a majority of the outstanding 123 Shares are present or are represented by proxy, or if no quorum is present at such meeting, at an adjourned meeting at which the holders of the 123 Shares then present would form the necessary quorum. If at the meeting the holders of a majority of the outstanding 123 Shares are not present or represented by proxy within 30 minutes after the time appointed for the meeting, the meeting will be adjourned to such date, not less than 15 days afterwards, and to such time and place as the chairman may designate, and not less than 10 days written notice will be given of the adjourned meeting. Subject to the foregoing, the formalities to be observed in respect of the giving of notice of any meeting or adjourned meeting and the conduct of any such meeting will be those prescribed from time to time in the by-laws of Noranda with respect to meetings of shareholders.

1A

# Number and Designation of, and Rights, Privileges, Restrictions and Conditions Attaching to, the Junior Preference Shares, Series 3

The third series of Junior Preference Shares of Noranda shall consist of 10,000,000 shares designated as Junior Preference Shares, Series 3 (the <u>Series 3 Shares</u>) with an issue price of \$25.00 per share (the <u>Issue Price</u>) and, in addition to the rights, privileges, restrictions and conditions attaching to the Junior Preference Shares as a class, shall have attached thereto the following rights, privileges, restrictions and conditions (the <u>Series 3 Provisions</u>):

## 1. Interpretation

(1) **Defined Terms.** In these Series 3 Provisions:

**Acquiring Person** means any person who is the Beneficial Owner of a Significant Interest in the outstanding Voting Shares; provided, however, that the term Acquiring Person shall not include:

- (a) Noranda or any Subsidiary of Noranda;
- (b) Brascan:
- (c) any person who becomes the Beneficial Owner of a Significant Interest in the outstanding Voting Shares as a result of one or any combination of:
  - (A) a Voting Share Reduction which, by reducing the number of Voting Shares outstanding, increases the percentage of Voting Shares Beneficially Owned by such person to a Significant Interest in the Voting Shares then outstanding,
  - (B) a Pro Rata Acquisition,
  - (C) an Exempt Acquisition,
  - (D) a Convertible Security Acquisition, or
  - (E) a 123 Share Acquisition;

provided further, however, that if a person shall become the Beneficial Owner of a Significant Interest in the Voting Shares then outstanding by reason of any one or a combination of a Voting Share Reduction, a Pro Rata Acquisition, an Exempt Acquisition, a Convertible Security Acquisition or a 123 Share Acquisition and thereafter becomes the Beneficial Owner of more than an additional 1% of the outstanding Voting Shares (other than pursuant to a Voting Share Reduction, a Pro Rata Acquisition, an Exempt Acquisition, a Convertible Security Acquisition), or a 123 Share Acquisition), then as of the date that such person becomes the Beneficial Owner of such additional Voting Shares, such person shall become an Acquiring Person;

- (d) any person who, on the Issue Date of the 123 Shares, is the Beneficial Owner of a Significant Interest in the outstanding Voting Shares, provided that if such person thereafter becomes the Beneficial Owner of more than an additional 1% of the outstanding Voting Shares (other than pursuant to a Voting Share Reduction, a Pro Rata Acquisition, an Exempt Acquisition, a Convertible Security Acquisition or a 123 Share Acquisition), then as of the date that such person becomes the Beneficial Owner of such additional Voting Shares, such person shall become an Acquiring Person; and
- (e) an underwriter or member of a banking or selling group that acquires Voting Shares from Noranda in connection with a distribution of securities pursuant to a prospectus or by way of private placement.
   Adjusted Net Worth means, as at the end of any fiscal quarter, the aggregate value of:
- (a) the outstanding share capital for all shares ranking junior to the Junior Preference Shares;
- (b) without duplication, any surplus, whether contributed or capital;
- (c) retained earnings; and
- (d) consolidated non-controlling interest, all as set forth in Noranda s consolidated balance sheet for such fiscal quarter filed in accordance with applicable securities laws.

**Affiliate**, when used to indicate a relationship with a specified corporation, means a person that directly, or indirectly through one or more controlled intermediaries, controls, or is controlled by, or is under common control with, such specified corporation.

**Associate** , when used to indicate a relationship with a specified person, means (i) a spouse of such specified person, (ii) any person of either sex with whom such specified person is living in a conjugal relationship outside marriage or (iii) any relative of such specified person or of a person mentioned in clauses (i) or (ii) of this definition if that relative has the same residence as the specified person.

**Basic Redemption Price** has the meaning attributed to it in subsection 3(2).

A person shall be deemed the Beneficial Owner, and to have Beneficial Ownership of and to Beneficially Own:

- (i) any securities of which such person or any of such person s Affiliates or Associates is the owner at law or in equity;
- (ii) any securities as to which such person or any of such person s Affiliates or Associates has the right to become the owner at law or in equity, where such right is exercisable immediately or within 60 days of the date of the determination of

Table of Contents 91

1B

1C

Beneficial Ownership and whether or not on condition or the occurrence of any contingency or payment of instalments, upon the exercise of any conversion right, exchange right or purchase right attaching to Convertible Securities, or pursuant to any agreement, arrangement, pledge or understanding, written or oral (other than customary agreements with and between underwriters and/or banking group and/or selling group members with respect to a distribution of securities pursuant to a prospectus or by way of private placement and other than pursuant to pledges of securities in the ordinary course of business); and

- (iii) any securities which are Beneficially Owned within the meaning of clauses (i) or (ii) of this definition by any other person with which, and in respect of which securities, such person is acting jointly or in concert;
  - provided, however, that a person shall not be deemed the Beneficial Owner of, or to have Beneficial Ownership of, or to Beneficially Own, any security:
  - (A) by reason of such security having been deposited or tendered pursuant to a Take-over Bid made by, or Control Transaction involving, such person or any of such person s Affiliates or Associates or any other person referred to in clause (iii) of this definition until the earlier of such deposited or tendered security being accepted unconditionally for payment or exchange or being taken up and paid for;
  - (B) by reason of such person, any of such person s Affiliates or Associates or any other person referred to in clause (iii) of this definition holding such security, if:
    - (I) the ordinary business of the person (in this definition, a <u>Manager</u>) includes the management of investment funds for others and such security is held by the Manager in the ordinary course of such business in the performance of such Manager s duties for the account of any other person (in this definition, a <u>Client</u>) including non-discretionary accounts held on behalf of a Client by a broker or dealer registered under applicable law;
    - (II) the person (in this definition, a <u>Trust Company</u>) is licensed to carry on the business of a trust company under applicable law and, as such, acts as a trustee or administrator or in a similar capacity in relation to the estates of deceased or incompetent persons (each, in this definition, an Estate Account) or in relation to other accounts (each, in this definition, an Other Account) and holds such security and is acting in the ordinary course of such duties for the Estate Account or for such Other Accounts;
    - (III) such person is a Crown agent or agency (in this definition, the <u>Crown Agent</u>);

1D

- (IV) the person is established by statute for purposes that include, and the ordinary business or activity of such person (in this definition, a <u>Statutory Body</u>) includes, the management of investment funds for employee benefit plans, pension plans, insurance plans of various public bodies and the Statutory Body holds such security for the purposes of its activities as such; or
- (V) the person (in this definition, an <u>Administrator</u>) is the administrator or trustee of one or more pension funds or plans (each, in this definition, a <u>Plan</u>) registered under the laws of Canada or any province thereof or the similar laws of the jurisdiction by which such Plan is governed or is such a Plan and the Administrator or Plan holds such security for the purposes of its activities as such;

but only if the Manager, the Trust Company, the Crown Agent, the Statutory Body, the Administrator or the Plan, as the case may be, is not then making and has not announced a current intention to make a Take-over Bid or a Control Transaction, other than an Offer to Acquire Voting Shares or other securities pursuant to a distribution by Noranda, or by means of ordinary market transactions (including prearranged trades entered into in the ordinary course of business of such person) executed through the facilities of a stock exchange or an organized over-the-counter market, alone or by acting jointly or in concert with any other person;

## (C) because such person:

- (I) is a Client of the same Manager as another person on whose account the Manager holds such security,
- (II) has an Estate Account or an Other Account with the same Trust Company as another person on whose account the Trust Corporation holds such security, or
- (III) is a Plan with the same Administrator as another Plan on whose account the Administrator holds such security,

# (D) because such person:

- (I) is a Client of a Manager and such security is owned at law or in equity by the Manager,
- (II) has an Estate Account or an Other Account with a Trust Company and such security is owned at law or in equity by the Trust Company, or
- (III) is a Plan and such security is owned at law or in equity by the Administrator of the Plan; or

1E

(E) because such person is the registered holder of securities as a result of carrying on the business of, or acting as nominee for, a securities depositary.

**Board Event** means any one of the following events:

- (a) if four quarterly dividends on the Series 3 Shares are in arrears, whether or not such dividends have been declared and whether or not there are any monies of Noranda properly applicable to the payment of dividends; or
- (b) if the Adjusted Net Worth for any fiscal quarter is less than \$2,500,000,000.

**Board of Directors** means the board of directors of Noranda.

**Brascan** means Brascan Corporation, any Affiliate of Brascan Corporation or any party acting jointly or in concert with Brascan Corporation or any of its Affiliates.

**Business Day** means a day other than a Saturday, a Sunday or any other day which is a statutory or civic holiday in Toronto, Ontario.

**Cash Conversion Price** has the meaning attributed to it in subsection 5(1).

## **Change of Control Event** means the occurrence of:

- (i) the first date of public announcement (which, for purposes of this definition, shall include, without limitation, a report filed pursuant to section 101 of the *Securities Act* (Ontario)) by Noranda or an Acquiring Person of facts indicating that an Acquiring Person has become such;
- (ii) the date of the commencement of, or first public announcement of the intent of any person (other than Brascan, Noranda or any Subsidiary of Noranda) to commence a Take-over Bid (other than by Brascan, Noranda or any Subsidiary of Noranda); or
- (iii) the date of the commencement of, or first public announcement of the intent of any person (other than Brascan) to commence a Control Transaction (other than by Brascan, Noranda or any Subsidiary of Noranda).

**Change of Control Retraction Date** means the date on which the Series 3 Shares are redeemed pursuant to subsection 4(2)(b)(i), (ii) or (iii), as applicable.

**Common Shares** means the common shares of Noranda, and any other securities into which those shares may be changed or for which those shares may be exchanged (whether or not Noranda is the issuer of such other securities).

**Control Transaction** means any merger, arrangement, amalgamation, reorganization, acquisition, material sale of shares or rights or interests therein or thereto or similar transactions involving Noranda, which will result in any person Beneficially Owning, whether directly as a result of such transaction or together with the Common Shares that such person Beneficially Owned before the completion of such transaction, a Significant Interest in Noranda.

**controlled**: a body corporate is controlled by another person if:

- (i) securities entitled to vote in the election of directors carrying more than 50% of the votes for the election of directors are held, directly or indirectly, by or on behalf of the other person; and
- (ii) the votes carried by such securities are entitled, if exercised, to elect a majority of the board of directors of such body corporate,

and controls, controlling and under common control with shall be interpreted accordingly.

**Conversion Notice** has the meaning attributed to it in subsection 5(1).

**Conversion Right** has the meaning attributed to it in subsection 5(1).

Convertible Securities means, at any time, any securities issued by Noranda (including rights, warrants and options but other than the 123 Shares) carrying any purchase, exercise, conversion or exchange right, pursuant to which the holder thereof may acquire Voting Shares or other securities convertible into or exercisable or exchangeable for Voting Shares (in each case, whether such right is exercisable immediately or after a specified period and whether or not on condition or the happening of any contingency).

**Convertible Security Acquisition** means the acquisition of Voting Shares upon the exercise of Convertible Securities acquired by a person pursuant to an Exempt Acquisition or a Pro Rata Acquisition.

Current Market Price means, at any date, the volume weighted average trading price of Common Shares on the Toronto Stock Exchange (or, if not then listed on that exchange, on another exchange or market chosen by the Board of Directors in their sole discretion on which Common Shares are then traded), for a period of 20 consecutive Trading Days ending on the fourth day immediately prior to that date or, if such fourth day is not a Trading Day, on the immediately preceding Trading Day.

**Dividend** and **Dividends** have the respective meanings attributed to them in subsection 2(1)(a).

**Dividend Payment Date** means the last day of the months of March, June, September and December in each year.

**Dividend Period** means the Initial Dividend Period and, thereafter, the period from and including a Dividend Payment Date to but excluding the next succeeding Dividend Payment Date.

**Exempt Acquisition** means an acquisition of Voting Shares or Convertible Securities (i) pursuant to a distribution of Voting Shares or Convertible Securities (and the conversion or exchange of such Convertible Securities) made by Noranda pursuant to a prospectus or private placement provided that the person does not acquire a greater percentage of the securities offered in the distribution than the percentage of Voting Shares Beneficially Owned by that person

Table of Contents 95

1F

# Edgar Filing: NORANDA INC - Form SC 13D/A

immediately prior to the distribution, or (ii) pursuant to an amalgamation, merger or other statutory procedure requiring shareholder approval.

**Extraordinary Dividend** means:

- (i) any dividend, other than a stock dividend paid wholly in Common Shares, declared or paid on the Common Shares that is, when taken together with the amount or value of all other dividends declared or paid in the 12 month period preceding the date of declaration of the dividend (the measurement period ) more than 125% of the aggregate amount or value of the dividends declared or paid on the Common Shares, other than stock dividends paid wholly in Common Shares, during the 12 month period ended on the day prior to the measurement period (or, where there were no dividends paid in the 12 month period ended on the day prior to the measurement period, the period of 12 months ending on the date on which Noranda last paid a dividend, other than a stock dividend paid wholly in Common Shares, on the Common Shares); or
- (ii) any dividend, other than a stock dividend paid wholly in Common Shares, declared or paid on the Common Shares that results in the Noranda Annual Dividend Rate exceeding a 10% compound annual growth rate, measured from and following the Issue Date. For this purpose, the Noranda Annual Dividend Rate is initially the aggregate amount or value of all dividends declared or paid on the Common Shares, other than stock dividends paid wholly in Common Shares, in the 12 month period immediately preceding the Issue Date and thereafter is the aggregate amount or value of all dividends declared or paid in any 12 month period immediately preceding the date of declaration of any other dividend on the Common Shares, together with the dividend then being declared; or
- (i) any other special dividend on, or distribution with respect to, the Common Shares which is, by its terms or declared intent, declared and paid outside the normal operations or normal dividend procedures of Noranda.
   Final Redemption Date means June 30, 2015.

**freely-tradeable** means, in respect of a Common Share, that (i) the Common Share is transferable and not subject to any escrow requirements or to any cease trade order imposed by a Canadian securities regulatory authority, and (ii) all hold periods imposed by Canadian securities legislation before such Common Share can be traded without a prospectus or in reliance on a prospectus exemption have expired.

**Initial Dividend Period** means the period from and including the Issue Date to but excluding the first Dividend Payment Date after the Issue Date.

**Issue Date** means the date any of the Series 3 Shares are first issued.

**Issue Price** has the meaning attributed to it in the introductory paragraph to these rights, privileges, restrictions attaching to the Series 1 Shares.

**Noranda** means Noranda Inc. and any corporation resulting, surviving or continuing from the merger, arrangement, amalgamation or reorganization of Noranda Inc., or a comparable transaction.

Table of Contents 97

1G

1H

# **Offer to Acquire** shall include:

- (i) an offer to purchase, or a solicitation of an offer to sell, Voting Shares, and
- (ii) an acceptance of an offer to sell Voting Shares, whether or not such offer to sell has been solicited, or any combination thereof, and the person accepting an offer to sell shall be deemed to be making an Offer to Acquire to the person that made the offer to sell.

**Offeror** means a person who has announced an intention to make or who has made a Take-over Bid (excluding an Offer to Acquire Common Shares or other securities made by a Manager, Trust Company, Crown Agent, Statutory Body, Administrator or Plan referred to in clause (B) of the definition of Beneficial Owner pursuant to a distribution by Noranda or by means of ordinary market transactions (including pre-arranged trades entered into in the ordinary course of business of such person) in the circumstances contemplated in said clause (B)), but only so long as the Take-over Bid so announced or made has not been withdrawn or terminated or has not expired.

**Offeror s Securities** means the aggregate of the Voting Shares Beneficially Owned by an Offeror on the date of an Offer to Acquire by an Offeror.

**123 Shares** means, collectively, the Series 1 Shares, the Series 2 Shares and the Series 3 Shares.

**123 Share Acquisition** means, in respect of holders of the 123 Shares, (i) the payment of a Dividend by the issuance of Common Shares pursuant to subsection 2(2)(a) or (ii) the issuance of Common Shares upon exercise by Noranda of its conversion right pursuant to section 5.

**Optional Redemption Price** has the meaning attributed to it in subsection 3(1).

**Pro Rata Acquisition** means an acquisition by a person of Voting Shares or Convertible Securities (i) as a result of a stock dividend, a stock split or other event pursuant to which such person receives or acquires Voting Shares or Convertible Securities on the same pro rata basis as all other holders of Voting Shares of the same class or series of Noranda, (ii) pursuant to a regular dividend reinvestment or other plan of Noranda made available by Noranda to the holders of Voting Shares where such plan permits the holder to direct that the dividends paid in respect of such Voting Shares be applied to the purchase from Noranda of further securities of Noranda, or (iii) pursuant to the receipt and/or exercise of rights (other than the 123 Shares) issued by Noranda to all of the holders of a series or class of Voting Shares on a pro rata basis to subscribe for or purchase Voting Shares or Convertible Securities, provided that the person does not acquire a greater percentage of the securities issuable on exercise of such rights than the percentage of Voting Shares Beneficially Owned by that person immediately prior to the commencement of the offering of rights and that such rights are acquired directly from Noranda and not from any other person.

**Redemption Date** means (i) in the case of a mandatory redemption pursuant to subsection 3(3), the Final Redemption Date, (ii) in the case of a redemption in connection with an exercise of the Retraction Right, the Change of Control Retraction Date, and (iii) in all other cases, the date on which the redemption is to occur.

1I

**Redemption Notice** has the meaning attributed to it in subsection 3(5).

**Redemption Price** means (i) if the redemption occurs on or before June 30, 2013, the Optional Redemption Price, and (ii) if the redemption occurs after June 30, 2013, the Basic Redemption Price.

**Re-Set Dividend Rate** means the greater of (i) 6.50%, and (ii) the Ten Year U.S. Treasury Bond Yield plus 2.35%.

**Re-Set Dividend Period** means, initially, the period from and including June 30, 2010 to but excluding June 30, 2012 and, thereafter, the period from and including June 30, 2012 to but excluding June 30, 2014 and the period from and including June 30, 2014 to but excluding June 30, 2015, respectively.

**Retraction Price** has the meaning attributed to it in subsection 4(2)(b).

**Retraction Right** has the meaning attributed to it in subsection 4(1).

**Series 1 Shares** means the first series of Junior Preference Shares of Noranda.

**Series 123 Directors** means, at any time, those directors of Noranda most recently elected by or appointed on behalf of the holders of the Series 1 Shares, Series 2 Shares and Series 3 Shares, collectively.

**Series 2 Shares** means the second series of Junior Preference Shares of Noranda, designated as Junior Preference Shares, Series 2.

**Series 3 Provisions** has the meaning attributed to it in the introductory paragraph to these rights, privileges, restrictions and conditions attaching to the Series 3 Shares.

**Series 3 Shares** has the meaning attributed to it in the introductory paragraph to these Series 3 Provisions.

**Significant Interest**: a person has a Significant Interest in the Voting Shares where that person Beneficially Owns 30% or more of the outstanding Voting Shares.

**Subsidiary**: a body corporate is a Subsidiary of another body corporate if:

- (i) it is controlled by (A) that other, or (B) that other and one or more bodies corporate, each of which is controlled by that other, or (C) two or more bodies corporate, each of which is controlled by that other; or
- (ii) it is a Subsidiary of a body corporate that is that other s Subsidiary.

**Take-over Bid** means an Offer to Acquire Voting Shares, where the Voting Shares subject to the Offer to Acquire and the Offeror s Securities, constitute a Significant Interest in the outstanding Voting Shares at the date of the Offer to Acquire.

**Tax Act** means the *Income Tax Act* (Canada).

**Ten Year U.S. Treasury Bond Yield** means on any date shall mean the average of the yields determined by two registered Canadian investment dealers, selected by the board of directors of Noranda, as being the yield to maturity on such date compounded semi-annually and calculated in accordance with generally accepted financial practice, which a non-callable United States Treasury Bond would carry if issued in U.S. dollars in the United States at 100% of its principal amount on such date with a term to maturity of ten years.

**Trading Day** means with respect to any stock exchange or over-the-counter market, a day on which shares may be traded through the facilities of such stock exchange or in such over-the-counter market, and otherwise means a day on which shares may be traded through the facilities of the principal stock exchange on which the Common Shares are listed (or, if the Common Shares are not listed on any stock exchange, then in the over-the-counter market).

**Transfer Agent** means Noranda, or such other person as from time to time may be the registrar and transfer agent for the Series 3 Shares.

**Transferee** and **Transferees** have the meanings attributed to them in subsection 5(2).

- **U.S.** Canadian Exchange Rate means, on any date:
  - (i) if, on such date, the Bank of Canada sets an average noon spot rate of exchange for the conversion of one Canadian dollar into United States dollars, such rate; or
  - (ii) in any other case, the rate for such date for the conversion of one Canadian dollar into United States dollars calculated in such manner as may be determined by the Board of Directors from time to time acting in good faith.
- **U.S. Dollar Equivalent** of any amount which is expressed in Canadian dollars means, on any date, the United States dollar equivalent of such amount determined by multiplying such amount by the U.S. Canadian Exchange Rate in effect on such date.

**Voting Share Reduction** means an acquisition or a redemption by Noranda of Voting Shares.

**Voting Shares** means collectively the Common Shares and any other shares in the capital stock or voting interests of Noranda entitled to vote generally in the election of directors.

- (2) **Ranking of Shares.** The expressions in priority to , on a parity with , ranking equally with , ranking prior to ranking junior to and similar expressions refer to the order of priority only in payment of dividends or in the distribution of assets in the event of any liquidation, dissolution of winding-up of Noranda, whether voluntary or involuntary, or any other distribution of the assets of Noranda for the purpose of winding-up its affairs.
- (3) **Holder.** References to a <u>holder</u> in these Series 3 Provisions mean the registered owner of such shares.
- (4) **Extended Meanings.** References in these Series 3 Provisions to words importing the singular number only include the plural and *vice versa*, words importing any gender include all genders and words importing persons include individuals, corporations, limited and unlimited

Table of Contents 100

1J

1K

liability companies, general and limited partnership, associations, trusts, unincorporated organizations, joint ventures and governmental authorities. The term including means including without limiting the generality of the foregoing .

- (5) **References to Statutes.** Except as otherwise noted herein, reference to any statute in these Series 3 Provisions is to that statute as in force from time to time, including any regulations, rules, policy statements or guidelines made under that statute, and includes any statute which may be enacted in substitution of that statute.
- (6) **References to Sections**. Reference in these Series 3 Provisions to any section or subdivision is to the designated section or subdivision of these Series 3 Provisions.
- (7) Acting Jointly or in Concert. For purposes of these Series 3 Provisions, a person is acting jointly or in concert with every other person who is a party to any agreement, commitment or understanding (whether formal or informal and whether or not in writing) with the first mentioned person to acquire or offer to acquire or vote Voting Shares (other than (i) in the case of agreements to acquire Voting Shares customary agreements with and between underwriters and/or banking group and/or selling group members with respect to a distribution of securities pursuant to a prospectus or by way of private placement and other than pursuant to pledges of securities in the ordinary course of business and (ii) in the case of agreements to vote Voting Shares, agreements to vote for a limited purpose and without the intention of acquiring or exercising control).

## (8) Other Payment Matters.

- (a) If any date on which any Dividend on the Series 3 Shares is payable or on or by which any other action is required to be taken by Noranda under these Series 3 Provisions is not a Business Day, then the Dividend will be payable, or such other action will be required to be taken, on or by the next day that is a Business Day.
- (b) In the event of the non-receipt of a cheque by a holder of Series 3 Shares entitled to the cheque, or the loss or destruction of the cheque, Noranda, on being furnished with reasonable evidence of non-receipt, loss or destruction, and an indemnity reasonably satisfactory to Noranda, will issue to the holder a replacement cheque for the amount of the original cheque.
- (c) Notwithstanding any other provision of these Series 3 Provisions, Noranda may deduct or withhold from any payment, distribution, issuance or delivery to be made pursuant to these share terms any amounts required or permitted by law to be deducted or withheld from any such payment, distribution, issuance or delivery and will remit such amounts to the relevant tax authority as required. Where the cash component of any payment, distribution, issuance or delivery to be made pursuant to these share terms is less than the amount that Noranda is so required or permitted to deduct or withhold, Noranda will be permitted to deduct and withhold from any non-cash payment, distribution, issuance or delivery to be made pursuant to these share terms any amounts required or permitted by law to be deducted or withheld from any such payment, distribution, issuance or delivery and to dispose of such property in order to remit any amount required to be remitted to any relevant tax authority.

1L

Notwithstanding the above, the amount of any payment, distribution, issuance or delivery made pursuant to these Series 3 Provisions will be considered to be the amount of the payment, distribution, issuance or delivery plus any amount deducted or withheld pursuant to this subsection 1(8)(c).

# (9) Currency.

Except as otherwise provided in these Series 3 Provisions, all references to currency herein are to lawful money of the United States.

## (10) **Designation Under the Tax Act.**

The amount of is specified in respect of each Series 3 Share for the purposes of subsection 191(4) of the Tax Act.

# 2. Dividends

## (1) Payment of Dividends.

- (a) Until June 30, 2010, the holders of Series 3 Shares will be entitled to receive and Noranda will pay thereon, if, as and when declared by the Board of Directors out of the assets of Noranda properly applicable to the payment of dividends, fixed preferential cumulative cash dividends (each a <u>Dividend</u> and collectively, the <u>Dividends</u>) in an amount per Series 3 Share per annum equal to \$1.625, payable quarterly, in equal instalments of \$0.4063 per Series 3 Share, with respect to each Dividend Period other than the Initial Dividend Period, on the Dividend Payment Date immediately following the end of each such Dividend Period.
- (b) The dividend payable in respect of the Initial Dividend Period, if declared by the Board of Directors, will be payable on June 30, 2005 in the amount per Series 3 Share obtained when:
  - (i) \$1.625 is multiplied by the number of days beginning on the Issue Date up to but excluding June 30, 2005;
  - (ii) the result in paragraph (i) is divided by 365.
- (c) During the Re-Set Dividend Period, the Series 3 Shares will be entitled to receive and Noranda will pay thereon, if, as and when declared by the Board of Directors out of the assets of Noranda properly applicable to the payment of dividends, fixed preferential cumulative cash dividends in an amount per Series 3 Share per annum equal to the Re-Set Dividend Rate, payable quarterly, in equal instalments per Series 3 Share, with respect to each Dividend Period, on the Dividend Payment Date immediately following the end of each such Dividend Period.
- (d) In any case where Dividends are payable for a period that ends on a date other than a Dividend Payment Date on which a Dividend was paid, Dividends will be paid in an amount per Series 3 Share obtained when:

(i) \$1.625 is multiplied by the number of days beginning on the date of the immediately preceding Dividend

(ii) the result in paragraph (i) is divided by 365.

Payment Date up to but excluding the date fixed for payment; and

(e) If on any Dividend Payment Date the dividend payable on such date is not paid in full on all the Series 3 Shares then issued and outstanding, such dividend, or the unpaid part thereof, will be paid at a subsequent date or dates determined by the Board of Directors on which Noranda has sufficient monies properly applicable to the payment of such dividends. The holders of Series 3 Shares will not be entitled to any dividends other than or in excess of the preferential cumulative dividends hereinbefore provided.

# (2) Method of Payment.

- (a) Dividends on the Series 3 Shares will (other than in the case of redemption, in which case payment of dividends will be made on surrender of the certificate representing the Series 3 Shares to be redeemed) be paid by cheque, in lawful money of Canada, payable at par at any branch in Canada of a chartered bank or trust company, in the amount, rounded to the nearest whole cent (\$0.01) of the applicable dividend or, at the election of Noranda and subject to applicable law and any requirement to obtain regulatory relief, by issuing freely-tradeable, fully paid and non-assessable Common Shares. The mailing by ordinary unregistered first class prepaid mail of such a cheque or share certificate, as applicable, to a holder of Series 3 Shares to the address of the holder as it appears on the register of holders, or, in the case of joint holders, to the address of that one whose name appears first in the register of holders as one of such joint holders on or before the applicable Dividend Payment Date will be deemed to be payment and will satisfy and discharge all liabilities for dividends payable on that Dividend Payment Date to the extent of the amount represented by the cheque (plus any tax required to be deducted and withheld from the payment) or, calculated in accordance with subsection 2(2)(b), such share certificate unless, in case of payment by cheque, such cheque is not paid on due presentation. Dividends represented by a cheque which has not been duly presented for payment within two years after it was issued or that otherwise remain unclaimed for a period of two years from the date on which they were declared to be payable and set apart for payment will be forfeited to Noranda. Noranda may pay dividends in any other manner as it may agree with any particular holder.
- (b) If Noranda elects to pay a Dividend by issuing Common Shares in accordance with subsection 2(2)(a), the number of Common Shares to be issued as payment of a Dividend will be equal to the quotient obtained by dividing (i) the Dividend payable by (ii) 95% of the U.S. Dollar Equivalent of the Current Market Price as of the Dividend Payment Date, less the amount of any tax required to be deducted and withheld by Noranda. In any case where the aggregate number of Common Shares to be issued to a holder on payment of a Dividend pursuant to this section includes a fraction of a Common Share, Noranda, in lieu of delivering a fractional share, will pay by cheque an amount equal to the balance of the Dividend payable not otherwise

Table of Contents 103

1**M** 

satisfied by the delivery of Common Shares. Noranda cannot pay a Dividend by issuing Common Shares if one or more of the following events has occurred:

- (i) the issuance or delivery of such Common Shares is prohibited pursuant to any agreement or arrangement entered into by Noranda to assure Noranda s solvency or continued operation;
- (ii) the issuance or delivery of such Common Shares is prohibited by law or by any regulatory or other authority having jurisdiction over Noranda and acting in conformity with law;
- (iii) for any reason beyond its control, Noranda is unable to issue Common Shares or is unable to deliver such Common Shares; or
- (iv) such Common Shares are not listed on each exchange on which the Common Shares are then listed for trading.

# 3. Redemption

- (1) **Redemption On or Before June 30, 2013.** On or before June 30, 2013, but subject to section 7 and to applicable law, Noranda, may upon giving notice as hereinafter provided, redeem at any time the whole or from time to time any part of the then outstanding Series 3 Shares from, subject to subsection 3(4), any one or more of the holders thereof as the Board of Directors may in its sole discretion determine by payment of an amount in cash for each share to be redeemed of \$25.25, together with an amount equal to accrued and unpaid dividends up to but excluding the date fixed for redemption and calculated in accordance with subsection 2(1)(c) (the <u>Optional Redemption Price</u>).
- Redemption From July 1, 2013 to June 29, 2015. From and including July 1, 2013 to and including June 29, 2015, but subject to section 7 and to applicable law, Noranda, may upon giving notice as hereinafter provided, redeem at any time the whole or from time to time any part of the then outstanding Series 3 Shares from, subject to subsection 3(4), the holders thereof as the Board of Directors may in its sole discretion determine by payment of an amount in cash for each share redeemed of \$25.00, together with an amount equal to accrued and unpaid dividends up to but excluding the date fixed for redemption and calculated in accordance with subsection 2(1)(d) (the <u>Basic Redemption Price</u>).
- (3) **Mandatory Redemption on the Final Redemption Date.** Subject to section 7 and to applicable law and to any notice of conversion given by Noranda pursuant to section 7, and notwithstanding any other redemptions of Series 1 Shares made by Noranda pursuant to any other provision hereof, Noranda shall, on the Final Redemption Date, redeem the then outstanding Series 1 Shares by payment of an amount in cash for each share redeemed equal to the Basic Redemption Price. If, as a result of applicable law or any provision herein, Noranda fails to redeem all of the Series 3 Shares on the Final Redemption Date, Noranda shall redeem as soon thereafter as possible the remaining Series 3 Shares, as nearly as practicable on a pro rata basis with any then outstanding Series 1 Shares and Series 2 Shares.

Table of Contents 104

1N

10

- (4) **Partial Redemption.** On or before June 30, 2012, but subject to the immediately following sentences, if any of the then issued and outstanding Series 3 Shares are to be redeemed, Series 3 Shares will be redeemed as nearly as practicable on a pro rata basis with the Series 1 Shares and Series 2 Shares then outstanding. From June 30, 2012 to June 29, 2015, no Series 3 Shares may be redeemed hereunder until all the Series 2 Shares have been redeemed. Following June 30, 2015, any redemption of the Series 3 Shares hereunder shall be made as nearly as practicable on a pro rata basis with any Series 1 Shares and Series 2 Shares then outstanding.
- Notice of Redemption. In the case of any Redemption pursuant to subsections (1), (2) or (3) of this section 3, Noranda will give to each holder of Series 3 Shares to be redeemed a notice in writing of the intention of Noranda to redeem such shares (the Redemption Notice) not less than 30 days and not more than 60 days prior to the Redemption Date, but in any event, Noranda will redeem the then outstanding Shares L Shares on the Redemption Date. The Redemption Notice must set out the number of Series 3 Shares held by the person to whom it is addressed which are to be redeemed, the Redemption Price and the Redemption Date. A Redemption Notice will be validly and effectively given on the date on which it is sent and such notice will be given and sent by ordinary unregistered first class prepaid mail or by delivery addressed to each holder of Series 3 Shares to be redeemed at the last address of such holder as it appears on the register of holders, or, in the case of joint holders, to the address of that one whose name appears first in the register of holders as one of such joint holders, provided that the accidental failure or omission to give any such notices as aforesaid to one or more of such holders shall not affect the validity of the redemption as to the other holders of the Series 3 Shares to be redeemed.
- Method of Payment. On the Redemption Date, Noranda will pay or cause to be paid to or to the order of the holders of the Series 3 Shares to be redeemed, the Redemption Price thereof on presentation and surrender at the principal office of the Transfer Agent, or at any other place or places within Canada designated in the Redemption Notice, of the certificate or certificates for the Series 3 Shares so called for redemption, together with such other documents as may be reasonably required to effect the redemption of the Series 3 Shares. Payment will be made by cheque payable at par at any branch in Canada of a chartered bank or trust company. If only a part of the Series 3 Shares represented by any certificate are redeemed, a new certificate for the balance will be issued at the expense of Noranda. Subject to subsection 3(5), from and after the date specified in any Redemption Notice, the Series 3 Shares called for redemption will be deemed to be redeemed and the holders of those Series 3 Shares will cease to be entitled to dividends and will not be entitled to exercise any of the rights of shareholders in respect of those Series 3 Shares unless payment of the Redemption Price is not duly made by Noranda on presentation and surrender of the certificate or certificates representing the Series 3 Shares.
- (7) **Deposit of Redemption Price.** At any time after the Redemption Notice is given, Noranda will have the right to deposit the redemption price of any or all Series 3 Shares called for redemption with any chartered bank or trust company in Canada named in the Redemption Notice, including the Transfer Agent (other than Noranda), to the credit of a special account for the respective holders of those Series 3 Shares to be paid on surrender to Noranda or that chartered bank or trust company of the certificate or certificates representing those Series 3 Shares. Any such deposit will constitute payment and satisfaction of the redemption price of the Series 3 Shares for which the deposit is made and the rights of the holders of those shares will be limited to receiving the proportion of the redemption price so deposited applicable to those shares, without interest, on presentation and

der of the certificate or certificates representing the Series 3 Shares being redeemed. Noranda will be entitled to

1P

surrender of the certificate or certificates representing the Series 3 Shares being redeemed. Noranda will be entitled to any interest on such deposit.

- (8) **Application of Net Proceeds to Redeem Series 3 Shares.** From and including June 30, 2010, Noranda will, subject to section 7 and applicable law, apply the net proceeds it realizes from:
  - (i) any issuance for cash of Common Shares, Preferred Shares, Junior Preference Shares or any securities that, under Canadian generally accepted accounting principles, would be treated as equity on the balance sheet of Noranda (which, for greater certainty, will not include the issue of Common Shares pursuant to any share compensation arrangement, including any stock option plan, share purchase plan or dividend reinvestment plan of Noranda, or pursuant to the terms of the Preferred Shares or the 123 Shares), or
  - (ii) the sale by Noranda of any capital assets outside the ordinary course of business exceeding Cdn. \$250 million, in any one transaction or series of related transactions (and for greater certainty, the entire amount of the net proceeds shall be applied net of any taxes paid or payable);

to redeem the 123 Shares; provided that from and including June 30, 2010 until June 29, 2012, so long as any Series 1 Shares are outstanding, the net proceeds will be used first to redeem any then outstanding Series 1 Shares; further provided that from and including June 30, 2012 until June 29, 2015, so long as any Series 1 Shares or Series 2 Shares are then outstanding, the net proceeds will be applied as nearly as practicable on a pro rata basis to any then outstanding Series 1 Shares and Series 2 Shares and from and including June 30, 2015, the net proceeds will be applied as nearly as practicable on a pro rata basis to any then outstanding Series 1 Shares, Series 2 Shares and Series 3 Shares.

# 4. Retraction Right

(1) **Right to Require Retraction.** Subject to section 7, applicable law, and compliance with subsections 4(2), (3) and (4), upon the occurrence of a Change of Control Event, a holder shall have the right at any time after the occurrence of a Change of Control Event (the <u>Retraction Right</u>) to require Noranda to redeem all of the holder s then outstanding Series 3 Shares.

## (2) Retraction Procedure.

(a) In order to exercise its Retraction Right, a holder must give Noranda notice of such exercise, and on or before the Change of Control Retraction Date, tender to Noranda, at its registered office or at any place at which the Series 3 Shares may be transferred, the certificate or certificates representing the Series 3 Shares which the holder wishes Noranda to redeem. Such tender shall be irrevocable unless payment of the Retraction Price shall not be duly made by Noranda to the holder on or before the Change of Control Retraction Date. In the event that payment of the Retraction Price is not made by Noranda on or before the Change of Control Retraction Date, Noranda shall forthwith thereafter return the holder s deposited share certificate or certificates to the holder.

1Q

- (b) Subject to section 7 and applicable law, Noranda shall redeem Series 3 Shares duly tendered pursuant an exercise of the Retraction Right as follows:
  - (i) upon the occurrence of an event contemplated under paragraph (i) of the definition of a Change of Control Event , Noranda shall redeem (x) on the 30 day following the occurrence of such event, all Series 3 Shares that have been tendered to Noranda on or before the third Business Day immediately prior to such 30th day and (y) thereafter, from time to time, within three Business Days following the exercise of the Retraction Right, all Series 3 Shares that have been tendered to Noranda;
  - (ii) upon the occurrence of an event contemplated under paragraph (ii) of the definition of a Change of Control Event , Noranda shall redeem, (x) on the date upon which Common Shares are first taken up under the Take-over Bid, all Series 3 Shares that have been tendered to Noranda on or before the third Business Day immediately prior to the date on which Common Shares are first taken up and (y) thereafter, from time to time, on the third Business Day following the exercise of the Retraction Right, all Series 3 Shares that have been tendered to Noranda; and
  - (iii) upon the occurrence of an event contemplated under paragraph (iii) of the definition of a Change of Control Event , Noranda shall redeem, (x) on the date that the Control Transaction is completed, all Series 3 Shares that have been tendered to Noranda on or before the third Business Day immediately prior to the date that the Control Transaction is completed and (y) thereafter, from time to time, within three Business Days following the exercise of the Retraction Right, all Series 3 Shares that have been tendered to Noranda:

in each case, by payment of an amount in cash for each Series 3 Share redeemed equal to (i) if the redemption occurs on or before June 30, 2013, the Optional Redemption Price, and (ii) if the redemption occurs after June 30, 2013, the Basic Redemption Price (the <u>Retraction Price</u>). For purposes of determining the Retraction Price, the date fixed for redemption shall be the date upon which Common Shares are first taken up under a Take-over Bid or the date that a Control Transaction is contemplated, as applicable, and, thereafter, the third Business Day immediately following the date on which Series 3 Shares are tendered by a holder to Noranda pursuant to subsection 4(2)(a).

- (c) The provisions of section 3 shall apply to any redemption pursuant to this section 4 except to the extent that such provisions are inconsistent with the provisions of this section 4.
- (3) **Exception.** Notwithstanding anything contained in this section 4, Noranda shall not redeem any Series 3 Shares tendered to Noranda pursuant to the exercise of the Retraction Right upon the occurrence of an event contemplated under paragraph (ii) or (iii) of the definition of Change of Control Event if the applicable Take-over Bid or Control Transaction made or announced is withdrawn or terminated or expires without any person becoming an Acquiring Person and, upon

1R

any such withdrawal, termination or expiry, any notice delivered pursuant to subsection 4(2)(a) shall be null and void and of no further force or effect.

(4) **Notice.** Noranda shall give written notice of the occurrence of any Change of Control Event to all holders within five Business Days of the date on which Noranda first becomes aware of such occurrence.

# 5. Conversion into Common Shares at the Option of Noranda on the Final Redemption Date.

- (1) **Conversion Right.** Subject to section 7, applicable law and any requirement to obtain regulatory relief, Noranda, at its option (the <u>Conversion Right</u>), may, upon giving written notice (the <u>Conversion Notice</u>) at least 30 days but no more than 60 days prior to the Final Redemption Date, convert all or any part of the then outstanding Series 1 Shares into that number of freely-tradeable, fully paid and non-assessable Common Shares per Series 1 Share determined by dividing the aggregate of \$25.00, plus an amount equal to accrued and unpaid dividends up to but excluding the Final Redemption Date and calculated in accordance with subsection 2(1)(c) (the whole constituting the <u>Cash Conversion Price</u>), by the greater of \$2.00 (as adjusted from time to time by the Board of Directors acting in good faith to give effect to any split, division, consolidation, combination or similar change in the capital of Noranda that affects the number of Common Shares then outstanding) and 90% of the U.S. Dollar Equivalent of the Current Market Price as of the Final Redemption Date.
- (2) **Conversion Notice.** The Conversion Notice must set out the number of Series 1 Shares held by the person to whom it is addressed which are to be converted. The Conversion Notice also must advise the holder that the Common Shares will be registered in the name of the holder of Series 1 Shares to be converted unless the Transfer Agent receives from the holder, at least 10 Business Days before the Final Redemption Date, at any principal office of the Transfer Agent, written notice in form and substance satisfactory to the Transfer Agent directing Noranda to arrange for the registration of such Common Shares in some other name or names (the <u>Transferee</u> or the <u>Transferee</u> ) and stating the name or names (with addresses), accompanied by payment to the Transfer Agent of any transfer tax which may be payable by reason of the transfer and a written declaration, if required by any applicable law or by Noranda, as to the residence of the Transferee(s) and such other matters as may be required by law or requested by Noranda in order to determine the entitlement of the Transferee(s) to such Common Shares, in which case such Common Shares will be registered in the name or names so directed in the written notice. Such conversion will be deemed to have been made at 5:00 p.m. (Toronto time) on the Final Redemption Date, so that the rights of the holder of such Series 1 Shares as the holder thereof will cease at such time and the person or persons entitled to receive Common Shares upon such conversion will be treated for all purposes as having become the holder or holders of record of such Common Shares at such time.
- (3) **Pro Rata Conversion.** If less than all of the then outstanding Series 1 Shares are at any time to be converted at the option of Noranda, then: (i) the particular Series 1 Shares to be so converted will be selected on a pro rata basis; and (ii) any Series 1 Shares not converted pursuant to this section 5 will be redeemed pursuant to subsection 3(3).

- (4) **Prohibition on Conversion.** Noranda cannot exercise the Conversion Right hereunder if: (i) subject to Section 5(7), a Change of Control Event has occurred; (ii) an Extraordinary Dividend has been paid; or (iii) any one or more of the following events has occurred:
  - (a) the issuance or delivery of such Common Shares is prohibited pursuant to any agreement or arrangement entered into by Noranda to assure Noranda s solvency or continued operation;
  - (b) the issuance or delivery of such Common Shares is prohibited by law or by any regulatory or other authority having jurisdiction over Noranda and acting in conformity with law;
  - (c) for any reason beyond its control, Noranda is unable to issue Common Shares or is unable to deliver such Common Shares; or
- (d) such Common Shares are not listed in each exchange on which the Common Shares are then listed for trading.

  (5) **Fractional Shares.** Where the aggregate number of Common Shares to be delivered to a holder of Series 3

  Shares pursuant to Noranda s exercise of the Conversion Right includes a fraction of a Common Share, Noranda, in lieu of delivering such fractional share, will pay by cheque an amount equal to the balance of the Cash Conversion Price not otherwise satisfied by the delivery of the Common Shares.

### (6) Delivery of Shares and Share Certificates.

- (a) On any conversion of Series 3 Shares pursuant to the exercise of the Conversion Right, the share certificates for Common Shares resulting therefrom will be delivered in the name of the holder of the Series 3 Shares converted or, if applicable, in the name of the Transferee; provided that such holder of Series 3 Shares will (i) pay any applicable security transfer taxes including, without limitation, any documentary, stamp, transfer or other taxes that may be payable in respect of any transfer involved in the issuance or delivery of such shares to a person other than such holder of Series 3 Shares or (ii) will have evidenced to the satisfaction of Noranda that such taxes, if any, have been paid.
- (b) Subject to these Series 3 Provisions upon the exercise of the Conversion Right any holder of Series 3 Shares to be converted (or, if applicable, any Transferee) will be deemed to have become a holder of Common Shares of record for all purposes on the Final Redemption Date, notwithstanding any delay in the delivery of certificates representing the Common Shares for which such Series 3 Shares have been converted.
- (c) Upon exercise of the Conversion Right, Noranda is not required to issue Common Shares to any person whose address is in, or whom Noranda or its transfer agent has reason to believe is a resident of, any jurisdiction outside of Canada, to the extent that such issue would require compliance by Noranda with the securities or other laws of such jurisdiction.

Table of Contents 109

1**S** 

#### **Table of Contents**

1T

(7) **Exception.** Notwithstanding section 5(4)(i), Noranda shall be entitled to exercise its Conversion Right hereunder in accordance with its terms if the applicable Change of Control Event prohibiting such conversion is the occurrence of (ii) or (iii) under the definition of Change of Control Event and the applicable Take-over Bid or Control Transaction made or announced is withdrawn or terminated or expires without any person becoming an Acquiring Person, upon any such withdrawal, termination or expiry.

#### 6. Purchase for Cancellation

Subject to section 7 and to applicable law, Noranda may, at any time, if obtainable, purchase for cancellation, by private agreement or otherwise, at the lowest price or prices at which, in the opinion of the Board of Directors, such shares are obtainable, all or from time to time any part of the then outstanding Series 3 Shares.

### 7. Restrictions on Dividends and Retirement of Shares

- (1) **General Restrictions**. As long as any Series 3 Shares are outstanding, Noranda will not at any time, without the approval of the holders of the Series 3 Shares given as provided in section 11:
  - (a) declare, pay or set apart for payment any dividends (other than stock dividends payable in shares of Noranda ranking junior to the Series 3 Shares) on shares of Noranda ranking junior to the Series 3 Shares;
  - (b) except out of the net cash proceeds of a substantially concurrent issue of shares of Noranda ranking junior to the Series 3 Shares, redeem or call for redemption, purchase or otherwise pay off, retire or make any return of capital in respect of any shares of Noranda ranking as to capital junior to the Series 3 Shares;
  - (c) redeem or call for redemption, purchase or otherwise retire for value or make any return of capital in respect of less than all of the Series 3 Shares then outstanding;
  - (d) except pursuant to any purchase obligation, sinking fund, retraction privilege or mandatory redemption provisions attaching thereto, redeem or call for redemption, purchase or otherwise pay off, retire or make any return of capital in respect of any Preferred Shares, ranking on a parity with the Series 3 Shares; or
- (e) issue any additional Series 3 Shares or any shares ranking senior to or on a parity with the Series 3 Shares, unless, in each such case, all accrued dividends up to and including those payable on the Dividend Payment Date for the last completed period for which dividends are payable on the Series 3 Shares in respect of which the rights of holders have not been extinguished, and all dividends then accrued on all other shares ranking senior to or on a parity with the Series 3 Shares with respect to the payment of dividends up to the immediately preceding respective date or dates for payment in respect of which the rights of holders of those shares have not been extinguished, have been declared, paid or set apart for payment.

1U

(2) **Extraordinary Dividends.** As long as 123 Shares having an aggregate issue price of at least \$300 million remain outstanding, Noranda will not at any time, without the approval of the holders of the Series 3 Shares given as provided in section 10 declare, pay or set apart for payment any Extraordinary Dividends.

### 8. Voting Rights and Election of Directors

- (1) **Election of Directors.** The holders of Series 3 Shares together with the holders of the Series 1 Shares and the holders of the Series 2 Shares will have at all times the right, voting collectively, to elect two directors to the Board of Directors at each meeting of shareholders of Noranda at which directors are to be elected. Forthwith upon the occurrence of (i) a Board Event, the holders of the Series 3 Shares together with the holders of the Series 1 Shares and the holders of the Series 2 Shares will, but only for so long as a Board Event is continuing, be entitled, voting collectively, to elect three additional directors (for a total of five). Upon the occurrence of a Board Event, the Board of Directors will, at its election and subject to applicable law, either (i) appoint such additional number of directors to which the holders of the Series 3 Shares (together with the holders of the Series 1 Shares and holders of the Series 2 Shares) are entitled to elect as a result of such occurrence and as may be proposed by the Series 123 Directors, or (ii) forthwith call a special meeting of the applicable shareholders of Noranda for the purpose of electing such additional number of directors and, if necessary, increasing the maximum number of directors that may be elected. The holders of Series 3 Shares are entitled to receive notice of and to attend all annual and special meetings of the Shareholders of Noranda called for the purpose of electing directors as provided in this subsection 8(1) and, for such purpose only, to one vote in respect of each Series 3 Share held at all such meetings.
- (2) **Vacancies.** If a Series 123 Director ceases to be a director for any reason other than as a result of a Board Event ceasing to continue, the Board of Directors will fill the vacancy thereby created by appointing as soon as reasonably possible that individual who is nominated by the remaining Series 123 Directors, provided that if there are no remaining 123 Directors then in office, the Board of Directors will select the individuals to be appointed.
- (3) **Meeting of Shareholders.** Except as otherwise provided by law or as specifically provided herein and except for meetings of the holders of Junior Preference Shares as a class and meetings of all holders of Series 3 Shares as a series, the holders of the Series 3 Shares will not be entitled as such to receive notice of, or to attend, or to vote at, any meeting of shareholders of Noranda.
- (4) Meeting of Holders of Series 3 Shares as a Series or Junior Preference Shares as a Class. At any meeting of holders of Series 3 Shares as a series or together with one or more other series or Junior Preference Shares as a class, each such holder will be entitled to one vote in respect of each Series 3 Shares held.

## 9. <u>Issue of Additional Common Shares</u>

For greater certainty, nothing in these Series 3 Provisions will affect or restrict the right of Noranda to increase the number of the Common Shares or to issue additional Common Shares from time to time.

1V

## 10. Approval of Holders of Series 3 Shares

In addition to any other approvals required by law, the approval of all amendments to the rights, privileges, restrictions and conditions attaching to the Series 3 Shares as a series and any other approval to be given by the holders of Series 3 Shares as a series will be deemed to have been sufficiently given if given by a resolution carried by the affirmative vote of not less than 66²/3% of the votes cast at a meeting of the holders of Series 3 Shares duly called and held on not less than 21 days notice at which the holders of at least a majority of the outstanding Series 3 Shares are present or are represented by proxy, or if no quorum is present at such meeting, at an adjourned meeting at which the holders of the Series 3 Shares then present would form the necessary quorum. If at the meeting the holders of a majority of the outstanding Series 3 Shares are not present or represented by proxy within 30 minutes after the time appointed for the meeting, the meeting will be adjourned to such date, not less than 15 days afterwards, and to such time and place as the chairman may designate, and not less than 10 days written notice will be given of the adjourned meeting. Subject to the foregoing, the formalities to be observed in respect of the giving of notice of any meeting or adjourned meeting and the conduct of any such meeting will be those prescribed from time to time in the by-laws of Noranda with respect to meetings of shareholders.

## 11. Capital Reorganizations and Amalgamations of Noranda

In the event of a capital reorganization, merger or amalgamation of Noranda or comparable transaction affecting the Series 3 Shares, holders of the Series 3 Shares will be entitled to receive pursuant to the exercise of the Conversion Right, and each such holder will accept, the number of shares or other securities or consideration of Noranda or a corporation resulting, surviving or continuing from the capital reorganization, merger, amalgamation or comparable transaction that such holder would have received had such holder been a holder of Common Shares immediately prior to the record date of the capital reorganization, merger, amalgamation or comparable transaction.

#### 12. Notices

- (1) **Notice to Noranda.** Subject to applicable law, any notice, request or other communication to be given to Noranda by a holder of Series 3 Shares must be in writing and will be valid and effective if given by mail (postage prepaid) or by electronic communication or by delivery to the registered office of Noranda and addressed to the attention of the Secretary. Any such notice, request or other communication, if given by mail, electronic communication or delivery, will be deemed to have been given and received only on actual receipt by Noranda.
- (2) **Presentation and Surrender of Certificates.** Any presentation and surrender by a holder of Series 3 Shares to Noranda or the Transfer Agent of certificates representing Series 3 Shares in connection with the redemption of Series 3 Shares must be made by registered mail (postage prepaid) or by delivery to the registered office of Noranda or to such office of the Transfer Agent as may be specified by Noranda, in each case addressed to the attention of the Secretary of Noranda. Any such presentation and surrender of certificates will be deemed to have been made and to be effective only on actual receipt by Noranda or the Transfer Agent, as the case may be. Any such presentation and surrender of certificates made by registered mail will be at the sole risk of the holder mailing the same.

1W

(3) Notice to Holders of Series 3 Shares. Subject to applicable law, any notice, including but not limited to the Redemption Notice and the Conversion Notice, request or other communication to be given to a holder of Series 3 Shares by or on behalf of Noranda must be in writing and will be valid and effective if given by ordinary unregistered first class mail (postage prepaid) or by electronic communication or by delivery to the address of the holder recorded in the securities register of Noranda or, in the case of joint holders, to the address of that one whose name appears first in the register of holders as one of such joint holders. Any such notice, request or other communication, if given by mail, will be deemed to have been given and received on the fifth Business Day following the date of mailing and, if given by electronic communication or by delivery, will be deemed to have been given and received on the date of electronic communication or delivery. Accidental failure or omission to give any notice, request or other communication to one or more holders of Series 3 Shares, or any defect in such notice, will not invalidate or otherwise alter or affect any action or proceeding to be taken by Noranda pursuant to that notice, request or other communication.

#### 13. Tax Election

Noranda will elect, in the manner and within the time provided under Section 191.2 in Part VI.1 of the Tax Act or any successor or replacement provision of similar effect, and take all other necessary action under the Tax Act, to pay or cause payment of the tax under Part VI.1 of the Tax Act at a rate such that the corporate holders of the Series 3 Shares will not be required to pay tax under Part IV.1 of the Tax Act on dividends received on such shares.

#### 14. Return of Unclaimed Funds to Noranda

Subject to applicable laws, Noranda will have the right, with respect to any funds deposited by Noranda to any chartered bank or trust company in respect of amounts due to holders of Series 3 Shares, on or after the second anniversary date of the deposit of such funds to any chartered bank or trust company, to require that such chartered bank or trust company return to Noranda any funds which remain unclaimed by holders of the Series 3 Shares. If at any time following the return of the unclaimed funds to Noranda, either Noranda or the chartered bank or trust company receives a request from a holder of Series 3 Shares for the holder s unclaimed entitlement to any amount due to the holder in respect of the holder s Series 3 Shares, Noranda will promptly arrange for the payment of such amount to the holder.

#### 15. Wire or Electronic Transfer of Funds

Notwithstanding any other right, privilege, restriction or condition attaching to the Series 3 Shares, Noranda may, at its option, make any payment due to a holder of Series 3 Shares hereunder by way of a wire or electronic transfer of funds to each registered holder of Series 3 Shares. In the event that a payment is made by way of a wire or electronic transfer of funds, Noranda will be responsible for any applicable charges or fees relating to the making of such transfer. As soon as practicable following the determination by Noranda that a payment is to be made by way of a wire or electronic transfer of funds, Noranda will notify each registered holder of Series 3 Shares at the address of such holder as it appears on the register of holders. Such notice by Noranda will request that each registered holder of Series 3 Shares provide the particulars of an account of such holder with a chartered bank or trust company in Canada to which the wire or electronic transfer of funds

1X

will be directed. In the event that Noranda does not receive account particulars from a registered holder of Series 3 Shares prior to the date such payment is to be made, Noranda will deposit the funds otherwise payable to such holder in a special account or accounts in trust for each respective holder. The making of a payment by way of a wire or electronic transfer of funds or, in the case where a registered holder of the Series 3 Shares has not provided Noranda with account particulars for a wire or electronic transfer of funds, the deposit by Noranda of the funds otherwise payable to such holder in a special account or accounts in trust for such holder, will be deemed to constitute payment by Noranda on the date thereof and will satisfy and discharge all liabilities of Noranda for such payment to the extent of the amount represented by such transfer.

### 16. Additional Series

As long as any of the 123 Shares are outstanding, Noranda may not issue any additional Preferred Shares, other than Cumulative Redeemable Preferred Shares, Series F issuable upon the conversion of Cumulative Redeemable Preferred Shares, Series G and Cumulative Redeemable Preferred Shares, Series G issuable upon the conversion of Cumulative Redeemable Preferred Shares, Series F, without the approval of the holders of the 123 Shares expressed by resolution of the holders of all such shares voting together. For this purpose, the approval to be given will be deemed to have been sufficiently given if given by a resolution carried by the affirmative vote of not less than 66²/3% of the votes cast at a meeting of the holders of 123 Shares duly called and held on not less than 21 days notice at which the holders of at least a majority of the outstanding 123 Shares are present or are represented by proxy, or if no quorum is present at such meeting, at an adjourned meeting at which the holders of the 123 Shares then present would form the necessary quorum. If at the meeting the holders of a majority of the outstanding 123 Shares are not present or represented by proxy within 30 minutes after the time appointed for the meeting, the meeting will be adjourned to such date, not less than 15 days afterwards, and to such time and place as the chairman may designate, and not less than 10 days written notice will be given of the adjourned meeting. Subject to the foregoing, the formalities to be observed in respect of the giving of notice of any meeting or adjourned meeting and the conduct of any such meeting will be those prescribed from time to time in the by-laws of Noranda with respect to meetings of shareholders.

## EXHIBIT B

Conditions of the Issuer Bid

#### **Table of Contents**

Notwithstanding any other provision of the Issuer Bid, the Corporation shall not be required to accept for exchange or exchange any Shares and may terminate, amend or cancel the Issuer Bid or may postpone the exchange of any Shares tendered if, at any time on or after March 8, 2005 and at or before the exchange of any such Shares for Preferred Shares, any of the following events shall have occurred:

- (a) there shall have been threatened, instituted or pending any action or proceeding by any government or governmental authority or regulatory or administrative agency in any jurisdiction, or by any other person in any jurisdiction, before any court or governmental authority or regulatory or administrative agency in any jurisdiction, (i) challenging or seeking to cease trade, or make illegal, or delay or otherwise directly or indirectly restrain or prohibit the making of the Offer, the acceptance for exchange of some or all of the Shares by the Corporation, the issuance of the Preferred Shares in exchange therefor or otherwise directly or indirectly relating in any manner to or affecting the Issuer Bid, or (ii) that otherwise, in the sole judgment of the Corporation, acting reasonably, has or may have a material adverse effect on value of, or trading in, the Shares or on the business, properties, condition (financial or otherwise), income, assets, liabilities, operations, results of operations or prospects of the Corporation and its subsidiaries, taken as a whole (a Material Adverse Effect ), or has or may materially impair the contemplated benefits of the Issuer Bid to the Corporation and its subsidiaries taken as a whole; or
- (b) there shall (i) have been any approval withheld or any statute, rule, regulation, stay, decree, judgment or order or injunction proposed (and likely to be enacted), sought, enacted, enforced, promulgated, amended, issued or deemed applicable to the Issuer Bid, the Corporation or any of its subsidiaries by any court, government, governmental authority or regulatory or administrative agency in any jurisdiction, or (ii) be any other prohibition at law, that, in the sole judgment of the Corporation, acting reasonably, might result directly or indirectly in any of the consequences referred to in clauses (i) or (ii) of paragraph (a) above or would or might prohibit, prevent, restrict or delay consummation of the Issuer Bid; or
- (c) there shall have occurred: (i) any general suspension of trading in, or limitation on prices for, securities on any securities exchange or in the over-the-counter market in Canada or the United States; (ii) the declaration of a banking moratorium or any suspension of payments in respect of banks in Canada or in the United States; (iii) the commencement of a war, armed hostilities or other international or national calamity, or the escalation of any existing war, armed hostilities or other international or national calamity, directly or indirectly involving Canada or the United States, which, in the sole judgment of the Corporation, acting reasonably, materially adversely affects or will materially adversely affect the financial markets or the business, operations or affairs of the Corporation and its subsidiaries, taken as a whole; (iv) any change in the general political, market, economic or financial conditions that has or may have a Material Adverse Effect other than a change relating to these conditions generally and not specifically to the Corporation and its subsidiaries; or
- (d) any change or changes shall have occurred (or any development shall have occurred involving any prospective change or changes) in the business, assets, liabilities, condition (financial or otherwise), operations or results of operations of the Corporation or its subsidiaries that, in the sole judgment of the Corporation, acting reasonably, has or may have a

#### **Table of Contents**

Material Adverse Effect, provided that (i) the outcome to the Falconbridge Offer shall not be considered such a change, and (ii) such change or changes do not generally affect other senior mining companies listed on the Toronto Stock Exchange; or

- (e) the Corporation shall not have received all necessary regulatory approvals from government bodies, regulatory agencies or any stock exchange; or
- (f) any change shall have occurred or been proposed to the Income Tax Act (Canada) or to the published administrative policies of the Canada Customs and Revenue Agency that, in the sole judgment of the Corporation, acting reasonably, is detrimental to the Corporation or adversely affects the tax consequences of the Issuer Bid for holders of Shares, generally; or
- (g) at the expiry time, fewer than 63,377,140 Shares have been tendered (or deemed to have been tendered) and not withdrawn by Brascan and its affiliates under the Offer.

2

#### **SCHEDULE B**

## **Terms of the Insider Bid**

1. <u>General Terms</u>. The Insider Bid offer (the **Falconbridge Offer**) shall be made by a circular bid to all shareholders and prepared in compliance with the OSA and other applicable provincial securities laws. The Falconbridge Offer shall be open for an initial period of at least thirty-five (35) days (calculated in accordance with the OSA) or such longer period as may be required to satisfy all of the conditions set forth in paragraph 3 below and the terms of the support agreement (the **Support Agreement**) between the Corporation and Falconbridge Limited ( **Falconbridge**) dated as of the date hereof.

Upon the terms and subject to the conditions of the Falconbridge Offer, the Purchaser will exchange all outstanding common shares of Falconbridge ( **Falconbridge Shares** ), not owned by the Corporation, deposited and not withdrawn under the Falconbridge Offer as required by applicable securities laws and in accordance with the terms of the Support Agreement.

Unless the Falconbridge Offer is withdrawn, Falconbridge Shares may be deposited pursuant to the Falconbridge Offer at any time prior to the close of business on the date of the first exchange of Falconbridge Shares. Any Falconbridge Shares deposited pursuant to the Falconbridge Offer may be withdrawn until exchanged.

The Purchaser shall have the right to vary the terms of the Falconbridge Offer to:

- (a) increase the consideration offered for the Falconbridge Shares;
- (b) extend the expiry date of the Falconbridge Offer on two separate occasions; or
- (c) waive conditions to the Falconbridge Offer (except that set forth as the first condition in Exhibit C).

and shall otherwise not vary the Falconbridge Offer without the consent of Falconbridge.

- 2. **Price of the Offer**. The Falconbridge Offer shall be made on the basis of 1.77 common shares of the Corporation (but without issuing fractional shares; entitlements to fractional shares will be paid in cash) per Falconbridge Share.
- 3. Conditions of the Offer. The Falconbridge Offer shall be subject to the conditions set out in Exhibit C.

## **EXHIBIT C**

Conditions of the Insider Bid

### **Table of Contents**

Notwithstanding any other provision of the Falconbridge Offer, the Corporation will not be required to accept for exchange or exchange any Falconbridge Shares tendered and may terminate, amend or cancel the Falconbridge Offer or may postpone the exchange of Falconbridge Shares tendered, if at any time on or after March 8, 2005 and at or before the exchange of any such Falconbridge Shares for Shares, any of the following events shall have occurred:

- (a) at the expiry time, the number of Falconbridge Shares tendered (or deemed tendered) and not withdrawn under the Falconbridge Offer is not sufficient to enable the Corporation to complete a second stage business combination in accordance with applicable securities laws;
- (b) there shall have been threatened, instituted or pending any action or proceeding by any government or governmental authority or regulatory or administrative agency in any jurisdiction, or by any other person in any jurisdiction, before any court or governmental authority or regulatory or administrative agency in any jurisdiction, (i) challenging or seeking to cease trade, or make illegal, or delay or otherwise directly or indirectly restrain or prohibit the making of the Falconbridge Offer, the acceptance for exchange of some or all of the Falconbridge Shares by the Corporation, the issuance of the Shares in exchange therefor or otherwise directly or indirectly relating in any manner to or affecting the Falconbridge Offer, or (ii) that otherwise, in the sole judgment of the Corporation, acting reasonably, has or may have a material adverse effect on the trading in, or the value of, the Falconbridge Shares or on the business, properties, condition (financial or otherwise), income, assets, liabilities, operations, results of operations or prospects of Falconbridge and its subsidiaries, taken as a whole, or has or may materially impair the contemplated benefits of the Falconbridge Offer to the Corporation and its subsidiaries taken as a whole (a Material Adverse Effect ); or
- (c) there shall (i) have been any approval withheld or any statute, rule, regulation, stay, decree, judgment or order or injunction proposed (and likely to be enacted), sought, enacted, enforced, promulgated, amended, issued or deemed applicable to the Falconbridge Offer, the Corporation or any of its subsidiaries by any court, government, governmental authority or regulatory or administrative agency in any jurisdiction, or (ii) be any other prohibition at law, that, in the sole judgment of the Corporation, acting reasonably, might result directly or indirectly in any of the consequences referred to in clauses (i) or (ii) of paragraph (a) above or would or might prohibit, prevent, restrict or delay consummation of the Falconbridge Offer; or
- (d) there shall have occurred: (i) any general suspension of trading in, or limitation on prices for, securities on any securities exchange or in the over-the-counter market in Canada or the United States; (ii) the declaration of a banking moratorium or any suspension of payments in respect of banks in Canada or in the United States; (iii) the commencement of a war, armed hostilities or other international or national calamity, or the escalation of any existing war, armed hostilities or other international calamity, directly or indirectly involving Canada or the United States, that, in the sole judgment of the Corporation, acting reasonably, materially adversely affects or will materially adversely affect the financial markets or the business, operations or affairs of Falconbridge and its subsidiaries, taken as a whole; (iv) any significant decrease in the market price of the Shares since the close of business on the date preceding the date of the Falconbridge Offer if such decrease has disproportionately affected Falconbridge as compared to other senior publicly traded mining companies listed on the Toronto Stock Exchange; (v) any change in the general political, market, economic or financial conditions that, in the sole judgment of the Corporation, acting reasonably, has or may have a Material Adverse Effect; or

#### **Table of Contents**

- (e) any change or changes shall have occurred (or any development shall have occurred involving any prospective change or changes) in the business, assets, liabilities, condition (financial or otherwise), operations, or results of operations of Falconbridge or its subsidiaries that, in the sole judgment of the Corporation, acting reasonably, has or may have a Material Adverse Effect, provided that such change or changes do not generally affect other senior mining companies listed on the Toronto Stock Exchange; or
  - (f) the Corporation has not taken up any Shares under the Issuer Bid; or
- (g) the Corporation shall not have received all necessary regulatory approvals from government bodies, regulatory agencies or the stock exchanges and exemption orders for the Falconbridge Offer under applicable securities legislation in Canada; or
- (h) any change shall have occurred or been proposed to the Income Tax Act (Canada) or to the published administrative policies of the Canada Customs and Revenue Agency that, in the sole judgment of the Corporation, acting reasonably, is detrimental to the Corporation or adversely affects the tax consequence of the Falconbridge Offer for shareholders, generally; or
- (i) the Corporation shall have determined, in its sole judgment, acting reasonably, that the Falconbridge Offer, if completed, has or may have a Material Adverse Effect; or
- (j) the Corporation will have become aware of any untrue statement of a material fact, or an omission to state a material fact that is required to be stated or that is necessary to make a statement not misleading in the light of the circumstances in which it was made and at the date it was made (after giving effect to all subsequent filings in relation to all matters covered in earlier filings), in any document filed by or on behalf of Falconbridge after March 8, 2005 with any regulatory authority in Canada or elsewhere; or
  - (k) the Corporation will have become entitled to terminate the Support Agreement in accordance with its terms.

2

# EXHIBIT 6 REGISTRATION RIGHTS AGREEMENT

This Registration Rights Agreement (the Agreement ) dated as of March 8, 2005 between **Noranda Inc.**, a corporation incorporated under the laws of the Province of Ontario, Canada (the **Corporation**), **Brascan Corporation**, a corporation subsisting under the laws of the Province of Ontario, Canada ( **Brascan** ) and **Brascade Corporation**, a corporation subsisting under the laws of Canada ( **Brascade** ). Brascan and Brascade are together referred to as the **Holders**.

#### **RECITALS**

WHEREAS the Holders currently hold 122,597,952 common shares of the Corporation and adjustable rate convertible debentures of the Corporation convertible into 2,727,323 common shares of the Corporation;

AND WHEREAS the Corporation is proposing to undertake a reorganization of its share capital and in consideration of the Holders agreement to support that reorganization, the Corporation has agreed to grant the Holders the rights provided by this Agreement.

NOW, THEREFORE, in consideration of the premises and the mutual covenants herein contained and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

# ARTICLE 1 DEFINITIONS.

- **1.1** For the purposes of this Agreement:
- 1.1.1 **Canadian Prospectus** shall mean a prospectus (including a short form prospectus) prepared in accordance with applicable Canadian Securities Laws for the purposes of qualifying securities for distribution or distribution to the public, as the case may be, in any province or territory of Canada.
- 1.1.2 **Canadian Securities Law** shall mean statutes and regulations applicable to the trading of securities in any province or territory of Canada including applicable rules, policy statements and blanket rulings and orders promulgated by Canadian securities regulatory authorities.
- 1.1.3 **Closing Date** means the date of closing of the reorganization of the Corporation s share capital contemplated on the date hereof.
- 1.1.4 **Commission** means the United States Securities and Exchange Commission or any other United States federal agency at the time administering the Securities Act.
- 1.1.5 **Common Shares** means common shares in the capital of the Corporation.

#### **Table of Contents**

- 1.1.6 **Current Stock Market Value** means, at any date set forth herein, the product of (i) the average of the closing sale prices on the Toronto Stock Exchange (or any Canadian securities exchange which is the successor to the Toronto Stock Exchange) of one Common Share over the ten trading days ending on the trading day immediately prior to such date, and (ii) the number of Registrable Shares (as defined below) held by a Holder or Holders to be registered on behalf of such Holder or Holders, as the case may be.
- 1.1.7 **Demand Registrations** has the meaning specified in section 2.1.
- 1.1.8 **Exchange Act** means the United States Securities Exchange Act of 1934, as amended, or any successor United States federal statute, and the rules and regulations of the Commission thereunder, all as the same shall be in effect from time to time.
- 1.1.9 **Holder Approval** means the approval of holders of at least 51% of the Registrable Shares.
- 1.1.10 **Person** means a natural person, a partnership, a company, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization or a governmental entity or any department, agency or political subdivision thereof.
- 1.1.11 **Piggyback Registration** has the meaning specified in section 3.1.
- 1.1.12 The terms **register**, **registered** and **registration** refer to a registration effected by preparing and filing a registration statement in compliance with the Securities Act and the effectiveness of such registration statement. In addition, unless inconsistent with the context: (i) the term **registration** and any references to the act of registering include the qualification under Canadian Securities Laws of a Canadian Prospectus in respect of a distribution or distribution to the public, as the case may be, of securities; (ii) the term **registered** as applied to any securities includes a distribution or distribution to the public, as the case may be, of securities so qualified; (iii) the terms **registration statement** includes a Canadian Prospectus; (iv) any references to a registration statement having become effective, or similar references, shall include a Canadian Prospectus for which a final MRRS decision document has been obtained from the relevant Canadian Principal Regulator under National Policy 43-201 Mutual Reliance Review System for Prospectuses and Annual Information Forms; and (v) the provisions of this Agreement shall be applied, *mutatis mutandis*, to any proposed distribution of securities hereunder in any province or territory of Canada or to which the prospectus requirements under any of the Canadian Securities Laws shall otherwise apply.
- 1.1.13 **Registrable Shares** means at any time any Common Shares held by the Holders, or issuable upon conversion or exchange of any securities (including any adjustable rate convertible debentures of the Corporation) held by the Holders; <u>provided</u> that Registrable Shares shall not include any shares (i) the sale of which has been registered and which shares have been sold pursuant to such registration either in Canada or the United States or (ii) which have been sold to the public pursuant to Rule 144 or equivalent provisions under Canadian Securities Laws. For all purposes of this

2

#### **Table of Contents**

Agreement, a Person will be deemed to be a Holder of Registrable Shares whenever such Person has the then existing right to acquire such Registrable Shares (by exercise, conversion or exchange of securities or otherwise), whether or not such acquisition has actually been effected.

- 1.1.14 **Registration Expenses** has the meaning specified in section 6.1 of this Agreement.
- 1.1.15 **Rule 144** means Rule 144 of the Commission under the Securities Act or any successor provision of similar effect.
- 1.1.16 **Securities Act** means the United States Securities Act of 1933, as amended, or any successor United States federal statute, and the rules and regulations of the Commission thereunder, all as the same shall be in effect from time to time.

# ARTICLE 2 DEMAND REGISTRATIONS

- **2.1** Requests for Registration. At any time and from time to time after the Closing Date, any Holder may request registration of all or part of its Registrable Shares for sale in Canada and/or the United States (provided, in the case of a United States registration, that the Corporation is, at such time, eligible to use the United States Multijurisdictional Disclosure System), provided, however, that:
- 2.1.1 the Corporation shall not be obligated to register, pursuant to this section 2.1, Registrable Shares with a Current Stock Market Value of less than C\$200,000,000, determined as of the date of the applicable request for registration hereunder;
- 2.1.2 the Corporation shall not be obligated to register Registrable Shares pursuant to this section 2.1 on more than three occasions in the aggregate;
- 2.1.3 the Corporation shall not be obligated to register Registrable Shares pursuant to this section 2.1 if the Corporation delivers notice to the holders of Registrable Shares as soon as practicable after any request hereunder that the Corporation in good faith believes that it will file a registration statement for a public offering of Common Shares within 90 days of such request, <u>provided</u>, <u>however</u>, that the Corporation shall have the right to refuse a registration on such grounds only one time during any period of 12 consecutive months;
- 2.1.4 the Corporation shall not be required to cause a registration statement pursuant to this section 2.1 to be declared effective within a period of (i) 180 days after the effective date of any registration statement of the Corporation filed pursuant to a Demand Registration or (ii) 90 days after the effective date of any other registration statement of the Corporation filed pursuant to section 3.1;
- 2.1.5 if the Corporation in good faith believes that any registration or continued registration of Registrable Shares would require the disclosure of information that would

3

#### **Table of Contents**

be materially detrimental to the Corporation and its shareholders or the disclosure of which would materially adversely affect the Corporation s ability to consummate a material acquisition, corporate reorganization or merger or other transaction involving the Corporation (a **Valid Business Reason**):

- 2.1.5.1 the Corporation may postpone filing a registration statement relating to a Demand Registration until such Valid Business Reason no longer exists, but in no event for more than 90 days, and
- 2.1.5.2 in case a registration statement has been filed relating to a Demand Registration, if the Valid Business Reason has not resulted from actions taken by the Corporation, the Corporation may cause such registration statement to be withdrawn or may postpone amending or supplementing such registration statement until such Valid Business Reason no longer exists, but in no event for more than 60 days (such period of postponement or withdrawal under sections 2.1.5.1 and 2.1.5.2, the **Postponement Period**);

and the Corporation shall give written notice of its determination to postpone or withdraw a registration statement and of the fact that the Valid Business Reason for such postponement or withdrawal no longer exists, in each case, promptly after the occurrence thereof. If the Corporation gives any notice of postponement or withdrawal of any registration statement, the Corporation shall not, during the period of postponement or withdrawal, register or continue the registration of any Common Shares. Each Holder of Registrable Shares agrees that, upon receipt of any notice from the Corporation that the Corporation has determined to withdraw any registration statement pursuant to clause 2.1.5.2 above, such party will discontinue its disposition of Registrable Shares pursuant to such registration statement and, if so directed by the Corporation, will deliver to the Corporation (at the Corporation s expense) all copies, other than permanent file copies, then in such party s possession, of the prospectus covering such Registrable Shares that was in effect at the time of receipt of such notice. If the Corporation shall have withdrawn a registration statement filed under this section 2.1, the Corporation shall pay all expenses incurred by the Holders in connection with such registration statement and shall not be considered to have effected an effective registration for the purposes of this Agreement until the Corporation shall have filed a new registration statement covering the Registrable Shares covered by the withdrawn registration statement and such registration statement shall have been declared effective and shall not have been withdrawn. If the Corporation shall give any notice of withdrawal or postponement of a registration statement, the Corporation shall, at such time as the Valid Business Reason that caused such withdrawal or postponement no longer exists (but in no event later than 60 days after the date of the withdrawal or postponement), use its best efforts to effect the registration of the Registrable Shares covered by the withdrawn or postponed registration statement in accordance with this section 2.1 (unless the initiating Holders shall have withdrawn such request, in which case the Corporation shall not be considered to have effected an effective registration for the purposes of this Agreement). The Corporation agrees that it shall have the right to postpone or withdraw a registration statement pursuant to this section 2.1.5 only one time during any period of 12 consecutive months.

4

## **Table of Contents**

- 2.1.6 All registrations requested pursuant to this section 2.1 shall be referred to herein as **Demand Registrations**".
- 2.2 Priority on Demand Registrations. The Holders of a majority of the Registrable Shares to be sold pursuant to a Demand Registration shall, upon mutual agreement, designate a managing underwriter of recognized standing, if applicable, for such offering, which underwriter shall be acceptable to the Corporation, acting reasonably. If the managing underwriter advises the Corporation in writing that in its opinion the number of Registrable Shares (i) creates a substantial risk that the price per share in such registration will be materially and adversely affected, or (ii) exceeds the number of Registrable Shares that can be sold in such offering, then the Corporation will include in such registration, prior to the inclusion of any securities that are not Registrable Shares, the number of Registrable Shares requested to be included that, in the opinion of such underwriter, can be sold, pro rata among the respective Holders on the basis of the number of Registrable Shares owned by such Holders.

## ARTICLE 3 PIGGYBACK REGISTRATIONS

- 3.1 Right to Piggyback. At any time and from time to time after the Closing Date, whenever the Corporation proposes to register any of its Common Shares (including, without limitation, a Demand Registration) in Canada, the United States or both and the registration form to be used may be used for the registration of Registrable Shares (a Piggyback Registration"), the Corporation will give prompt written notice to all Holders of Registrable Shares of its intention to effect such a registration (which notice shall be given not less than (i) in the case of a Demand Registration, ten days after receipt by the Corporation of a request therefor pursuant to section 2.1, and (ii) in all other cases, 20 days prior to the date the registration statement is to be filed) and, subject to the terms hereof, will include, subject to sections 3.3, 3.4 and 3.5, in such registration and in such jurisdictions as it proposed to effect the registration all Registrable Shares with respect to which the Corporation has received written requests for inclusion therein within 15 days after the receipt of the Corporation s notice.
- 3.2 <u>Determination not to Register and Delays in Registration</u>. If at any time after giving written notice of its intention to register any Common Shares and prior to the effective date of the registration statement filed in connection with such registration, the Corporation shall determine for any reason not to register or to delay registration of such Common Shares, the Corporation may, at its election, give written notice of such determination to all Holders, and (i) in the case of a determination not to register, shall be relieved of its obligation to register any Registrable Shares in connection with such abandoned registration, without prejudice, however, to the rights of Holders under section 2.1 and (ii) in the case of a determination to delay such registration of Common Shares, shall be permitted to delay the registration of such Registrable Shares for the same period as the delay in registering such other Common Shares.

5

#### **Table of Contents**

- **3.3 Priority on Primary Registrations.** If a Piggyback Registration is an underwritten primary registration on behalf of the Corporation, and the managing underwriter advises the Corporation in writing that in its opinion the number of Registrable Shares and other securities requested to be included in such registration (i) creates a substantial risk that the price per share in such registration will be materially and adversely affected, or (ii) exceeds the number of Registrable Shares and other securities that can be sold in such offering, then the Corporation will include in such registration: <u>first</u>, the securities the Corporation proposes to sell; <u>second</u>, the Registrable Shares requested to be included in such registration to the extent that, in the opinion of such underwriter, they can be sold, <u>pro rata</u> among the Holders of such Registrable Shares on the basis of the number of Registrable Shares owned by such Holders and requested to be so registered, and <u>third</u>, other securities requested to be included in such registration to the extent that, in the opinion of such underwriter, they can be sold; provided, however, that in the case of any Demand Registration, the terms of section 2.2 shall supersede and control the terms of this section 3.3.
- **3.4 Priority on Secondary Registrations.** If a Piggyback Registration is not an underwritten primary registration but is an underwritten secondary registration on behalf of holders of the Corporation s Common Shares, and the managing underwriter advises the Corporation in writing that in its opinion the number of securities requested to be included in such registration (i) creates a substantial risk that the price per share in such registration will be materially and adversely affected, or (ii) exceeds the number of Registrable Shares and other securities that can be sold in such offering, then the Corporation will include in such registration only: <u>first</u>, the securities requested to be included therein by the holders requesting such registration, <u>second</u>, any Registrable Shares requested to be included in such registration to the extent that, in the opinion of such underwriter, they can be sold, <u>pro rata</u> among the Holders of such Registrable Shares on the basis of the number of Registrable Shares owned by such Holders and requested to be so registered, and <u>third</u>, other securities requested to be included in such registration to the extent that, in the opinion of such underwriter, they can be sold; <u>provided</u>, <u>however</u>, that in the case of any Demand Registration, the terms of section 2.2 shall supersede and control the terms of this section 3.4.
- 3.5 Other Registrations. If the Corporation has previously received a request for a Demand Registration pursuant to Article 2 or has previously filed a registration statement with respect to Registrable Securities pursuant to this Article 3, and if such previous request or registration has not been withdrawn or abandoned, the Corporation will not, except (i) after receiving Holder Approval or (ii) as otherwise contemplated by section 3.5, file or cause to be effected any other registration of any of its Common Shares or securities convertible or exchangeable into Common Shares (except on Form F-4, Form F-7, Form F-8, Form S-8 or any successor forms or except pursuant to a similar registration under Canadian Securities Laws), whether on its own behalf or at the request of any holder or holders of such securities, until a period of 90 days has elapsed from the effective date of such Demand Registration or previous registration.
- **3.6** Holders Withdrawal Right. Any Holder shall have the right to withdraw its request for inclusion of its Registrable Shares in any registration statement pursuant to section 3.1 by

6

#### **Table of Contents**

giving written notice to the Corporation of its request to withdraw; <u>provided</u>, <u>however</u>, that (i) such request must be made in writing prior to the execution of the underwriting agreement (or such other similar agreement) with respect to such registration and (ii) such withdrawal shall be irrevocable and, after making such withdrawal, a Holder shall no longer have any right to include Registrable Shares in the registration as to which such withdrawal was made.

## ARTICLE 4 HOLDBACK AGREEMENTS

- **4.1 Holders of Registrable Shares.** If requested by the managing underwriter of any registration effected pursuant to an underwriting agreement pursuant to section 2.1 or 3.1, each of the Holders of Registrable Shares agrees (and shall sign an agreement to such effect in the form mutually agreed with the managing underwriter, if the managing underwriter requests such agreement), not to effect any public sale or distribution of Common Shares, or any securities convertible into or exchangeable or exercisable for Common Shares, or any hedging or similar transactions, during the 15 days prior to and a period of up to 180 days following the effective date of any underwritten Demand Registration or any underwritten Piggyback Registration (except as part of such underwritten registration); provided that all directors, officers and holders of at least 5% of any class of capital stock of the Corporation agree to the same.
- 4.2 The Corporation. If requested by the managing underwriter of any registration effected pursuant to an underwriting agreement pursuant to section 2.1 or 3.1, the Corporation agrees (and shall sign an agreement to such effect in the form mutually agreed with the managing underwriter, if the managing underwriter requests such agreement), (i) not to effect any public sale or distribution of its Common Shares, or any securities convertible into or exchangeable or exercisable for Common Shares, during the 15 days prior to and a period of up to 180 days following the effective date of any underwritten Demand Registration or any underwritten Piggyback Registration (except as part of such underwritten registration), and (ii) to use all commercially reasonable efforts to cause each holder of at least 5% (on a fully-diluted basis) of its Common Shares (other than Common Shares acquired in a public trading market), or any securities convertible into or exchangeable or exercisable for Common Shares, purchased from the Corporation at any time after the date of this Agreement to agree (and to sign an agreement to such effect in the form mutually agreed with the managing underwriter, if the managing underwriter requests such agreement), not to effect any public sale or distribution of any such securities during such period (except as part of such underwritten registration).

## ARTICLE 5 REGISTRATION PROCEDURES

**5.1** Corporation s Obligations. If and whenever required to register any Registrable Shares pursuant to sections 2.1 and 3.1 of this Agreement, the Corporation will use its best efforts to effect the registration and the sale of such Registrable Shares in accordance

7

#### **Table of Contents**

with the intended method of disposition thereof, and pursuant thereto the Corporation will as expeditiously as possible:

- 5.1.1 prepare and file with the Commission and/or Canadian securities regulatory authorities within 60 days of the date of such request a registration statement with respect to such Registrable Shares and cause such registration statement to become and remain effective for such period as may be reasonably necessary to effect the sale of such securities as described in such request;
- 5.1.2 prepare and file with the Commission and/or Canadian securities regulatory authorities such amendments and supplements to such registration statement and the prospectus used in connection therewith as may be necessary to keep such registration statement effective and comply with the provisions of the Securities Act and/or Canadian Securities Laws, as applicable, with respect to the disposition of all securities covered by such registration statement during such period in accordance with the intended methods of disposition by the sellers thereof set forth in such registration statement;
- 5.1.3 furnish to each seller of Registrable Shares and the underwriters of the securities being registered such number of copies of such registration statement, each amendment and supplement thereto, the prospectus included in such registration statement (including each preliminary prospectus) and such other documents as such seller or underwriters may reasonably request in order to facilitate the disposition of the Registrable Shares owned by such seller or the sale of such securities by such underwriters;
- 5.1.4 use its commercially reasonable efforts to register or qualify such Registrable Shares under such other securities or blue sky laws of such jurisdictions as any seller or, in the case of an underwritten public offering, the managing underwriter, reasonably requests and do any and all other acts and things which may be reasonably necessary to enable such seller to consummate the disposition in such jurisdictions of the Registrable Shares owned by such seller; provided, however, that the Corporation will not be required to (i) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this subsection or (ii) consent to general service of process in any such jurisdiction;
- 5.1.5 use its commercially reasonable efforts to cause all such Registrable Shares to be listed not later than five business days after the effective date of the registration statement on each securities exchange on which similar securities issued by the Corporation are then listed;
- 5.1.6 provide a transfer agent and registrar for all such Registrable Shares not later than the effective date of such registration statement;
- 5.1.7 enter into such customary agreements (including underwriting agreements in customary form) and take all such other actions as the Holders of a majority of the Registrable Shares being sold or the underwriters, if any, reasonably request in order to expedite or facilitate the disposition of such Registrable Shares;

8

#### **Table of Contents**

- 5.1.8 make available for inspection by any seller of Registrable Shares, any underwriter participating in any disposition pursuant to such registration statement, and any attorney, accountant or other agent retained by any such seller or underwriter, all financial and other records, pertinent corporate documents and properties of the Corporation, and cause the Corporation s officers, directors, employees and independent accountants to supply all information reasonably requested by any such seller, underwriter, attorney, accountant or agent in connection with the preparation of such registration statement;
- 5.1.9 notify each seller of such Registrable Shares, promptly after it shall receive notice thereof, of the time when such registration statement has become effective or a supplement to any prospectus forming a part of such registration statement has been filed;
- 5.1.10 notify each seller of such Registrable Shares of any request by the Commission or Canadian securities regulatory authorities for the amending or supplementing of such registration statement or prospectus or for additional information:
- 5.1.11 prepare and file with the Commission and/or Canadian securities regulatory authorities, promptly upon the request of any seller of such Registrable Shares, any amendments or supplements to such registration statement or prospectus which, in the written opinion of counsel selected by the holders of a majority of the Registrable Shares being registered, may be required under the Securities Act or Canadian Securities Laws in connection with the distribution of Registrable Shares by such seller;
- 5.1.12 prepare and promptly file with the Commission and/or Canadian securities regulatory authorities and promptly notify each seller of such Registrable Shares of the filing of such amendment or supplement to such registration statement or prospectus as may be necessary to correct any statements or omissions if, at the time when a prospectus relating to such securities is required to be delivered under the Securities Act and/or Canadian Securities Laws, any event shall have occurred as the result of which any such prospectus or any other prospectus as then in effect would include a misleading or untrue statement of a material fact or omit to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading;
- 5.1.13 advise each seller of such Registrable Shares, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission or any Canadian securities regulatory authorities suspending the effectiveness of such registration statement or the initiation or threatening of any proceeding for such purpose and promptly use its best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order is issued:
- 5.1.14 at least 72 hours prior to the filing of any registration statement or prospectus or any amendment or supplement to such registration statement or prospectus furnish a copy thereof to each seller of such Registrable Shares and refrain from filing any such registration statement, prospectus, amendment or supplement to which counsel selected by the Holders of a majority of the Registrable Shares being registered shall have objected in writing on the grounds that such amendment or supplement may not comply

9

#### **Table of Contents**

in all material respects with the requirements of the Securities Act and/or Canadian Securities Laws;

- 5.1.15 at the request of any seller of such Registrable Shares furnish on the date or dates provided for in the underwriting agreement, if any, or upon the effective date of the registration statement: (i) an opinion of counsel, addressed to the underwriters, if any, and the sellers of Registrable Shares, covering such matters as such underwriters, if any, and sellers may reasonably request and as are customarily covered by the issuer s counsel in an underwritten offering; and (ii) a letter or letters from the independent certified public accountants of the Corporation addressed to the underwriters, if any, and the sellers of Registrable Shares, covering such matters as such underwriters, if any, and sellers may reasonably request and as are customarily covered in accountant s letters in connection with an underwritten offering;
- 5.1.16 during such time as any Holders of Registrable Securities may be engaged in a distribution of Registrable Shares under the Securities Act, comply with Regulation M promulgated under the Exchange Act or any equivalent provision under Canadian Securities Laws, in each case to the extent applicable;
- 5.1.17 participate with the Holders of Registrable Securities and make available at reasonable times its senior executives, including its chief executive officer and chief financial officer, at any road show in connection with an underwritten offering; and
- 5.1.18 otherwise use its best efforts to comply with the provisions of the Securities Act and/or Canadian Securities Laws with respect to the disposition of all securities covered by such registration statement in accordance with the intended method of disposition and, if applicable, to make generally available to its security holders, as soon as reasonably practicable, an earnings statement satisfying the provisions of section 11(a) of the Securities Act and Rule 158 thereunder or any equivalent provision under Canadian Securities Laws.
- **5.2** <u>Holders</u> <u>Obligations</u>. Each Holder of Registrable Shares that sells Registrable Shares pursuant to a registration under this Agreement agrees as follows:
- 5.2.1 such seller shall cooperate as reasonably requested by the Corporation with the Corporation in connection with the preparation of the registration statement, and for so long as the Corporation is obligated to file and keep effective the registration statement, shall provide to the Corporation, in writing, for use in the registration statement, all such information regarding such seller and its plan of distribution of the Registrable Shares as may be reasonably necessary to enable the Corporation to prepare the registration statement and prospectus covering the Registrable Shares, to maintain the currency and effectiveness thereof and otherwise to comply with all applicable requirements of law in connection therewith; and
- 5.2.2 during such time as such seller may be engaged in a distribution of the Registrable Shares, such seller shall (i) comply with Regulation M promulgated under the Exchange Act or any equivalent provision under Canadian Securities Laws, in each case to the

10

#### **Table of Contents**

extent applicable, (ii) distribute the Registrable Shares under the registration statement solely in the manner described in the registration statement and (iii) cease distribution of such Registrable Shares pursuant to such registration statement upon receipt of written notice from the Corporation that the prospectus covering the Registrable Shares contains any misleading or untrue statement of a material fact or omits a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances in which they were made, not misleading.

**5.3 Failure to Comply.** Any Holder who shall fail to provide the Corporation with any of the information required by section 5.2.1 shall be deemed to have requested that its Registrable Shares be withdrawn from such registration statement and (i) in the case of a registration pursuant to section 3.1, the deemed withdrawal shall be irrevocable and the Holder shall no longer have any right to include Registrable Shares in the registration (as set forth in section 3.6) and (ii) in the case of a registration pursuant to section 2.1, the Corporation shall be deemed to have registered Registrable Shares for the purpose of calculating the number of Demand Registrations to which the Holders are still entitled pursuant to section 2.1.2.

# ARTICLE 6 REGISTRATION EXPENSES.

- **6.1** Corporation s Expenses on a Demand Registration. All expenses incident to the Corporation s performance of or compliance with Article 2 of this Agreement, including, without limitation, all registration, listing and filing fees, fees of transfer agents and registrars, fees and expenses of compliance with securities or blue sky laws, fees of the National Association of Securities Dealers, Inc. (and any Canadian equivalent), printing expenses, road show expenses, fees and disbursements of counsel for the Corporation, fees and expenses of the Corporation s independent certified public accountants, the expenses and fees for listing the securities to be registered on each securities exchange on which any Common Shares are then listed and the fees and expenses of any underwriters (excluding underwriting discounts and commissions attributable to the Registrable Shares included in such registration) and other Persons retained by the Corporation (all such expenses being herein called **Registration Expenses**), will be borne by the Holders. The Corporation will pay its internal expenses (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), the expense of any annual audit or quarterly review and the expense of any liability insurance obtained by the Corporation.
- **6.2** Corporation s Expenses on a Piggyback Registration. All expenses incident to the Corporation s performance of or compliance with Article 3 of this Agreement, including, without limitation, all Registration Expenses will be borne pro rata by the Corporation and the Holders, based on the number of Registrable Shares included in the applicable offering relative to the number of Registrable Shares included in such offering by the Corporation and the Holders. The Corporation will pay its internal expenses (including, without limitation, all salaries and expenses of its officers and employees performing

11

### **Table of Contents**

legal or accounting duties), the expense of any annual audit or quarterly review and the expense of any liability insurance obtained by the Corporation.

**6.3** Withdrawal. Notwithstanding section 6.1 above, the Holders of the Registrable Shares agree that in the event any such Holders withdraw any registration demand, such Holders shall either pay the Registration Expenses incurred in such registration or count such withdrawn demand toward their three permitted demands, as set forth in section 2.1.

# ARTICLE 7 INDEMNIFICATION.

7.1 Indemnification by the Corporation. In the event of a registration of the Registrable Shares pursuant to the terms hereof, the Corporation agrees to indemnify, hold harmless and defend, to the fullest extent permitted by law, each seller of Registrable Shares, its officers, directors and partners and each Person who controls such seller (within the meaning of the Securities Act, the Exchange Act or Canadian Securities Laws) against all losses, claims, damages, liabilities and expenses (including, without limitation, legal and other expenses except as limited by section 7.4), including any of the foregoing incurred in settlement of any litigation, commenced or threatened, caused by (i) any misleading or untrue or alleged misleading or untrue statement of a material fact contained in any registration statement under which such Registrable Shares were registered, any prospectus or preliminary prospectus contained therein or any amendment thereof or supplement thereto, (ii) any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances in which they were made, not misleading, or (iii) any violation by the Corporation of the Securities Act, the Exchange Act and Canadian Securities Laws or any rule or regulation promulgated under such laws applicable to the Corporation and relating to action or inaction required of the Corporation. In connection with an underwritten offering, the Corporation will indemnify such underwriters, their officers and directors and each Person who controls such underwriters (within the meaning of the Securities Act, the Exchange Act or Canadian Securities Laws) to the same extent as provided above with respect to the indemnification of the sellers of Registrable Shares (and with the same exception above with respect to information furnished by such underwriter or controlling person thereof) and in connection therewith the Corporation shall enter into an underwriting agreement in customary form containing such provisions for indemnification and contribution as shall be reasonably requested by the underwriters. The reimbursements required by this section 7.1 will be made by periodic payments during the course of the investigation or defense, as and when bills are received or expenses incurred. Notwithstanding the foregoing, the Corporation shall not be liable to any such indemnified party in any such case to the extent such claim arises solely out of any untrue statement or alleged untrue statement of a material fact or omission or alleged omission of a material fact made in such registration statement or amendment thereof or supplement thereto or in any such prospectus or any preliminary, final or summary prospectus or amendment or supplement thereto in reliance upon, and in conformity with, written information furnished to the Corporation by, or on behalf of, such indemnified party specifically for use therein.

12

#### **Table of Contents**

- **7.2 Indemnification by the Holders.** In the event of a registration of the Registrable Shares pursuant to the terms hereof, each Holder that sells any Registrable Shares pursuant thereto agrees to jointly and severally indemnify, hold harmless and defend, to the fullest extent permitted by law, the Corporation, its directors and officers and each Person who controls the Corporation (within the meaning of the Securities Act, the Exchange Act or Canadian Securities Laws) and each underwriter and controlling person thereof against all losses, claims, damages, liabilities and expenses (including, without limitation, legal and other expenses except as limited by section 7.4), including any of the foregoing incurred in settlement of any litigation, commenced or threatened, caused by (i) any misleading or untrue or alleged misleading or untrue statement of a material fact contained in any registration statement under which such Registrable Shares were registered, any prospectus or preliminary prospectus contained therein, or any amendment thereof or supplement thereto, (ii) any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances in which they were made, not misleading, or (iii) any violation by the Corporation of the Securities Act, the Exchange Act and Canadian Securities Laws or any rule or regulation promulgated under such laws applicable to the Corporation and relating to action or inaction required of the Corporation, but only to the extent that such misleading or untrue or alleged misleading or untrue statement or omission or alleged omission is contained in any information so furnished in writing to the Corporation or any managing underwriter by such seller or a controlling person thereof expressly for use therein. The reimbursements required by this section 7.2 will be made by periodic payments during the course of the investigation or defense, as and when bills are received or expenses incurred.
- 7.3 Procedure. Any Person entitled to indemnification hereunder will (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give such notice shall not limit the rights of such Person except to the extent such failure to give notice shall materially prejudice the rights of the indemnifying party) and (ii) unless in such indemnified party s reasonable judgment (with written advice of counsel) a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party will not enter into any settlement without the indemnified party s prior written consent unless such settlement includes an unconditional release of the indemnified party from liability relating to the claim. An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim will not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment (with written advice of counsel) of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim.
- **7.4** Alternative Contribution. Each party hereto agrees that, if for any reason the indemnification provisions contemplated by section 7.1 or section 7.2 are unavailable to or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities or expenses (or actions in respect thereof) referred to therein, then

13

#### **Table of Contents**

each indemnifying party shall contribute to the amount paid or payable by such indemnified party as a result of such losses, claims, damages, liabilities or expenses (or actions in respect thereof) in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party as well as any other relevant equitable considerations. The relative fault of such indemnifying party and indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact relates to information supplied by such indemnifying party or indemnified party, and the parties relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The parties hereto agree that it would not be just and equitable if contribution pursuant to this section 7.4 were determined by pro rata allocation (even if the holders or any underwriters or all of them were treated as one entity for such purpose) or by any other method of allocation which does not take account of the equitable considerations referred to in this section 7.4. The amount paid or payable by an indemnified party as a result of the losses, claims, damages, liabilities or expenses (or actions in respect thereof) referred to above shall be deemed to include any legal or other fees or expenses reasonably incurred by such indemnified party in connection with investigating or, except as provided in section 7.3, defending any such action or claim. Notwithstanding the provisions of this section 7.4, no holder shall be required to contribute an amount greater than the dollar amount of the proceeds received by such holder with respect to the sale of any Registrable Shares. No person guilty of fraudulent misrepresentation (within the meaning of section 11(f) of the Securities Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation. The holders obligations in this section 7.4 to contribute shall be several in proportion to the amount of Registrable Shares registered by them and not joint.

**7.5** Full Force and Effect. The indemnification and contribution provided for under this Agreement will remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling Person of such indemnified party.

## ARTICLE 8 OTHER COVENANTS AND AGREEMENTS

8.1 Compliance with Rule 144. The Corporation shall (i) make and keep public information available, as those terms are understood and defined in Rule 144, (ii) file with the Commission and Canadian securities regulatory authorities in a timely manner all reports and other documents required of the Corporation under the Securities Act, the Exchange Act and Canadian Securities Laws and (iii) at the request of any holder who proposes to sell securities in compliance with Rule 144 or an equivalent prospectus exemption under Canadian Securities Laws, forthwith furnish to such holder a written statement of compliance with the reporting requirements of the Commission as set forth in Rule 144 and compliance with equivalent provisions of Canadian Securities Laws and make available to such holders such information as will enable the holders to make sales

14

#### **Table of Contents**

pursuant to Rule 144 or an equivalent prospectus exemption under Canadian Securities Laws.

- **8.2** Participation in Underwritten Registrations. No Person may participate in any registration hereunder which is underwritten unless such Person (a) agrees to sell such Person s securities on the basis provided in any underwriting arrangements approved by the Person or Persons entitled hereunder to approve such arrangements and (b) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents required under the terms of such underwriting arrangements.
- **8.3 No Inconsistent Agreements.** The Corporation will not hereafter enter into any agreement with respect to its securities which is inconsistent with the rights granted to the holders of the Registrable Shares in this Agreement.
- **8.4** Transfer of Registration Rights. The rights and obligations granted to each Holder under this Agreement may not be assigned to any Person.
- **8.5** Termination of Registration Rights. The registration rights hereunder shall terminate on such date as the Holders together, directly or indirectly, own less than 5% of the then issued and outstanding Common Shares.

## ARTICLE 9 GENERAL

- 9.1.1 **Remedies**. Any Person having rights under any provision of this Agreement will be entitled to enforce such rights specifically, to recover damages caused by reason of any breach of any provision of this Agreement and to exercise all other rights granted by law.
- 9.1.2 <u>Amendments and Waivers</u>. Except as otherwise expressly provided herein, the provisions of this Agreement may be amended or waived at any time only by the written agreement of the Corporation and Holder Approval; provided that any such amendment or waiver shall apply equally to all Holders of Registrable Shares. Any waiver, permit, consent or approval of any kind or character on the part of any such holders of any provision or condition of this Agreement must be made in writing and shall be effective only to the extent specifically set forth in writing.
- 9.1.3 <u>Successors and Assigns</u>. Except as otherwise expressly provided herein, all covenants and agreements contained in this Agreement by or on behalf of any of the parties hereto will bind and inure to the benefit of the respective successors and assigns of the parties hereto. In addition and whether or not any express assignment has been made, the provisions of this Agreement which are for the benefit of the Holders are also for the benefit of, and enforceable by, any subsequent holder of Registrable Shares who consents in writing to be bound by this Agreement, each of which subsequent holders shall be deemed to be a Holder for all purposes hereof.

15

## **Table of Contents**

- 9.1.4 **<u>Final Agreement</u>**. This Agreement constitutes the final agreement of the parties concerning the matters referred to herein, and supersedes all prior agreements and understandings.
- 9.1.5 <u>Severability</u>. Whenever possible, each provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement is held to be prohibited by or invalid under applicable law, such provision will be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of this Agreement.
- 9.1.6 **Descriptive Heading.** The descriptive headings of this Agreement are inserted for convenience of reference only and do not constitute a part of and shall not be utilized in interpreting this Agreement.
- 9.1.7 **Notices.** Any notice, request, consent, agreement or approval which may or is required to be given pursuant to this Agreement shall be in writing and shall be sufficiently given or made if delivered or telecopied in the case of::
  - (a) the Corporation, addressed as follows:

Noranda Inc. 181 Bay Street Suite 200, BCE Place Toronto, Ontario M5J 2T3

**Attention:** General Counsel

Telecopier: (416) 982-7111 Telephone: (416) 982-7423

with a copy to:

McCarthy Tétrault LLP Toronto Dominion Tower Suite 4700 Toronto, Ontario M5K 1E6

**Attention:** Garth M. Girvan

Telecopier: (416) 868-0673 Telephone: (416) 362-1812

(b) the Holders, addressed as follows:

Brascan Corporation 181 Bay Street Suite 300, Box 762 Toronto, Ontario M5J 2T3

Table of Contents 137

16

### **Table of Contents**

**Attention:** Jeffrey M. Blidner

Telecopier: (416) 363-9491 Telephone: (416) 363-2856

with a copy to:

Torys LLP 79 Wellington St. W. Suite 3000, Box 270 Toronto, Ontario M5K 1N2

**Attention:** Brian A. Davis

Telecopier: (416) 865-7380 Telephone: (416) 865-7500

or to such other address as the relevant party may from time to time advise by notice in writing given pursuant to this section. The date of receipt of any such notice, request, consent, agreement, or approval shall be deemed to be the date of delivery or telecopy (if during normal business hours, or, if not, the next Business Day).

- 9.1.8 **Governing Law**. The validity, meaning and effect of this agreement shall be determined in accordance with the laws of the province of Ontario.
- 9.1.9 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which when so executed and delivered shall be deemed an original, and such counterparts together shall constitute one instrument. Each party shall receive a duplicate original of the counterpart copy or copies executed by it and the Corporation.
- 9.1.10 **Injunctive Relief**. Each of the parties hereto acknowledges that in the event of a breach by any of them of any material provision of this Agreement, the aggrieved party may be without an adequate remedy at law. Each of the parties therefore agrees that in the event of such a breach hereof the aggrieved party may elect to institute and prosecute proceedings in any court of competent jurisdiction to enforce specific performance or to enjoin the continuing breach hereof. By seeking or obtaining any such relief, the aggrieved party shall not be precluded from seeking or obtaining any other relief to which it may be entitled.
- 9.1.11 <u>Attorneys Fees</u>. In the event of any action, arbitration or suit based upon or arising out of any actual or alleged breach by any party of any representation, warranty, covenant or agreement in this Agreement, the prevailing party shall be entitled to recover its reasonable attorneys fees and expenses of such action, arbitration or suit from the other party, in addition to any other relief ordered by any proper arbitration proceeding or court.

17

### **Table of Contents**

This Agreement was executed on the date first set forth above.

## NORANDA INC.

By: << Derek G. Pannell>>
Name: Derek G. Pannell

Title: President and Chief Executive

Officer

## **BRASCAN CORPORATION**

By: << *Jeffrey M. Blidner*>>
Name: Jeffrey M. Blidner
Title: Managing Partner

## **BRASCADE CORPORATION**

By: << Bryan K. Davis>> Name: Bryan K. Davis

Title: Vice-President and Treasurer

18